United States SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FO	RM	8-K

Current Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (date of earliest event reported): December 7, 2020

CANNAE HOLDINGS, INC.

(Exact name of Registrant as Specified in its Charter)

Delaware (State or Other Jurisdiction of Incorporation or Organization)

1-38300 (Commission File Number)

82-1273460 (IRS Employer Identification Number)

1701 Village Center Circle Las Vegas, Nevada 89134 (Addresses of Principal Executive Offices)

(702) 323-7330 (Registrant's Telephone Number, Including Area Code)

(Former Name or Former Address, if Changed Since Last Report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:				
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)			
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)			
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))			
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))			
Securities registered pursuant to Section 12(b) of the Act:				
	Title of Each Class	Trading Symbol	Name of Each Exchange on Which Registered	
C	Title of Each Class annae Common Stock, \$0.0001 par value			
Indic		Symbol CNNE ng growth company as defined in Rule 40!	on Which Registered New York Stock Exchange	
Indic chap	cannae Common Stock, \$0.0001 par value rate by check mark whether the registrant is an emergin	Symbol CNNE ng growth company as defined in Rule 40!	on Which Registered New York Stock Exchange	

Item 1.01

Entry Into a Material Agreement

On December 7, 2020, Cannae Holdings, LLC ("<u>Cannae</u>"), a subsidiary of Cannae Holdings, Inc., entered into a common stock subscription agreement (the "<u>Subscription Agreement</u>"), with Paysafe Limited, an exempted limited company incorporated under the laws of Bermuda (the "<u>Company</u>") and Foley Trasimene Acquisition Corp. II ("<u>FTAC II</u>"), pursuant to which, Cannae has agreed to purchase from the Company and the Company has agreed to issue and sell to Cannae, \$350,000,000 (the "<u>Purchase Price</u>") of common shares, par value \$0.001 per share, of the Company (the "<u>Company Common Shares</u>") at a purchase price of \$10.00 per share (the "<u>PIPE Investment</u>"). The closing of the PIPE Investment is conditioned on all conditions set forth in that certain Agreement and Plan of Merger, dated as of December 7, 2020, by and among the Company, FTAC II, Paysafe Group Holdings Limited ("<u>PGHL</u>") and the other parties thereto (the "<u>Merger Agreement</u>"), having been satisfied or waived and other customary closing conditions, and the transactions contemplated by the Merger Agreement (the "<u>Business Combination</u>") will be consummated immediately following the closing of the PIPE Investment. The proceeds from the Subscription Agreement will be used to partially fund the cash consideration to be paid by FTAC II to PGHL at the closing of the Business Combination. The Business Combination is expected to close in the first half of 2021. As a result of the PIPE Investment and other investments by Cannae in Trasimene Capital FT, LP II ("<u>FTAC II Sponsor</u>"), and upon the consummation of the Business Combination, Cannae is expected to hold approximately 7.5% of the outstanding Company Common Shares. In connection with the PIPE Investment, the Company has agreed to pay Cannae a fee of 1.6% of the Purchase Price upon the consummation of the Business Combination.

The Subscription Agreement will terminate upon the earliest to occur of (i) the termination of the Merger Agreement, (ii) the mutual written agreement of the parties thereto or (iii) at Cannae's election, on or after December 7, 2021, subject to automatic extension if any action for specific performance or other equitable relief by PGHL or the Company with respect to the Merger Agreement, the other transactions specified in the Merger Agreement or otherwise regarding the Business Combination is commenced or pending on or prior to December 7, 2021...

The foregoing description of the Subscription Agreement is not complete and is qualified in its entirety by reference to the Subscription Agreement, which is attached as Exhibit 10.1, to this Current Report and incorporated herein by reference

Shareholders Agreement

In connection with the execution of the Merger Agreement, PGHL, the Company, Pi Jersey Topco Limited ("Pi Topco"), Cannae, FTAC II Sponsor, certain funds affiliated with CVC (the "CVC Investors") and certain funds affiliated with Blackstone (the "Blackstone Investors") have agreed to enter into a Shareholders Agreement (the "Shareholders Agreement") at the closing of the Business Combination. Pursuant to terms of the Shareholders Agreement, effective as of the date the closing of the Business Combination occurs (the "Closing Date"), the Company's board of directors (the "Board") is anticipated to be comprised of eleven directors as follows: (i) four directors designated by Cannae and FTAC II Sponsor (together, the "FTAC Investors"), (ii) four directors designated by Cannae, the CVC Investors and the Blackstone Investors (which such directors will be independent directors) and (v) the chief executive officer of PGHL.

Following the Closing Date, the FTAC Investors' right to designate directors to the Board is subject to (a) the amount of Company Common Shares held by the FTAC Investors at a given point in time, as compared to the Company Common Shares held by the FTAC Investors on the Closing Date and (b) the amount of Company Common Shares held by the FTAC Investors as compared to the number of Company Common Shares then outstanding at a given time. So long as the FTAC Investors hold at least 50% of the Company Common Shares held by the FTAC Investors on the Closing Date, the FTAC Investors will have the right to designate four directors and Cannae will have the right to jointly with the CVC Investors on the Closing Date, they will have the right to designate (1) if the FTAC Investors hold at least 7.5% of the aggregate outstanding Company Common Shares, four directors and Cannae will have the right to jointly with the CVC Investors and the Blackstone Investors, designate two directors, and to consent to any individual nominated for

election to the Board seat initially occupied by the chief executive officer of PGHL; (2) if the FTAC Investors hold at least 6.25% (but less than 7.5%) of the aggregate outstanding Company Common Shares, two directors; and (3) if the FTAC Investors hold at least 2.5% (but less than 6.25%) of the aggregate outstanding Company Common Shares, one director.

Additionally, following the Closing Date, each of the CVC Investors' and Blackstone Investors' rights to designate directors to the Board is subject to the aggregate amount of Company Common Shares held by such investors as compared to the number of Company Common Shares outstanding at any time. If the CVC Investors or the Blackstone Investors, as the case may be, directly hold or indirectly, as set forth on the books and records of PGHL or Pi Topco, as applicable, are attributed at least 7.5% of the aggregate outstanding Company Common Shares, the CVC Investors or the Blackstone Investors, are, respectively, entitled to designate two directors. If the CVC Investors or the Blackstone Investors, as the case may be, directly hold or indirectly, as set forth on the books and records of PGHL or Pi Topco, as applicable, are attributed at least 2.5% (but less than 7.5%) of the aggregate outstanding Company Common Shares, then the CVC Investors or the Blackstone Investors, are, respectively, entitled to appoint one director. If the CVC Investors or the Blackstone Investors, as the case may be, hold at least 7.5% of the aggregate outstanding Company Common Shares, then the CVC Investors or the Blackstone Investors, are, respectively, entitled to jointly with Cannae and the Blackstone Investors (in the case of the CVC Investors) and the CVC Investors in the case of the Blackstone Investors) designate two directors and to consent to any individual nominated for election to the Board seat initially occupied by the chief executive officer of PGHL. Additionally, each of the CVC Investors and the Blackstone Investors have agreed not to transfer any Company Common Shares for a period beginning on the Closing Date and ending on the earlier of (A) 180 days thereafter or (B) if the volume weighted average price of the Company Common Shares equals or exceeds \$12.00 per share for any 20 trading days within a 30 trading day period, 60 days thereafter. The foregoing description of the Shareholders Agreement is not complete and is qu

Registration Rights Agreement

In connection with the execution of the Merger Agreement, the Company, Pi Topco, PGHL, Cannae, FTAC II Sponsor, certain funds affiliated with CVC and certain funds affiliated with Blackstone have agreed to entered into a Registration Rights Agreement (the "Registration Rights Agreement") at the closing of the Business Combination. The Registration Rights Agreement will provide these holders (and their permitted transferees) with the right to require the Company, at the Company's expense, to register Company Common Shares that they hold on customary terms for a transaction of this type. The Registration Rights Agreement will also provide that the Company pay certain expenses of the electing holders relating to such registrations and indemnify them against certain liabilities that may arise under the Securities Act of 1933, as amended. The foregoing description of the Registration Rights Agreement is not complete and is qualified in its entirety by reference to the form of Registration Rights Agreement, which is attached as Exhibit 99.2 to this Current Report.

Amended and Restated Sponsor Agreement

In connection with the execution of the Merger Agreement, FTAC II amended and restated (a) that certain letter agreement (the "Sponsor Agreement"), dated August 21, 2020, by and among FTAC II and FTAC II Sponsor and (b) each of the letter agreements, dated as of August 21, 2020, by and between FTAC II and William P. Foley, II, Richard N. Massey, Mark D. Linehan, Erika Meinhardt, David W. Ducommun, Michael L. Gravelle, C. Malcolm Holland and Bryan D. Coy (the "Insiders"), pursuant to which, among other things, FTAC II Sponsor, the Insiders, Cannae and Cannae Holdings, Inc. (collectively, the "Sponsor Persons") agreed (i) to vote any shares of FTAC II's securities in favor of the Transactions and other FTAC II Stockholder Matters, (ii) not to redeem any shares of Class A Common Stock or Class B Common Stock in connection with the stockholder redemption, (iii) not to take any action to solicit any offers relating to an alternative business combination, (iv) to use reasonable best efforts to obtain required regulatory approvals, (v) not to transfer any Company Common Shares for a period beginning on the Closing Date and ending on the earlier of (A) 270 days thereafter or (B) if the volume weighted average price of the Company Common Shares equals or exceeds \$12.00 per share for any 20 trading days within a 30 trading day period, 150 days thereafter, and (vi) to be bound to certain other obligations as described therein (the "Amended and Restated Sponsor Agreement"). The foregoing description of the Amended and Restated Sponsor Agreement is not complete and is qualified in its entirety by reference to the Amended and Restated Sponsor Agreement as Exhibit 10.2 to this Current Report and incorporated herein by reference

Item 7.01. Regulation FD Disclosure.

On December 7, 2020, Cannae Holdings, Inc. issued a press release (the "<u>Press Release</u>") announcing the PIPE Investment. The Press Release is attached hereto as Exhibit 99.3 and incorporated by reference herein.

The information in this Item 7.01, including Exhibit 99.3, is furnished and shall not be deemed "filed" for purposes of Section 18 of the Exchange Act, or otherwise subject to liabilities under that section, and shall not be deemed to be incorporated by reference into the filings of Cannae Holdings, Inc. under the Securities Act or the Exchange Act, regardless of any general incorporation language in such filings. This Current Report will not be deemed an admission as to the materiality of any information of the information in this Item 7.01, including Exhibit 99.3.

Item 9.01. Financial Statement and Exhibits.

(d) Exhibits.

The Exhibit Index is incorporated by reference herein.

EXHIBIT INDEX

Exhibit No.	Description
10.1	Subscription Agreement, dated as of December 7, 2020, by and among Foley Trasimene Acquisition Corp. II, Paysafe Limited and Cannae Holdings, LLC.
10.2	Amended and Restated Sponsor Agreement, dated as of December 7, 2020, by and among Foley Trasimene Acquisition Corp. II, Trasimene Capital FT, LP II, Cannae Holdings, Inc., Cannae Holdings, LLC, and the Insiders.
99.1	Form of Shareholders Agreement.
99.2	Form of Registration Rights Agreement.
99.3	Press Release of Cannae Holdings, Inc., dated December 7, 2020.
104	Cover Page Interactive Data File - the cover page XBRL tags are embedded within the Inline XBRL document.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Cannae Holdings, Inc.

Date: December 7, 2020 By: /s/ Michael L. Gravelle

Name: Michael L. Gravelle

Title: Executive Vice President, General Counsel, and Corporate

Secretary

SUBSCRIPTION AGREEMENT

This SUBSCRIPTION AGREEMENT (this "Subscription Agreement") is entered into this 7th day of December, 2020, by and among Paysafe Limited, an exempted limited company incorporated under the laws of Bermuda (the "Issuer"), Foley Trasimene Acquisition Corp. II, a Delaware corporation (the "SPAC"), and the undersigned ("Subscriber" or "you"). Defined terms used but not otherwise defined herein shall have the respective meanings ascribed thereto in the Merger Agreement (as defined below).

WHEREAS, the Issuer, Paysafe Group Holdings Limited, a private limited company incorporated under the laws of England and Wales ("Paysafe"), the SPAC and the other parties named therein will, immediately following the execution of this Subscription Agreement, enter into that certain Agreement and Plan of Merger, dated as of the date hereof (as amended, modified, supplemented or waived from time to time in accordance with its terms, the "Merger Agreement"), pursuant to which, *inter alia*, (i) an indirect, wholly owned subsidiary of the Issuer will be merged with and into the SPAC, with the SPAC surviving as a wholly owned subsidiary of the Issuer (the "Merger") and (ii) Paysafe will contribute to the Issuer or a wholly owned subsidiary of the Issuer all of the outstanding equity interests in Pi Jersey Holdco 1.5 Limited, a Jersey private limited company (such contribution together with the Merger and the other transactions contemplated by the Merger Agreement, the "Transactions");

WHEREAS, in connection with the Transactions, Subscriber desires to subscribe for and purchase from the Issuer that number of shares of the Issuer's common shares (the "Common Shares") set forth on the signature page hereto (the "Subscribed Shares") for a purchase price of \$10.00 per share, and for the aggregate purchase price set forth on Subscriber's signature page hereto (the "Purchase Price"), and the Issuer desires to issue and sell to Subscriber the Subscribed Shares in consideration of the payment of the Purchase Price therefor by or on behalf of Subscriber to the Issuer, all on the terms and subject to the conditions set forth herein; and

WHEREAS, certain other "qualified institutional buyers" (as defined in Rule 144A under the Securities Act of 1933, as amended (the "Securities Act")) or "accredited investors" (within the meaning of Rule 501(a) under the Securities Act) (each, an "Other Subscriber") have, severally and not jointly, entered into separate subscription agreements with the Issuer (the "Other Subscription Agreements"), pursuant to which such Other Subscribers have agreed to purchase Common Shares on the Closing Date (as defined below) at the same per share purchase price as Subscriber, and the aggregate amount of securities to be sold by the Issuer pursuant to this Subscription Agreement and the Other Subscription Agreements equals, as of the date hereof, 200,000,000 Common Shares.

NOW, THEREFORE, in consideration of the foregoing and the mutual representations, warranties and covenants, and subject to the conditions, herein contained, and intending to be legally bound hereby, the parties hereto hereby agree as follows:

For ease of administration, this single Subscription Agreement is being executed so as to enable each Subscriber identified on the signature page to enter into a Subscription Agreement, severally, but not jointly. The parties agree that (i) the Subscription Agreement shall

be treated as if it were a separate agreement with respect to each Subscriber listed on the signature page, as if each Subscriber entity had executed a separate Subscription Agreement naming only itself as Subscriber, and (ii) no Subscriber listed on the signature page shall have any liability under the Subscription Agreement for the obligations of any other Subscriber so listed.

- 1. <u>Subscription</u>. Subject to the terms and conditions hereof, at the Closing (as defined below), Subscriber hereby agrees, upon the substantially concurrent consummation of the Transactions, to subscribe for and purchase, and the Issuer hereby agrees to issue and sell to Subscriber, upon the payment of the Purchase Price, the Subscribed Shares (such subscription and issuance, the "**Subscription**").
 - 2. Representations, Warranties and Agreements.
- 2.1 <u>Subscriber's Representations, Warranties and Agreements</u>. To induce the Issuer to issue the Subscribed Shares, Subscriber hereby represents and warrants to the Issuer and the SPAC and acknowledges and agrees with the Issuer and the SPAC, as of the date hereof and as of the Closing Date, as follows:
 - 2.1.1 If Subscriber is not an individual, Subscriber has been duly formed or incorporated and is validly existing in good standing under the laws of its jurisdiction of incorporation or formation, with power and authority to enter into, deliver and perform its obligations under this Subscription Agreement. If Subscriber is an individual, Subscriber has the authority to enter into, deliver and perform its obligations under this Subscription Agreement.
 - 2.1.2 If Subscriber is not an individual, this Subscription Agreement has been duly authorized, validly executed and delivered by Subscriber. If Subscriber is an individual, the signature on this Subscription Agreement is genuine, and Subscriber has legal competence and capacity to execute the same. Assuming that this Subscription Agreement constitutes the valid and binding agreement of the other parties hereto, this Subscription Agreement is the valid and binding obligation of Subscriber, is enforceable against Subscriber in accordance with its terms, except as may be limited or otherwise affected by (i) bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or other laws relating to or affecting the rights of creditors generally, and (ii) principles of equity, whether considered at law or equity.
 - 2.1.3 The execution, delivery and performance by Subscriber of this Subscription Agreement and the consummation of the transactions contemplated herein do not and will not (i) conflict with or result in a breach or violation of any of the terms or provisions of, or constitute a default under, or result in the creation or imposition of any lien, charge or encumbrance upon any of the property or assets of Subscriber or any of its subsidiaries pursuant to the terms of any indenture, mortgage, charge, deed of trust, loan agreement, lease, license or other agreement or instrument to which Subscriber or any of its subsidiaries is a party or by which Subscriber or any of its subsidiaries is bound or to which any of the property or assets

of Subscriber or any of its subsidiaries is subject, which would reasonably be expected to have a material adverse effect on the legal authority of Subscriber to enter into and timely perform its obligations under this Subscription Agreement (a "Subscriber Material Adverse Effect"), (ii) if Subscriber is not an individual, result in any violation of the provisions of the organizational documents of Subscriber or any of its subsidiaries or (iii) result in any violation of any statute or any judgment, order, rule or regulation of any court or governmental agency or body, domestic or foreign, having jurisdiction over Subscriber or any of its subsidiaries or any of their respective properties that would reasonably be expected to have a Subscriber Material Adverse Effect.

2.1.4 Subscriber (i) is (a) a "qualified institutional buyer" (as defined in Rule 144A under the Securities Act) or an "accredited investor" within the meaning of Rule 501(a) under the Securities Act, (b) an Institutional Account as defined in FINRA Rule 4512(c) and (c) a sophisticated institutional investor, experienced in investing in transactions of the type contemplated by this Subscription Agreement and capable of evaluating investment risks independently, both in general and with regard to all transactions and investment strategies involving a security or securities, including Subscriber's participation in the purchase of the Subscribed Shares, in each case, satisfying the applicable requirements set forth on Schedule I, (ii) is acquiring the Subscribed Shares only for its own account and not for the account of others, or if Subscriber is subscribing for the Subscribed Shares as a fiduciary or agent for one or more investor accounts, each owner of such account is a qualified institutional buyer, and Subscriber has full investment discretion with respect to each such account, and the full power and authority to make the acknowledgements, representations, warranties and agreements herein on behalf of each owner of each such account, for investment purposes only and not with a view to any distribution of the Subscribed Shares in any manner that would violate the securities laws of the United States or any other applicable jurisdiction and (iii) is not acquiring the Subscribed Shares with a view to, or for offer or sale in connection with, any distribution thereof in violation of the Securities Act (and shall provide the requested information on Schedule I following the signature page hereto). Subscriber is not an entity formed for the specific purpose of acquiring the Subscribed Shares.

2.1.5 Subscriber understands that the Subscribed Shares are being offered in a transaction not involving any public offering within the meaning of the Securities Act and that the Subscribed Shares have not been registered under the Securities Act. Subscriber understands that the Subscribed Shares may not be resold, transferred, pledged or otherwise disposed of by Subscriber absent an effective registration statement under the Securities Act, except (i) to the Issuer or a subsidiary thereof, (ii) to non-U.S. persons pursuant to offers and sales that occur solely outside the United States within the meaning of Regulation S under the Securities Act or (iii) pursuant to another applicable exemption from the registration requirements of the Securities Act, and in each of cases (i) and (iii), in accordance with any applicable securities laws of the states and other jurisdictions of the United States, and that any certificates representing the Subscribed Shares shall contain a legend to such effect (provided that such legends will be eligible for removal upon compliance with the

relevant resale provisions of Rule 144). Subscriber acknowledges that the Subscribed Shares will not be eligible for resale pursuant to Rule 144A promulgated under the Securities Act. Subscriber understands and agrees that the Subscribed Shares will be subject to the foregoing restrictions and, as a result, Subscriber may not be able to readily resell the Subscribed Shares and may be required to bear the financial risk of an investment in the Subscribed Shares for an indefinite period of time. Subscriber understands that it has been advised to consult independent legal counsel prior to making any offer, resale, pledge or transfer of any of the Subscribed Shares. Subscriber has determined based on its own independent review and such professional advice as it deems appropriate that its purchase of the Subscribed Shares are a suitable investment for Subscriber, notwithstanding the substantial risks inherent in investing in or holding the Subscribed Shares.

- 2.1.6 Subscriber understands and agrees that Subscriber is purchasing the Subscribed Shares directly from the Issuer. Subscriber further acknowledges that there have been no representations, warranties, covenants or agreements made to Subscriber by the Issuer, the SPAC or any of their respective officers or directors, expressly or by implication, other than those representations, warranties, covenants and agreements expressly set forth in this Subscription Agreement.
- 2.1.7 Subscriber represents and warrants that its acquisition and holding of the Subscribed Shares will not constitute or result in a non-exempt prohibited transaction under Section 406 of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), Section 4975 of the Internal Revenue Code of 1986, as amended (the "Code"), or any applicable other federal, state, local, non-U.S. or other laws or regulations that are similar to such provisions of ERISA or the Code (collectively, "Similar Laws").
- 2.1.8 In making its decision to purchase the Subscribed Shares, Subscriber represents that it has relied solely upon independent investigation made by Subscriber and the representations, warranties and covenants of the Issuer and the SPAC contained in this Subscription Agreement. Without limiting the generality of the foregoing, Subscriber has not relied on any statements or other information provided by anyone (including Credit Suisse Securities (USA) LLC, J.P. Morgan Securities LLC and BofA Securities, Inc. (collectively, in their capacity as placement agents, the "Placement Agents")), other than the Issuer and the SPAC and their respective representatives concerning the Issuer or the SPAC or the Subscribed Shares or the offer and sale of the Subscribed Shares. Subscriber acknowledges and agrees that Subscriber has received such information as Subscriber deems necessary in order to make an investment decision with respect to the Subscribed Shares, including with respect to the Issuer, Paysafe, the SPAC and the Transactions. Subscriber represents and agrees that Subscriber and Subscriber's professional advisor(s), if any, have (i) received, reviewed and understood the offering materials made available to Subscriber and (ii) had the full opportunity to ask such questions, receive such answers and obtain such information as Subscriber and such Subscriber's professional advisor(s), if any, have deemed necessary to make an investment decision with respect

to the Subscribed Shares. Subscriber represents and warrants it is relying exclusively on its own sources of information, investment analysis and due diligence (including professional advice you deem appropriate) with respect to the Transactions, the Subscribed Shares and the business, condition (financial and otherwise), management, operations, properties and prospects of the Issuer, Paysafe, and the SPAC, including but not limited to all business, legal, regulatory, accounting, credit and tax matters.

- 2.1.9 Subscriber acknowledges and agrees that (a) each of the Placement Agents is acting solely as placement agent in connection with the Transactions and is not acting as an underwriter or in any other capacity in connection with the Subscriptions and is not and shall not be construed as a fiduciary for Subscriber or any other person or entity in connection with the Transactions, (b) the Placement Agents have not made and will not make any representation or warranty, whether express or implied, of any kind or character and have not provided any advice or recommendation in connection with the Transactions, (c) the Placement Agents will have no responsibility with respect to (i) any representations, warranties or agreements made by any person or entity under or in connection with the Transactions or any of the documents furnished pursuant thereto or in connection therewith, or the execution, legality, validity or enforceability (with respect to any person) or any thereof, or (ii) the business, condition (financial and otherwise), management, operations, properties or prospects of, or any other matter concerning the Issuer, Paysafe, the SPAC or the Transactions, and (d) the Placement Agents shall have no liability or obligation (including without limitation, for or with respect to any losses, claims, damages, obligations, penalties, judgments, awards, liabilities, costs, expenses or disbursements incurred by Subscriber, the Issuer, the SPAC or any other person or entity), whether in contract, tort or otherwise, to Subscriber, or to any person claiming through Subscriber, in respect of the Transactions.
- 2.1.10 Subscriber became aware of this offering of the Subscribed Shares solely by means of direct contact between Subscriber and the Issuer, the SPAC or one of their respective representatives. Subscriber did not become aware of this offering of the Subscribed Shares, nor were the Subscribed Shares offered to Subscriber, by any other means. Subscriber acknowledges that the Issuer represents and warrants that the Subscribed Shares (i) were not offered by any form of general solicitation or general advertising, including methods described in section 502(c) of Regulation D under the Securities Act and (ii) assuming the representations and warranties of the Issuer are true and correct in all material respects, are not being offered in a manner involving a public offering under, or in a distribution in violation of, the Securities Act, or any applicable state securities laws.
- 2.1.11 Subscriber understands and agrees that no federal or state agency has passed upon or endorsed the merits of the offering of the Subscribed Shares or made any findings or determination as to the fairness of an investment in the Subscribed Shares.
- 2.1.12 Subscriber represents and warrants that Subscriber is not (i) a person or entity named on the List of Specially Designated Nationals and

Blocked Persons administered by the U.S. Treasury Department's Office of Foreign Assets Control ("**OFAC**") or in any Executive Order issued by the President of the United States and administered by OFAC ("**OFAC List**"), or a person or entity prohibited by any OFAC sanctions program, (ii) a Designated National as defined in the Cuban Assets Control Regulations, 31 C.F.R. Part 515 or (iii) a non-U.S. shell bank or providing banking services indirectly to a non-U.S. shell bank. Subscriber agrees to provide law enforcement agencies, if requested thereby, such records as required by applicable law, provided that Subscriber is permitted to do so under applicable law. If Subscriber is a financial institution subject to the Bank Secrecy Act (31 U.S.C. Section 5311 et seq.), as amended by the USA PATRIOT Act of 2001, and its implementing regulations (collectively, the "**BSA/PATRIOT Act**"), Subscriber represents that it maintains policies and procedures reasonably designed to comply with applicable obligations under the BSA/PATRIOT Act. Subscriber also represents that, to the extent required, it maintains policies and procedures reasonably designed for the screening of its investors against the OFAC sanctions programs, including the OFAC List. Subscriber further represents and warrants that, to the extent required, it maintains policies and procedures reasonably designed to ensure that the funds held by Subscriber and used to purchase the Subscribed Shares were legally derived.

2.1.13 If Subscriber is an employee benefit plan that is subject to Title I of ERISA, a plan, an individual retirement account or other arrangement that is subject to section 4975 of the Code or an employee benefit plan that is a governmental plan (as defined in section 3(32) of ERISA), a church plan (as defined in section 3(33) of ERISA), a non-U.S. plan (as described in section 4(b)(4) of ERISA) or other plan that is not subject to the foregoing but may be subject to provisions under any other Similar Laws or an entity whose underlying assets are considered to include "plan assets" of any such plan, account or arrangement (each, a "Plan"), Subscriber represents and warrants that none of the Issuer, the SPAC nor any of their respective affiliates (the "Transaction Parties") has acted as the Plan's fiduciary, or has been relied on for advice, with respect to its decision to acquire and hold the Subscribed Shares, and none of the Transaction Parties shall at any time be relied upon as the Plan's fiduciary with respect to any decision to acquire, continue to hold or transfer the Subscribed Shares.

2.1.14 Except as expressly disclosed in a Schedule 13D or Schedule 13G (or amendments thereto) filed by such Subscriber with the United States Securities and Exchange Commission (the "Commission") with respect to the beneficial ownership of the SPAC's common stock, Subscriber is not currently (and at all times through Closing will refrain from being or becoming) a member of a "group" (within the meaning of Section 13(d)(3) or Section 14(d)(2) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or any successor provision) acting for the purpose of acquiring, holding or disposing of equity securities of the Issuer or the SPAC (within the meaning of Rule 13d-5(b)(1) under the Exchange Act).

- 2.1.15 Subscriber is not a foreign person (as defined in 31 C.F.R. Part 800.224) in which the national or subnational governments of a single foreign state have a substantial interest (as defined in 31 C.F.R. Part 800.244) and that will acquire a substantial interest in the Issuer as a result of the purchase and sale of Subscribed Shares hereunder such that a declaration to the Committee on Foreign Investment in the United States would be mandatory under 31 C.F.R. Part 800.401, and no foreign person will have control (as defined in 31 C.F.R. Part 800.208) over the Issuer from and after the Closing as a result of the purchase and sale of the Subscribed Shares hereunder.
- 2.1.16 Subscriber has, and on each date the Purchase Price would be required to be funded to the Issuer pursuant to <u>Section 3.1</u> will have, sufficient immediately available funds to pay the Purchase Price pursuant to <u>Section 3.1</u>. Subscriber was not formed for the purpose of acquiring the Subscribed Shares.
- 2.1.17 No broker, finder or other financial consultant has acted on behalf of Subscriber in connection with this Subscription Agreement or the transactions contemplated hereby in such a way as to create any liability on the Issuer or the SPAC.
- 2.1.18 Subscriber agrees that, from the date of this Subscription Agreement until the Closing or the earlier termination of this Subscription Agreement, none of Subscriber, its controlled affiliates, or any person or entity acting on behalf of Subscriber or any of its controlled affiliates or pursuant to any understanding with Subscriber or any of its controlled affiliates will engage in any Short Sales with respect to securities of the SPAC. For the purposes hereof, "Short Sales" shall include, without limitation, all "short sales" as defined in Rule 200 promulgated under Regulation SHO under the Exchange Act, and all types of direct and indirect stock pledges (other than pledges in the ordinary course of business as part of prime brokerage arrangements), forward sale contracts, options, puts, calls, swaps and similar arrangements (including on a total return basis), including through non-U.S. broker dealers or foreign regulated brokers.
- 2.2 <u>Issuer's Representations, Warranties and Agreements</u>. To induce Subscriber to purchase the Subscribed Shares, the Issuer hereby represents and warrants to Subscriber and agrees with Subscriber, as of the date hereof and as of the Closing Date, as follows:
 - 2.2.1 The Issuer is validly existing and in good standing under the laws of Bermuda (by which we mean that the Issuer has paid all fees due to the Bermuda Government which are currently required in order for the Issuer to maintain its existence in Bermuda), with all requisite power and authority to own, lease and operate its properties and conduct its business as presently conducted and to enter into, deliver and perform its obligations under this Subscription Agreement.
 - 2.2.2 The Subscribed Shares have been duly authorized and, when issued and delivered to Subscriber against full payment for the Subscribed Shares in accordance with the terms of this Subscription Agreement and registered with the Issuer's transfer agent, the Subscribed Shares will be validly issued, fully paid and non-assessable and will not have been issued in violation of or subject to any preemptive or similar rights whether created under the Issuer's bye-laws or similar constitutive agreements or under the Companies Act 1981 of Bermuda, as amended.

- 2.2.3 This Subscription Agreement has been duly authorized, validly executed and delivered by the Issuer and, assuming that this Subscription Agreement constitutes the valid and binding obligation of the other signatories hereto, is the valid and binding obligation of the Issuer, is enforceable against Issuer in accordance with its terms, except as may be limited or otherwise affected by (i) bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or other laws relating to or affecting the rights of creditors generally and (ii) principles of equity, whether considered at law or equity.
- 2.2.4 The execution, delivery and performance of this Subscription Agreement (including compliance by the Issuer with all of the provisions hereof), the issuance and sale of the Subscribed Shares and the consummation of the other transactions contemplated herein will not (i) conflict with or result in a breach or violation of any of the terms or provisions of, or constitute a default under, or result in the creation or imposition of any lien, charge or encumbrance upon any of the property or assets of the Issuer or any of its subsidiaries pursuant to the terms of any indenture, mortgage, charge, deed of trust, loan agreement, lease, license or other agreement or instrument to which the Issuer or any of its subsidiaries is a party or by which the Issuer or any of its subsidiaries is bound or to which any of the property or assets of the Issuer or any of its subsidiaries is subject, which would reasonably be expected to have a material adverse effect on the validity of the Common Shares or the legal authority of the Issuer to enter into and timely perform its obligations under this Subscription Agreement (collectively, an "Issuer Material Adverse Effect"), (ii) result in any violation of the provisions of the organizational documents of the Issuer or any of its subsidiaries or (iii) result in any violation of any statute or any judgment, order, rule or regulation of any court or governmental agency or body, domestic or foreign, having jurisdiction over the Issuer or any of its subsidiaries or any of its properties that would reasonably be expected to have an Issuer Material Adverse Effect.
- 2.2.5 Neither the Issuer, nor any person acting on its behalf has, directly or indirectly, made any offers or sales of any security of the Issuer nor solicited any offers to buy any security under circumstances that would adversely affect reliance by the Issuer on Section 4(a)(2) of the Securities Act for the exemption from registration for the transactions contemplated hereby or would require registration of the issuance of the Subscribed Shares under the Securities Act.
- 2.2.6 Neither the Issuer, nor any person acting on its behalf has conducted any general solicitation or general advertising, including methods described in section 502(c) of Regulation D under the Securities Act, in connection with the offer or sale of any of the Subscribed Shares and neither the Issuer, nor any person acting on its behalf has offered any of the Subscribed Shares in a manner involving a public offering under, or in a distribution in violation of, the Securities Act or any state securities laws.

2.2.7 Concurrently with the execution and delivery of this Subscription Agreement, the Issuer is entering into the Other Subscription Agreements providing for the sale of an aggregate of 200,000,000 Common Shares for an aggregate purchase price of \$2,000,000,000 (including the Subscribed Shares purchased and sold under this Subscription Agreement) (collectively, the "PIPE Securities"). There are no Other Subscription Agreements, side letter agreements or other agreements or understandings (including written summaries of any oral understandings) with any Other Subscriber or any other investor or potential investor with respect to the purchase of securities of the Issuer or the SPAC (other than pursuant to the Forward Purchase Agreement or the Merger Agreement) (collectively, the "PIPE Agreements") which include terms and conditions that are materially more advantageous to any such Other Subscriber, investor or potential investor (as compared to Subscriber) other than PIPE Agreements with certain Other Subscribers with pre-existing relationships with the Founder solely to the extent such PIPE Agreements provide for a cash fee to such Other Subscribers in an amount equal to the fees that would have otherwise been payable by the SPAC to the Placement Agents if such Other Subscribers did not have the pre-existing relationship with the Founder, but is not payable by the SPAC to the Placement Agents as a result of such pre-existing relationship with the Founder. The Other Subscription Agreements have not been amended or modified in any material respect following the date of this Subscription Agreement.

2.2.8 As of the date of this Subscription Agreement, the authorized share capital of the Issuer consists of 1,000 Common Shares, 1,000 of which are issued and outstanding. All issued and outstanding Common Shares have been duly authorized and validly issued, are fully paid, non-assessable and are not subject to preemptive rights. Except as set forth above and pursuant to the Other Subscription Agreements and the Merger Agreement (a true and correct copy of which has been provided to Subscriber), there are no outstanding options, warrants or other rights to subscribe for, purchase or acquire from the Issuer any Common Shares, or any other equity interests in the Issuer, or securities convertible into or exchangeable or exercisable for such equity interests. There are no shareholder agreements, voting trusts or other agreements or understandings to which the Issuer is a party or by which it is bound relating to the voting of any securities of the Issuer, other than as contemplated by the Merger Agreement and the Transaction Agreements (as defined in the Merger Agreement).

2.2.9 Assuming the accuracy of Subscriber's representations and warranties set forth in Section 2.1 of this Subscription Agreement, (i) no registration under the Securities Act is required for the offer and sale of the Subscribed Shares by the Issuer to Subscriber and (ii) no consent, approval, order or authorization of, or registration, qualification, designation, declaration or filing with, any federal, state or local Governmental Authority is required on the part of the Issuer in connection with the consummation of the transactions contemplated by this Subscription Agreement, except for filings pursuant to Regulation D of the Securities Act and applicable state securities laws and filings required to consummate the Transactions as provided under the Merger Agreement.

- 2.2.10 As of the date hereof, there are no pending or, to the knowledge of the Issuer, threatened, Actions, which, if determined adversely, would, individually or in the aggregate, reasonably be expected to have an Issuer Material Adverse Effect. As of the date hereof, there is no unsatisfied judgment or any open injunction binding upon the Issuer, which would, individually or in the aggregate, reasonably be expected to have an Issuer Material Adverse Effect.
- 2.2.11 The Issuer is in compliance with all applicable laws, except where such non-compliance would not reasonably be expected to have an Issuer Material Adverse Effect. The Issuer has not received any written communication from a governmental entity that alleges that the Issuer is not in compliance with or is in default or violation of any applicable law, except where such non-compliance, default or violation would not, individually or in the aggregate, be reasonably expected to have an Issuer Material Adverse Effect.
- 2.2.12 The Issuer is not required to obtain any consent, waiver, authorization or order of, give any notice to, or make any filing or registration with, any court or other federal, state, local or other governmental authority, self-regulatory organization or other person in connection with the execution, delivery and performance by the Issuer of this Subscription Agreement (including, without limitation, the issuance of the Subscribed Shares), other than (i) filings with the Commission, (ii) filings required by applicable state securities laws, (iii) filings required in accordance with Section 4, (iv) those required by the New York Stock Exchange (the "NYSE") or Nasdaq, and (v) filings, the failure of which to obtain would not be reasonably be expected to have, individually or in the aggregate, an Issuer Material Adverse Effect.
- 2.2.13 Immediately following the Closing, the SPAC will be a wholly-owned subsidiary of the Issuer and there will be no outstanding options, warrants or other rights to subscribe for, purchase or acquire from the SPAC any equity interests in the SPAC, or securities convertible into or exchangeable or exercisable for such equity interests.
- 2.2.14 No broker, finder or other financial consultant has acted on behalf of the Issuer in connection with this Subscription Agreement or the transactions contemplated hereby in such a way as to create any liability on Subscriber.
 - 2.2.15 The Issuer is classified as a Subchapter C corporation for U.S. federal income tax purposes.
- 2.2.16 The Issuer acknowledges that, notwithstanding anything herein to the contrary, the Subscribed Shares may be pledged by Subscriber in connection with a bona fide margin agreement, provided such pledge shall be (i) pursuant to an available exemption from the registration requirements of the Securities Act or (ii) pursuant to, and in accordance with, a registration statement that is effective under the Securities Act at the time of such pledge, and Subscriber effecting a pledge of Subscribed Shares shall not be required to provide Issuer with any notice thereof; provided, however, that neither the Issuer or its counsel shall be required to take any action (or refrain from taking any action) in connection with any such pledge.

- 2.3 <u>SPAC's Representations, Warranties and Agreements</u>. To induce Subscriber to purchase the Subscribed Shares, the SPAC, hereby represents and warrants to Subscriber and agrees with Subscriber, as of the date hereof and as of the Closing Date, as follows:
 - 2.3.1 The SPAC has been duly incorporated and is validly existing as a corporation in good standing under the laws of the Delaware General Corporation Law, with corporate power and authority to own, lease and operate its properties and conduct its business as presently conducted and to enter into, deliver and perform its obligations under this Subscription Agreement.
 - 2.3.2 This Subscription Agreement has been duly authorized, validly executed and delivered by the SPAC and, assuming that this Subscription Agreement constitutes the valid and binding obligation of the other signatories hereto, is the valid and binding obligation of the SPAC, is enforceable against the SPAC in accordance with its terms, except as may be limited or otherwise affected by (i) bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or other laws relating to or affecting the rights of creditors generally and (ii) principles of equity, whether considered at law or equity.
 - 2.3.3 The SPAC made available to Subscriber (including via the Commission's EDGAR system) a true, correct and complete copy of each form, report, statement, schedule, prospectus, proxy, registration statement and other documents filed by the SPAC with the Commission prior to the date of this Subscription Agreement (the "SEC Documents"). None of the SEC Documents filed under the Exchange Act, contained, when filed or, if amended prior to the date of this Subscription Agreement, as of the date of such amendment with respect to those disclosures that are amended, any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading; provided that the SPAC makes no such representation or warranty with respect to the registration statement on Form F-4 to be filed by the Issuer with respect to the Transactions (the "F-4 Registration Statement") or any other information relating to Paysafe or any of its affiliates included in any SEC Document or filed as an exhibit thereto. The SPAC has timely filed each report, statement, schedule, prospectus, and registration statement that the SPAC was required to file with the Commission since its inception and through the date hereof. As of the date hereof, there are no material outstanding or unresolved comments in comment letters from the Commission staff with respect to any of the SEC Documents.

- 2.3.4 The execution, delivery and performance of this Subscription Agreement and the consummation of the other transactions contemplated herein will not (i) conflict with or result in a breach or violation of any of the terms or provisions of, or constitute a default under, or result in the creation or imposition of any lien, charge or encumbrance upon any of the property or assets of the SPAC or any of its subsidiaries pursuant to the terms of any indenture, mortgage, charge, deed of trust, loan agreement, lease, license or other agreement or instrument to which the SPAC or any of its subsidiaries is a party or by which the SPAC or any of its subsidiaries is bound or to which any of the property or assets of the SPAC or any of its subsidiaries is subject, which would reasonably be expected to have a material adverse effect on the legal authority of the SPAC to enter into and timely perform its obligations under this Subscription Agreement (collectively, a "SPAC Material Adverse Effect"), (ii) result in any violation of the provisions of the organizational documents of the SPAC or any of its subsidiaries or (iii) result in any violation of any statute or any judgment, order, rule or regulation of any court or governmental agency or body, domestic or foreign, having jurisdiction over the SPAC or any of its subsidiaries or any of its properties that would reasonably be expected to have a SPAC Material Adverse Effect.
- 2.3.5 As of the date hereof, there are no pending or, to the knowledge of the SPAC, threatened, Actions, which, if determined adversely, would, individually or in the aggregate, reasonably be expected to have a SPAC Material Adverse Effect. As of the date hereof, there is no unsatisfied judgment or any open injunction binding upon the SPAC which would, individually or in the aggregate, reasonably be expected to have a SPAC Material Adverse Effect.
- 2.3.6 The SPAC is in compliance with all applicable laws, except where such non-compliance would not reasonably be expected to have a SPAC Material Adverse Effect. The SPAC has not received any written communication from a governmental entity that alleges that the SPAC is not in compliance with or is in default or violation of any applicable law, except where such non-compliance, default or violation would not, individually or in the aggregate, be reasonably expected to have a SPAC Material Adverse Effect.
- 2.3.7 The SPAC is not required to obtain any consent, waiver, authorization or order of, give any notice to, or make any filing or registration with, any court or other federal, state, local or other governmental authority, self-regulatory organization or other person in connection with the execution, delivery and performance by the SPAC of this Subscription Agreement, other than (i) filings with the Commission, (ii) filings required by applicable state securities laws, (iii) filings required in accordance with Section 7, (iv) those required by the NYSE or Nasdaq, and (v) filings, the failure of which to obtain would not be reasonably be expected to have, individually or in the aggregate, a SPAC Material Adverse Effect.
- 2.3.8 Immediately following the Closing, the SPAC will be a wholly-owned subsidiary of the Issuer and there will be no outstanding options, warrants or other rights to subscribe for, purchase or acquire from the SPAC any equity interests in the SPAC, or securities convertible into or exchangeable or exercisable for such equity interests.

2.3.9 No broker, finder or other financial consultant has acted on behalf of the SPAC in connection with this Subscription Agreement or the transactions contemplated hereby in such a way as to create any liability on Subscriber.

3. Settlement Date and Delivery.

3.1 Closing. The closing of the Subscription contemplated hereby (the "Closing") shall occur on the date of, and immediately prior to (but subject to), the consummation of the Transactions (the date of the Closing, the "Closing Date"). Upon written notice from (or on behalf of) the Issuer and the SPAC to Subscriber (the "Closing Notice") at least ten (10) Business Days prior to the date that the Issuer and the SPAC reasonably expect all conditions to the closing of the Transactions to be satisfied (the "Expected Closing Date"), Subscriber shall deliver to the Issuer no later than three (3) Business Days prior to the Expected Closing Date, the Purchase Price for the Subscribed Shares, by wire transfer of United States dollars in immediately available funds to the account specified by the Issuer and the SPAC in the Closing Notice, such funds to be held by the Issuer in escrow until the Closing. If the Transactions are not consummated on or prior to the fifth (5th) Business Day after the Expected Closing Date, the Issuer shall promptly (but no later than two (2) Business Days thereafter) return the Purchase Price to Subscriber by wire transfer of United States dollars in immediately available funds to an account specified by Subscriber. Notwithstanding such return, (i) a failure to close on the Expected Closing Date shall not, by itself, be deemed to be a failure of any of the conditions to Closing set forth in this Section 3 to be satisfied or waived on or prior to the Closing Date, and (ii) Subscriber shall remain obligated (A) to redeliver funds to the Issuer following the Issuer's delivery to Subscriber of a new Closing Notice and (B) to consummate the Closing upon satisfaction of the conditions set forth in this Section 3. At the Closing, upon satisfaction (or, if applicable, waiver) of the conditions set forth in this Section 3, the Issuer shall issue to Subscriber (or the funds and accounts designated by Subscriber if so designated by Subscriber, or its nominee in accordance with its delivery instructions) or to a custodian designated by Subscriber, as applicable the Subscribed Shares, free and clear of any liens or other restrictions whatsoever (other than those arising under state or federal securities laws), which Subscribed Shares, unless otherwise determined by the Issuer, shall be uncertificated, with record ownership reflected only in the register of shareholders of the Issuer (a copy of which showing Subscriber as the owner of the Subscribed Shares on and as of the Closing Date shall be provided to Subscriber on the Closing Date or promptly thereafter). For purposes of this Subscription Agreement, "Business Day" means any day that, in New York, New York, is neither a legal holiday nor a day on which banking institutions are generally authorized or required by law or regulation to close.

3.2 Conditions to Closing of the Issuer.

The Issuer's obligations to sell and issue the Subscribed Shares at the Closing are subject to the fulfillment or (to the extent permitted by applicable law) written waiver by the Issuer, on or prior to the Closing Date, of each of the following conditions:

- 3.2.1 Representations and Warranties Correct. The representations and warranties made by Subscriber in Section 2.1 hereof shall be true and correct in all material respects when made (other than representations and warranties that are qualified as to materiality or Subscriber Material Adverse Effect, which representations and warranties shall be true and correct in all respects), and shall be true and correct in all material respects on and as of the Closing Date (unless they specifically speak as of another date in which case they shall be true and correct in all material respects as of such date) (other than representations and warranties that are qualified as to materiality or Subscriber Material Adverse Effect, which representations and warranties shall be true in all respects) with the same force and effect as if they had been made on and as of said date, but in each case without giving effect to consummation of the Transactions.
- 3.2.2 <u>Compliance with Covenants</u>. Subscriber shall have performed, satisfied and complied in all material respects with the covenants, agreements and conditions required by this Subscription Agreement to be performed, satisfied or complied with by Subscriber at or prior to the Closing.
- 3.2.3 <u>Closing of the Transactions</u>. All conditions precedent to each of the Issuer's, Paysafe's and the SPAC's obligations to consummate, or cause to be consummated, the Transactions set forth in the Merger Agreement shall have been satisfied or waived by the party entitled to the benefit thereof under the Merger Agreement (other than those conditions that may only be satisfied at the consummation of the Transactions, but subject to satisfaction or waiver by such party of such conditions as of the consummation of the Transactions), and the Transactions will be consummated immediately following the Closing.
- 3.2.4 <u>Legality</u>. There shall not be in force any order, judgment, injunction, decree, writ, stipulation, determination or award, in each case, entered by or with any Governmental Authority, statute, rule or regulation enjoining or prohibiting the consummation of the Subscription.

3.3 Conditions to Closing of Subscriber.

Subscriber's obligation to purchase the Subscribed Shares at the Closing is subject to the fulfillment or (to the extent permitted by applicable law) written waiver by Subscriber, on or prior to the Closing Date, of each of the following conditions:

3.3.1 Representations and Warranties Correct. The representations and warranties made by the Issuer in Section 2.2 hereof, and the SPAC in Section 2.3 hereof, shall be true and correct in all material respects when made (other than representations and warranties that are qualified as to materiality or Issuer Material Adverse Effect, which representations and warranties shall be true and correct in all material respects on and as of the Closing Date (unless they specifically speak as of another date in which case they shall be true and correct in all material respects as of such date) (other than representations and warranties that are qualified as to materiality or Issuer Material Adverse Effect, which representations and warranties shall be true and correct in all respects) with the same force and effect as if they had been made on and as of said date, but in each case

without giving effect to consummation of the Transactions; <u>provided</u>, that in the event this condition would otherwise fail to be satisfied as a result of a breach of one or more of the representations and warranties of the Issuer contained in this Subscription Agreement and the facts underlying such breach would also cause a condition to SPAC's obligations under the Merger Agreement to fail to be satisfied, this condition shall nevertheless be deemed satisfied in the event SPAC waives such condition with respect to such breach under the Merger Agreement.

- 3.3.2 <u>Compliance with Covenants</u>. The Issuer shall have performed, satisfied and complied in all material respects with the covenants, agreements and conditions required by this Subscription Agreement to be performed, satisfied or complied with by the Issuer at or prior to the Closing, except where the failure of such performance or compliance would not or would not reasonably be expected to prevent, materially delay, or materially impair the ability of the Issuer to consummate the Closing.
- 3.3.3 <u>Closing of the Transactions</u>. (i) All conditions precedent to the consummation of the Transactions set forth in the Merger Agreement shall have been satisfied or waived by the party entitled to the benefit thereof under the Merger Agreement (other than those conditions that may only be satisfied at the consummation of the Transactions, but subject to satisfaction or waiver by such party of such conditions as of the consummation of the Transactions), (ii) no amendment, modification or waiver of the Merger Agreement (as the same exists on the date hereof as provided to Subscriber) or any terms thereof shall have occurred that would reasonably be expected to materially and adversely affect the economic benefits that Subscriber would reasonably expect to receive under this Subscription Agreement without having received Subscriber's prior written consent (not to be unreasonably withheld, conditioned or delayed) and (iii) the Transactions will be consummated immediately following the Closing.
- 3.3.4 <u>Legality</u>. There shall not be in force any order, judgment, injunction, decree, writ, stipulation, determination or award, in each case, entered by or with any governmental authority, statute, rule or regulation enjoining or prohibiting the transactions contemplated by this Subscription Agreement.

4. Registration Statement.

4.1 The Issuer agrees that, within thirty (30) calendar days after the consummation of the Transactions (the "Filing Date"), the Issuer will file with the Commission (at the Issuer's sole cost and expense) a registration statement (the "Registration Statement") registering the resale of the Subscribed Shares (the "Registrable Securities"), and the Issuer shall use its commercially reasonable efforts to have the Registration Statement declared effective as soon as practicable after the filing thereof, but no later than the earlier of (i) the 75th calendar day (or 135th calendar day if the Commission notifies the Issuer that it will "review" the Registration Statement) following the Closing Date and (ii) the 5th Business Day after the date the Issuer is notified (orally or in writing, whichever is earlier) by the Commission that the Registration Statement will not be "reviewed" or will not be subject to further review (such

earlier date, the "Effectiveness Date"); provided, however, that the Issuer's obligations to include the Registrable Securities in the Registration Statement are contingent upon Subscriber furnishing a completed and executed selling shareholders questionnaire in customary form to the Issuer that contains the information required by Commission rules for a Registration Statement regarding Subscriber, the securities of the Issuer held by Subscriber and the intended method of disposition of the Registrable Securities to effect the registration of the Registrable Securities, and Subscriber shall execute such documents in connection with such registration as the Issuer may reasonably request that are customary of a selling stockholder in similar situations, including providing that the Issuer shall be entitled to postpone and suspend the effectiveness or use of the Registration Statement, if applicable, during any customary blackout or similar period or as permitted hereunder; provided, that Subscriber shall not in connection with the foregoing be required to execute any lock-up or similar agreement or otherwise be subject to any contractual restriction on the ability to transfer the Registrable Securities. For purposes of clarification, any failure by the Issuer to file the Registration Statement by the Filing Date or to effect such Registration Statement by the Effectiveness Date shall not otherwise relieve the Issuer of its obligations to file or effect the Registration Statement as set forth above in this Section 4. For purposes of this Section 4, Registrable Securities shall include, as of any date of determination, the Subscribed Shares and any other equity security of the Issuer issued or issuable with respect to the Subscribed Shares by way of share split, dividend, distribution, recapitalization, merger, exchange, replacement or similar event or otherwise.

- 4.2 In the case of the registration effected by the Issuer pursuant to this Subscription Agreement, the Issuer shall, upon reasonable request, inform Subscriber as to the status of such registration. At its expense the Issuer shall:
 - 4.2.1 except for such times as the Issuer is permitted hereunder to suspend the use of the prospectus forming part of a Registration Statement, use its commercially reasonable efforts to keep such registration, and any qualification, exemption or compliance under state securities laws which the Issuer determines to obtain, continuously effective with respect to Subscriber, and to keep the applicable Registration Statement or any subsequent shelf registration statement free of any material misstatements or omissions, until the earlier of the following: (i) Subscriber ceases to hold any Registrable Securities and (ii) the date all Registrable Securities held by Subscriber may be sold without restriction under Rule 144, including without limitation, any volume and manner of sale restrictions which may be applicable to affiliates under Rule 144 and without the requirement for the Issuer to be in compliance with the current public information required under Rule 144(c)(1) (or Rule 144(i)(2), if applicable);
 - 4.2.2 advise Subscriber, as promptly as practicable but in any event within five (5) Business Days:
 - (a) when a Registration Statement or any post-effective amendment thereto has become effective;

- (b) of the issuance by the Commission of any stop order suspending the effectiveness of any Registration Statement or the initiation of any proceedings for such purpose;
- (c) of the receipt by the Issuer of any notification with respect to the suspension of the qualification of the Registrable Securities included therein for sale in any jurisdiction or the initiation or threatening of any proceeding for such purpose; and
- (d) subject to the provisions in this Subscription Agreement, of the occurrence of any event that requires the making of any changes in any Registration Statement or prospectus so that, as of such date, the statements therein are not misleading and do not omit to state a material fact required to be stated therein or necessary to make the statements therein (in the case of a prospectus, in the light of the circumstances under which they were made) not misleading.

Notwithstanding anything to the contrary set forth herein, the Issuer shall not, when so advising Subscriber of such events, provide Subscriber with any material, nonpublic information regarding the Issuer other than to the extent that providing notice to Subscriber of the occurrence of the events listed in (a) through (d) above constitutes material, nonpublic information regarding the Issuer;

- 4.2.3 use its commercially reasonable efforts to obtain the withdrawal of any order suspending the effectiveness of any Registration Statement as soon as reasonably practicable;
- 4.2.4 upon the occurrence of any event contemplated in Section 4.2.2(d), except for such times as the Issuer is permitted hereunder to suspend, and has suspended, the use of a prospectus forming part of a Registration Statement, the Issuer shall use its commercially reasonable efforts to as soon as reasonably practicable prepare a post-effective amendment to such Registration Statement or a supplement to the related prospectus, or file any other required document so that, as thereafter delivered to purchasers of the Registrable Securities included therein, such prospectus will not include any untrue statement of a material fact or omit to state any material fact necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading; and
- 4.2.5 use its commercially reasonable efforts to cause all Subscribed Shares to be listed on each securities exchange or market, if any, on which the Issuer's Common Shares are then listed.
- 4.3 Notwithstanding anything to the contrary in this Subscription Agreement, the Issuer shall be entitled to delay or postpone the effectiveness of the Registration Statement, and from time to time to require Subscriber not to sell under the Registration Statement or to suspend the effectiveness thereof, if the filing, effectiveness or continued use of any Registration Statement would require the Issuer to make any public disclosure of material non-public information, which disclosure, in the good faith determination of the board of directors of the Issuer, after consultation with counsel to the Issuer, (a) would be required to be made in any

Registration Statement in order for the applicable Registration Statement not to contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements contained therein not misleading, (b) would not be required to be made at such time if the Registration Statement were not being filed, and (c) the Issuer has a bona fide business purpose for not making such information public (each such circumstance, a "Suspension Event"); provided, however, that the Issuer may not delay or suspend the Registration Statement on more than two occasions or for more than sixty (60) consecutive calendar days, or more than one hundred and twenty (120) total calendar days, in each case during any twelve-month period. Upon receipt of any written notice from the Issuer of the happening of any Suspension Event during the period that the Registration Statement is effective or if as a result of a Suspension Event the Registration Statement or related prospectus contains any untrue statement of a material fact or omits to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made (in the case of the prospectus) not misleading, Subscriber agrees that (i) it will immediately discontinue offers and sales of the Subscribed Shares under the Registration Statement (excluding, for the avoidance of doubt, sales conducted pursuant to Rule 144) until Subscriber receives copies of a supplemental or amended prospectus (which the Issuer agrees to promptly prepare) that corrects the misstatement(s) or omission(s) referred to above and receives notice that any post-effective amendment has become effective or unless otherwise notified by the Issuer that it may resume such offers and sales, and (ii) it will maintain the confidentiality of any information included in such written notice delivered by the Issuer except (A) for disclosure to Subscriber's employees, agents and professional advisers who need to know such information and are obligated to keep it confidential, (B) for disclosures to the extent required in order to comply with reporting obligations to its limited partners who have agreed to keep such information confidential and (C) as required by law. If so directed by the Issuer, Subscriber will deliver to the Issuer or, in Subscriber's sole discretion destroy, all copies of the prospectus covering the Subscribed Shares in Subscriber's possession; provided, however, that this obligation to deliver or destroy all copies of the prospectus covering the Subscribed Shares shall not apply (i) to the extent Subscriber is required to retain a copy of such prospectus (a) in order to comply with applicable legal, regulatory, self-regulatory or professional requirements or (b) in accordance with a bona fide pre-existing document retention policy or (ii) to copies stored electronically on archival servers as a result of automatic data back-up.

4.4 The parties agree that:

4.4.1 The Issuer agrees to indemnify and hold harmless, to the extent permitted by law, Subscriber (to the extent a seller under the Registration Statement), its directors, officers, employees, and agents, and each person who controls Subscriber (within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act) from and against any and all out-of-pocket losses, claims, damages, liabilities and expenses (including, without limitation, any reasonable attorneys' fees and expenses incurred in connection with defending or investigating any such action or claim) (collectively, "Losses"), as incurred, that arise out of or are based upon any untrue or alleged untrue statement of material fact contained in any Registration Statement, prospectus included in any Registration Statement or preliminary prospectus or any amendment thereof or supplement thereto or arising out of or relating to any omission or alleged omission to state a material fact required to be

stated therein or necessary to make the statements therein (in the case of any prospectus or form of prospectus or supplement thereto, in light of the circumstances under which they were made) not misleading, except insofar as the same are caused by or contained in any information furnished in writing to the Issuer by or on behalf of Subscriber expressly for use therein or Subscriber has omitted a material fact from such information or otherwise violated the Securities Act, Exchange Act or any state securities law or any other law, rule or regulation thereunder; provided, however, that the indemnification contained in this Section 4.4 shall not apply to amounts paid in settlement of any Losses if such settlement is effected without the consent of the Issuer (which consent shall not be unreasonably withheld, conditioned or delayed), nor shall the Issuer be liable for any Losses to the extent they arise out of or are based upon a violation which occurs (A) in reliance upon and in conformity with written information furnished by Subscriber, (B) in connection with any failure of such person to deliver or cause to be delivered a prospectus made available by the Issuer in a timely manner, (C) as a result of offers or sales effected by or on behalf of any person by means of a "free writing prospectus" (as defined in Rule 405 under the Securities Act) that was not authorized in writing by the Issuer, or (D) in connection with any offers or sales effected by or on behalf of Subscriber in violation of Section 4.3 hereof. The Issuer shall notify Subscriber promptly of the institution, threat or assertion of any proceeding arising from or in connection with the transactions contemplated by this Section 4 of which the Issuer is aware.

4.4.2 Subscriber agrees, severally and not jointly with any person that is a party to the Other Subscription Agreements, to indemnify and hold harmless, to the extent permitted by law, the Issuer, its directors, officers, employees and agents and each person who controls the Issuer (within the meaning of Section 15 of the Securities Act and Section 20 of the Exchange Act) against any and all Losses, as incurred, that arise out of or are based upon any untrue or alleged untrue statement of material fact contained in any Registration Statement, prospectus included in any Registration Statement or preliminary prospectus or any amendment thereof or supplement thereto or arising out of or relating to any omission of a material fact required to be stated therein or necessary to make the statements therein (in the case of any prospectus or form of prospectus or supplement thereto, in light of the circumstances under which they were made) not misleading, but only to the extent that such untrue statement or omission is contained in any information or affidavit so furnished in writing by Subscriber expressly for use therein; provided, however, that the indemnification contained in this Section 4.4 shall not apply to amounts paid in settlement of any Losses if such settlement is effected without the consent of Subscriber (which consent shall not be unreasonably withheld, conditioned or delayed). Notwithstanding anything to the contrary herein, in no event shall the liability of Subscriber be greater in amount than the dollar amount of the net proceeds received by Subscriber upon the sale of the Subscribed Shares purchased pursuant to this Subscription Agreement giving rise to such indemnification obligation.

4.4.3 Any person entitled to indemnification herein shall (1) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (provided that the failure to give prompt notice shall not impair any person's right to indemnification hereunder to the extent such failure has not prejudiced the indemnifying party) and (2) permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party shall not be subject to any liability for any settlement made by the indemnified party without its consent. An indemnifying party who elects not to assume the defense of a claim shall not be obligated to pay the fees and expenses of more than one counsel for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of legal counsel to any indemnified party a conflict of interest exists between such indemnified party and any other of such indemnified parties with respect to such claim. No indemnifying party shall, without the consent of the indemnified party (which consent shall not be unreasonably withheld, conditioned or delayed), consent to the entry of any judgment or enter into any settlement which cannot be settled in all respects by the payment of money (and such money is so paid by the indemnifying party pursuant to the terms of such settlement) or which settlement does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.

4.4.4 The indemnification provided for under this Subscription Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party and shall survive the transfer of the Subscribed Shares purchased pursuant to this Subscription Agreement.

4.4.5 If the indemnification provided under this Section 4.4 from the indemnifying party is unavailable or insufficient to hold harmless an indemnified party in respect of any Losses referred to herein, then the indemnifying party, in lieu of indemnifying the indemnified party, shall contribute to the amount paid or payable by the indemnified party as a result of such losses, claims, damages, liabilities and expenses in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party, as well as any other relevant equitable considerations. The relative fault of the indemnifying party and indemnified party shall be determined by reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact, was made by, or relates to information supplied by, such indemnifying party or indemnified party, and the indemnifying party's and indemnified party's relative intent, knowledge, access to information and opportunity to correct or prevent such action. The amount paid or payable by a party as a result of the Losses or other liabilities referred to above shall be deemed to include, subject to the limitations set forth above, any legal or other fees, charges or expenses reasonably incurred by such party in connection with any investigation or proceeding. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution pursuant to this Section 4.4 from any person who was not guilty of such fraudulent misrepresentation. In no event shall the liability of Subscriber be greater in amount than the dollar amount of the net proceeds received by Subscriber upon the sale of the Subscribed Shares purchased pursuant to this Subscription Agreement giving rise to such contribution obligation.

5. Termination. This Subscription Agreement shall terminate and be void and of no further force and effect, and all rights and obligations of the parties hereunder shall terminate without any further liability on the part of any party in respect thereof, upon the earliest to occur of (i) such date and time as the Merger Agreement is validly terminated in accordance with its terms, (ii) upon the mutual written agreement of each of the parties hereto to terminate this Subscription Agreement and (iii) at Subscriber's election, on or after December 7, 2021 (the "Termination Date"), if the Closing has not occurred by such date, provided, that if any Action for specific performance or other equitable relief by PGHL or the Issuer with respect to the Merger Agreement, any other Transaction Agreement, or otherwise with respect to the Transactions is commenced or pending on or before the Termination Date, then the Termination Date shall be automatically extended without any further action by any party until the date that is 30 days following the date on which a final, non-appealable Governmental Order has been entered with respect to such Action and the Termination Date shall be deemed to be such later date for all purposes of this Agreement; provided that nothing herein will relieve any party from liability for any willful breach hereof prior to the time of termination, and each party will be entitled to any remedies at law or in equity to recover losses, liabilities or damages arising from such breach. The Issuer shall promptly notify Subscriber of the termination of the Merger Agreement promptly after the termination of such agreement. Upon the termination of this Subscription Agreement in accordance with this Section 5, any monies paid by Subscriber to the Issuer in connection herewith shall be promptly (and in any event within two (2) Business Days after such termination) returned to Subscriber.

6. Miscellaneous.

- 6.1 <u>Further Assurances</u>. At the Closing, the parties hereto shall execute and deliver such additional documents and take such additional actions as the parties reasonably may deem to be practical and necessary in order to consummate the Subscription as contemplated by this Subscription Agreement.
 - 6.1.1 Subscriber acknowledges that the Issuer, the SPAC and others will rely on the acknowledgments, understandings, agreements, representations and warranties made by Subscriber contained in this Subscription Agreement. Prior to the Closing, Subscriber agrees to promptly notify the Issuer and the SPAC if any of the acknowledgments, understandings, agreements, representations and warranties made by Subscriber set forth herein are no longer accurate in all material respects. The Issuer and the SPAC acknowledge that Subscriber and others will rely on the acknowledgments, understandings, agreements, representations and warranties made by the Issuer and the SPAC contained in this Subscription Agreement.
 - 6.1.2 Each of the Issuer, the SPAC and Subscriber is entitled to rely upon this Subscription Agreement and is irrevocably authorized to produce this Subscription Agreement or a copy hereof to any interested party in any administrative or legal proceeding or official inquiry with respect to the matters covered hereby.

- 6.1.3 The Issuer or the SPAC may request from Subscriber such additional information as the Issuer or the SPAC may deem necessary to evaluate the eligibility of Subscriber to acquire the Subscribed Shares, and Subscriber shall provide such information as may be reasonably requested, to the extent within Subscriber's possession and control or otherwise readily available to Subscriber, provided that the Issuer and the SPAC each agree to keep confidential any such information provided by Subscriber.
- 6.1.4 Each of Subscriber, the Issuer and the SPAC shall pay all of their own respective expenses in connection with this Subscription Agreement and the transactions contemplated herein.
- 6.1.5 Each of Subscriber, the Issuer and the SPAC shall take, or cause to be taken, all actions and do, or cause to be done, all things necessary, proper or advisable to consummate the transactions contemplated by this Subscription Agreement on the terms and conditions described therein no later than immediately prior to the consummation of the Transactions.
- 6.2 <u>Notices</u>. Any notice or communication required or permitted hereunder shall be in writing and either delivered personally, emailed or sent by overnight mail via a reputable overnight carrier, or sent by certified or registered mail, postage prepaid, and shall be deemed to be given and received (i) when so delivered personally, (ii) when sent, with no mail undeliverable or other rejection notice, if sent by email, or (iii) three (3) Business Days after the date of mailing to the address below or to such other address or addresses as such person may hereafter designate by notice given hereunder:
 - (i) if to Subscriber, to such address or addresses set forth on the signature page hereto;
 - (ii) if to the Issuer, to:

c/o Paysafe Group Holdings Limited Floor 27, 25 Canada Square London, England, E14 5LQ

Attention: Elliott Wiseman, Group General Counsel & Chief Compliance Officer

Email:

Email:

with a required copy (which copy shall not constitute notice) to:

Simpson Thacher & Bartlett LLP 425 Lexington Avenue New York, NY 10017

Attention: Elizabeth Cooper

Katherine Krause ecooper@stblaw.com

Katherine.Krause@stblaw.com

(iii) if to the SPAC, to:

Foley Trasimene Acquisition Corp. II 1701 Village Center Circle Las Vegas, NV 89134

Attention: Michael L. Gravelle, General Counsel

Email:

with a required copy (which copy shall not constitute notice) to:

Weil, Gotshal & Manges LLP 767 Fifth Avenue New York, NY 10153 Attention: Michael J. Aiello

ention: Michael J. Alelio

Matthew Gilroy

Email: michael.aiello@weil.com matthew.gilroy@weil.com

- 6.3 Entire Agreement. This Subscription Agreement constitutes the entire agreement, and supersedes all other prior agreements, understandings, representations and warranties, both written and oral, among the parties, with respect to the subject matter hereof, including any commitment letter entered into relating to the subject matter hereof.
- 6.4 <u>Modifications and Amendments</u>. This Subscription Agreement may not be amended, modified, supplemented or waived except by an instrument in writing, signed by the party against whom enforcement of such amendment, modification, supplement or waiver is sought.
- 6.5 <u>Assignment</u>. Neither this Subscription Agreement nor any rights, interests or obligations that may accrue to the parties hereunder (including Subscriber's rights to purchase the Subscribed Shares) may be transferred or assigned without the prior written consent of each of the other parties hereto (other than the Subscribed Shares acquired hereunder, if any, and then only in accordance with this Subscription Agreement); <u>provided</u> that Subscriber's rights and obligations hereunder may be assigned to any fund or account managed by the same investment manager as Subscriber, without the prior consent of the Issuer and the SPAC, provided that such assignee(s) agrees in writing to be bound by the terms hereof, and upon such assignment by a Subscriber, the assignee(s) shall become Subscriber hereunder and have the rights and obligations and be deemed to make the representations and warranties of Subscriber provided for herein to the extent of such assignment; <u>provided further</u> that, no assignment shall relieve the assigning party of any of its obligations hereunder, including any assignment to any fund or account managed by the same investment manager as Subscriber.

6.6 <u>Benefit</u>. Except as otherwise provided herein, this Subscription Agreement shall be binding upon, and inure to the benefit of the parties hereto and their heirs, executors, administrators, successors, legal representatives, and permitted assigns, and the agreements, representations, warranties, covenants and acknowledgments contained herein shall be deemed to be made by, and be binding upon, such heirs, executors, administrators, successors, legal representatives and permitted assigns. This Subscription Agreement shall not confer rights or remedies upon any person other than the parties hereto and their respective successors and assigns, except that the Placement Agents shall be third party beneficiaries to the representations and warranties made by the Issuer and Subscriber in this Subscription Agreement.

6.7 <u>Governing Law.</u> This Subscription Agreement, and any claim or cause of action hereunder based upon, arising out of or related to this Subscription Agreement (whether based on law, in equity, in contract, in tort or any other theory) or the negotiation, execution, performance or enforcement of this Subscription Agreement, shall be governed by and construed in accordance with the Laws of the State of Delaware, without giving effect to the principles of conflicts of law thereof.

6.8 Consent to Jurisdiction; Waiver of Jury Trial. Each of the parties irrevocably consents to the exclusive jurisdiction and venue of the Court of Chancery of the State of Delaware, provided that if subject matter jurisdiction over the matter that is the subject of the legal proceeding is vested exclusively in the U.S. federal courts, such legal proceeding shall be heard in the U.S. District Court for the District of Delaware (together with the Court of Chancery of the State of Delaware, "Chosen Courts"), in connection with any matter based upon or arising out of this Subscription Agreement. Each party hereby waives, and shall not assert as a defense in any legal dispute, that (i) such person is not personally subject to the jurisdiction of the Chosen Courts for any reason, (ii) such legal proceeding may not be brought or is not maintainable in the Chosen Courts, (iii) such person's property is exempt or immune from execution, (iv) such legal proceeding is brought in an inconvenient forum or (v) the venue of such legal proceeding is improper. Each party hereby consents to service of process in any such proceeding in any manner permitted by Delaware law, further consents to service of process by nationally recognized overnight courier service guaranteeing overnight delivery, or by registered or certified mail, return receipt requested, at its address specified pursuant to Section 6.2 and waives and covenants not to assert or plead any objection which they might otherwise have to such manner of service of process. Notwithstanding the foregoing in this Section 6.8, a party may commence any action, claim, cause of action or suit in a court other than the Chosen Courts solely for the purpose of enforcing an order or judgment issued by the Chosen Courts. TO THE EXTENT NOT PROHIBITED BY APPLICABLE LAW WHICH CANNOT BE WAIVED, EACH OF THE PARTIES WAIVES ANY RIGHT TO TRIAL BY JURY ON ANY CLAIMS OR COUNTERCLAIMS ASSERTED IN ANY LEGAL DISPUTE RELATING TO THIS SUBSCRIPTION AGREEMENT WHETHER NOW EXISTING OR HEREAFTER ARISING. IF THE SUBJECT MATTER OF ANY SUCH LEGAL DISPUTE IS ONE IN WHICH THE WAIVER OF JURY TRIAL IS PROHIBITED, NO PARTY SHALL ASSERT IN SUCH LEGAL DISPUTE A NONCOMPULSORY COUNTERCLAIM ARISING OUT OF OR RELATING TO THIS SUBSCRIPTION AGREEMENT. FURTHERMORE, NO PARTY SHALL SEEK TO CONSOLIDATE ANY SUCH LEGAL DISPUTE WITH A SEPARATE ACTION OR OTHER LEGAL PROCEEDING IN WHICH A JURY TRIAL CANNOT BE WAIVED.

6.9 <u>Severability</u>. If any provision of this Subscription Agreement shall be invalid, illegal or unenforceable, the validity, legality or enforceability of the remaining provisions of this Subscription Agreement shall not in any way be affected or impaired thereby and shall continue in full force and effect.

6.10 No Waiver of Rights, Powers and Remedies. No failure or delay by a party hereto in exercising any right, power or remedy under this Subscription Agreement, and no course of dealing between the parties hereto, shall operate as a waiver of any such right, power or remedy of such party. No single or partial exercise of any right, power or remedy under this Subscription Agreement by a party hereto, nor any abandonment or discontinuance of steps to enforce any such right, power or remedy, shall preclude such party from any other or further exercise thereof or the exercise of any other right, power or remedy hereunder. The election of any remedy by a party hereto shall not constitute a waiver of the right of such party to pursue other available remedies. No notice to or demand on a party not expressly required under this Subscription Agreement shall entitle the party receiving such notice or demand to any other or further notice or demand in similar or other circumstances or constitute a waiver of the rights of the party giving such notice or demand to any other or further action in any circumstances without such notice or demand.

6.11 Remedies.

6.11.1 The parties agree that irreparable damage would occur if this Subscription Agreement is not performed or the Closing is not consummated in accordance with its specific terms or is otherwise breached and that money damages or other legal remedies would not be an adequate remedy for any such damage. It is accordingly agreed that the parties hereto shall be entitled to equitable relief, including in the form of an injunction or injunctions, to prevent breaches or threatened breaches of this Subscription Agreement and to enforce specifically the terms and provisions of this Subscription Agreement in an appropriate court of competent jurisdiction as set forth in Section 6.8, this being in addition to any other remedy to which any party is entitled at law or in equity, including money damages. The right to specific enforcement shall include the right of the parties hereto to cause the other parties hereto to cause the transactions contemplated hereby to be consummated on the terms and subject to the conditions and limitations set forth in this Subscription Agreement. The parties hereto further agree (i) to waive any requirement for the security or posting of any bond in connection with any such equitable remedy, (ii) not to assert that a remedy of specific enforcement pursuant to this Section 6.11 is unenforceable, invalid, contrary to applicable law or inequitable for any reason and (iii) to waive any defenses in any action for specific performance, including the defense that a remedy at law would be adequate.

6.11.2 The parties acknowledge and agree that this <u>Section 6.11</u> is an integral part of the transactions contemplated hereby and without that right, the parties hereto would not have entered into this Subscription Agreement.

6.11.3 If the Closing does not occur prior to the consummation of the Transactions due to a breach by Subscriber (as determined by a court of competent jurisdiction) of any of its obligations hereunder, then Paysafe or one or more of the Paysafe Equityholders (as defined below) shall have the right (exercisable by written notice to Subscriber (the "Demand Notice") on or before the date that is thirty (30) days after the consummation of the Transactions and as a nonexclusive remedy for any such breach and in addition to and without in any limiting or amending the provisions of Section 6.11.1) to cause Subscriber to purchase from Paysafe (or its assignee(s) or designee(s), including the Paysafe Equityholders) all or a portion of the number of Subscribed Shares that Subscriber failed to purchase at the Closing (the "Subject Shares") for a purchase price of \$10 per Subject Share. At any time and from time to time as of, from and after the consummation of the Transactions under the Merger Agreement, Paysafe may designate one or more of its direct or indirect equityholders that are transferees of the Common Shares from Paysafe that were held by Paysafe as of the consummation of the Transactions (the "Paysafe Equityholders") as sellers of the Subject Shares on a pro rata basis based upon such equityholders' ownership of Paysafe as of immediately prior to the consummation of the Transactions. The consummation of such purchase and sale shall take place on a date fixed by Paysafe or the Paysafe Equityholders, as applicable, in the Demand Notice, which date shall be not sooner than ten (10) Business Days after the date of Subscriber's receipt of the Demand Notice. Subscriber, Paysafe and each Paysafe Equityholder, as applicable, shall be entitled to receive customary representations and warranties regarding such purchase and sale, and Subscriber agrees to execute and deliver all customary purchase documentation as Paysafe or such Paysafe Equityholder may reasonably request. Each such Subscriber and the Issuer shall (and shall cause their and their respective affiliates' directors, officers, employees, partners, agents and representatives to) promptly (but in no event later than ten (10) Business Days) following Subscriber's receipt of the Demand Notice make all governmental filings and obtain all governmental consents and approvals that are required to be made or obtained prior to the consummation of such purchase and sale, including compliance with the notification and reporting requirements of the HSR Act in respect of such purchase and sale.

6.12 <u>Survival of Representations and Warranties and Covenants</u>. All representations and warranties made by the parties hereto, and all covenants and other agreements of the parties hereto, in this Subscription Agreement shall survive the Closing. For the avoidance of doubt, if for any reason the Closing does not occur prior to the consummation of the Transactions, all representations, warranties, covenants and agreements of the parties hereunder shall survive the consummation of the Transactions and remain in full force and effect.

6.13 No Broker or Finder. Each of the Issuer, the SPAC and Subscriber, severally and each as to itself, agrees to indemnify and hold the other parties hereto harmless from any claim or demand for commission or other compensation by any broker, finder, financial consultant or similar agent claiming to have been employed by or on behalf of such party and to bear the cost of legal expenses incurred in defending against any such claim.

- 6.14 <u>Headings and Captions</u>. The headings and captions of the various subdivisions of this Subscription Agreement are for convenience of reference only and shall in no way modify or affect the meaning or construction of any of the terms or provisions hereof.
- 6.15 <u>Counterparts</u>. This Subscription Agreement may be executed in one or more counterparts, all of which when taken together shall be considered one and the same agreement and shall become effective when counterparts have been signed by each party and delivered to the other parties, it being understood that the parties need not sign the same counterpart. In the event that any signature is delivered by facsimile transmission or any other form of electronic delivery, such signature shall create a valid and binding obligation of the party executing (or on whose behalf such signature is executed) with the same force and effect as if such signature page were an original thereof.
- 6.16 <u>Construction</u>. The words "include," "includes," and "including" will be deemed to be followed by "without limitation." Pronouns in masculine, feminine, and neuter genders will be construed to include any other gender, and words in the singular form will be construed to include the plural and vice versa, unless the context otherwise requires. The words "this Subscription Agreement," "herein," "hereof," "hereby," "hereunder," and words of similar import refer to this Subscription Agreement as a whole and not to any particular subdivision unless expressly so limited. The parties hereto intend that each representation, warranty, and covenant contained herein will have independent significance. If any party hereto has breached any representation, warranty, or covenant contained herein in any respect, the fact that there exists another representation, warranty or covenant relating to the same subject matter (regardless of the relative levels of specificity) which such party hereto has not breached will not detract from or mitigate the fact that such party hereto is in breach of the first representation, warranty, or covenant. All references in this Subscription Agreement to numbers of shares, per share amounts and purchase prices shall be appropriately adjusted to reflect any stock split, stock dividend, stock combination, recapitalization or the like occurring after the date hereof.
- 6.17 <u>Mutual Drafting</u>. This Subscription Agreement is the joint product of the parties hereto and each provision hereof has been subject to the mutual consultation, negotiation and agreement of the parties and shall not be construed for or against any party hereto.

7. Cleansing Statement; Disclosure.

7.1 The SPAC shall, by 9:00 a.m., New York City time, on the first (1st) Business Day immediately following the date of this Subscription Agreement, issue one or more press releases or file with the Commission a Current Report on Form 8-K (collectively, the "**Disclosure Document**") disclosing all material terms of the transactions contemplated hereby and by the Other Subscription Agreements and the Transactions. Upon the issuance of the Disclosure Document, to the actual knowledge of Issuer and SPAC, Subscriber shall not be in possession of any material, non-public information received from Issuer, SPAC or any of their respective officers, directors, or employees or agents, and Subscriber shall no longer be subject to any confidentiality or similar obligations under any current agreement, whether written or oral, with Issuer, SPAC, the Placement Agents or any of their respective affiliates, relating to the transactions contemplated by this Subscription Agreement.

- 7.2 Neither the SPAC nor Issuer shall publicly disclose the name of Subscriber or any affiliate or investment adviser of Subscriber, or include the name of Subscriber or any affiliate or investment adviser of Subscriber in any press release or in any filing with the Commission or any regulatory agency or trading market, without the prior written consent (including by e-mail) of Subscriber, except as required by the federal securities laws, rules or regulations and to the extent such disclosure is required by other laws, rules or regulations, at the request of the staff of the Commission or regulatory agency or under regulations of the NYSE, in which case the Issuer or SPAC, as applicable, shall provide Subscriber with prior written notice (including by e-mail) of such permitted disclosure, and shall reasonably consult with Subscriber regarding such disclosure.
- 8. Trust Account Waiver. In addition to the waiver of the Issuer pursuant to Section 7.03 of the Merger Agreement, and notwithstanding anything to the contrary set forth herein, each of the Issuer and Subscriber acknowledges that the SPAC has established a trust account containing the proceeds of its initial public offering and from certain private placements (collectively, with interest accrued from time to time thereon, the "Trust Account"). Each of the Issuer and Subscriber agrees that (i) it has no right, title, interest or claim of any kind in or to any monies held in the Trust Account, and (ii) it shall have no right of set-off or any right, title, interest or claim of any kind ("Claim") to, or to any monies in, the Trust Account, in each case in connection with this Subscription Agreement, and hereby irrevocably waives any Claim to, or to any monies in, the Trust Account that it may have in connection with this Subscription Agreement; provided, however, that nothing in this Section 8 shall be deemed to limit Subscriber's right, title, interest or claim to the Trust Account by virtue of such Subscriber's record or beneficial ownership of securities of the SPAC, including, but not limited to, any redemption right with respect to any such securities of the SPAC. In the event any of the Issuer and Subscriber has any Claim against the SPAC under this Subscription Agreement, the Issuer and Subscriber shall pursue such Claim solely against the SPAC and its assets outside the Trust Account and not against the property or any monies in the Trust Account. Each of the Issuer and Subscriber agrees and acknowledges that such waiver is material to this Subscription Agreement and has been specifically relied upon by the SPAC to induce the SPAC to enter into this Subscription Agreement and each of the Issuer and Subscriber further intends and understands such waiver to be valid, binding and enforceable under applicable law. In the event the Issuer or Subscriber, in connection with this Subscription Agreement, commences any action or proceeding which seeks, in whole or in part, relief against the funds held in the Trust Account or distributions therefrom or any of the SPAC's stockholders, whether in the form of monetary damages or injunctive relief, Issuer or Subscriber, as applicable, shall be obligated to pay to the SPAC all of its legal fees and costs in connection with any such action in the event that the SPAC prevails in such action or proceeding.
- 9. Non-Reliance. Subscriber acknowledges that it is not relying upon, and has not relied upon, any statement, representation or warranty made by any person, firm or corporation, other than the representations and warranties of the Issuer and the SPAC expressly set forth in this Subscription Agreement, in making its investment or decision to invest in the Issuer. Subscriber agrees that no other Subscriber pursuant to this Subscription Agreement or any other agreement related to the private placement of shares of the Issuer's capital stock (including the controlling persons, officers, directors, partners, agents or employees of any such Subscriber) shall be liable to any other Subscriber pursuant to this Subscription Agreement or any other agreement related to the private placement of shares of the Issuer's capital stock for any action heretofore or hereafter taken or omitted to be taken by any of them in connection with the purchase of the Subscribed Shares hereunder.

- 10. <u>Rule 144</u>. From and after such time as the benefits of Rule 144 promulgated under the Securities Act or any other similar rule or regulation of the Commission that may allow Subscriber to sell securities of the Issuer to the public without registration are available to holders of the Issuer's common stock and until the third anniversary of the Closing Date, the Issuer agrees to:
 - 10.1 make and keep public information available, as those terms are understood and defined in Rule 144; and
- 10.2 file with the Commission in a timely manner all reports and other documents required of the Issuer under the Securities Act and the Exchange Act so long as the Issuer remains subject to such requirements and the filing of such reports and other documents is required for the applicable provisions of Rule 144.

If the Subscribed Shares are eligible to be sold without restriction under, and without the Issuer being in compliance with the current public information requirements of, Rule 144 under the Securities Act, then at Subscriber's request, the Issuer will cause its transfer agent to remove the applicable restrictive legend. In connection therewith, if required by the Issuer's transfer agent, the Issuer will promptly cause an opinion of counsel to be delivered to and maintained with its transfer agent, together with any other authorizations, certificates and directions required by the transfer agent that authorize and direct the transfer agent to issue such Subscribed Shares without any such legend; provided that, notwithstanding the foregoing, Issuer will not be required to deliver any such opinion, authorization, certificate or direction if it reasonably believes that removal of the legend could result in or facilitate transfers of securities in violation of applicable law.

11. <u>Agreements with Respect to Tax Matters</u>. For so long as Subscriber holds Subscribed Shares, (i) the Issuer will, not more frequently than once a year, determine whether it reasonably believes that it is a passive foreign company (a "**PFIC**") as defined in Section 1297 of the Code with respect to the preceding taxable year and, if it determines that it is a PFIC, it will (A) notify Subscriber of its determination within thirty (30) days of such determination and (B) use commercially reasonable efforts to provide Subscriber all information and documents that are reasonably requested and required for Subscriber or any of its underlying holders to make and maintain a qualified electing fund election pursuant to Section 1295 of the Code with respect to the Issuer and any PFIC in which the Issuer holds a direct or indirect controlling interest as soon as reasonably practicable following any request for such information by the Subscriber, and (ii) the Issuer will not make an election to be treated as other than a corporation for U.S. federal income tax purposes.

[Signature Page Follows]

IN WITNESS WHEREOF, each of the Issuer, the SPAC and Subscriber has executed or caused this Subscription Agreement to be executed by its duly authorized representative as of the date set forth below.

PAYSAFE LIMITED

By: /s/ Philip McHugh

Name: Philip McHugh

Title: Chief Executive Officer

FOLEY TRASIMENE ACQUISITION CORP. II

By: /s/ Michael L. Gravelle

Name: Michael L. Gravelle

Title: General Counsel and Corporate Secretary

Accepted and agreed this 7th day of December, 2020.	
SUBSCRIBER:	
Signature of Subscriber: Cannae Holdings, LLC By Cannae Holdings, Inc., its Managing Member	Signature of Joint Subscriber, if applicable:
By: /s/ Michael L. Gravelle	By:
Name: Michael L. Gravelle	Name:
Title: Executive Vice President, General Counsel and Corporate	Title:
Secretary	
Date: December 7, 2020	
Name of Subscriber:	Name of Joint Subscriber, if applicable:
Cannae Holdings, LLC	
(Please print. Please indicate name and	(Please Print. Please indicate name and
capacity of person signing above)	capacity of person signing above)
Name in which securities are to be registered (if different from the name of Subscriber listed directly above): Email Address:	
If there are joint investors, please check one:	
\square Joint Tenants with Rights of Survivorship	
☐ Tenants-in-Common	
☐ Community Property	
Subscriber's EIN:	Joint Subscriber's EIN:
Business Address-Street:	Mailing Address-Street (if different):

City, State, Zip: 601 Riverside Avenue Jacksonville, FL 32204	City, State, Zip:			
Attn:	Attn:			
Telephone No.:	Telephone No.:			
Facsimile No.:	Facsimile No.:			
Aggregate Number of Subscribed Shares subscribed for:				

35,000,000

Aggregate Purchase Price: \$350,000,000.

You must pay the Purchase Price by wire transfer of U.S. dollars in immediately available funds, to be held in escrow until the Closing, to the account specified by the Issuer in the Closing Notice.

SCHEDULE I ELIGIBILITY REPRESENTATIONS OF SUBSCRIBER

X

is not:

A.	A. QUALIFIED INSTITUTIONAL BUYER STATUS (Please check the applicable subparagraphs):							
	1.	\Box We are a "qualified institutional buyer" (as defined in Rule 144A under the Securities Act of 1933, as amended (the "Securities Act") (a "QIB")).						
	2.	\square We are subscribing for the Subscribed Shares as a fiduciary or agent for one or more investor accounts, and each owner of such account is a QIB.						
***	OR ***							
B.	. INSTITUTIONAL ACCREDITED INVESTOR STATUS (Please check the applicable subparagraphs):							
	1.	☑ We are an "accredited investor" (within the meaning of Rule 501(a) under the Securities Act) or an entity in which all of the equity holders are accredited investors within the meaning of Rule 501(a) under the Securities Act, and have marked and initialed the appropriate box on the following page indicating the provision under which we qualify as an "accredited investor."						
	2.	☑ We are not a natural person.						
***	AND **	**						
C. A	FFILIA	TE STATUS						
	(Please check the applicable box) SUBSCRIBER:							
		is:						

This page should be completed by Subscriber and constitutes a part of the Subscription Agreement.

an "affiliate" (as defined in Rule 144 under the Securities Act) of the Issuer or acting on behalf of an affiliate of the Issuer.

-	scriber has indicated, by marking and initialing the appropriate box below, the provision(s) below which apply to Subscriber and under criber accordingly qualifies as an "accredited investor."
	Any bank as defined in section $3(a)(2)$ of the Securities Act, or any savings and loan association or other institution as defined in section $3(a)(5)(A)$ of the Securities Act whether acting in its individual or fiduciary capacity;
	Any broker or dealer registered pursuant to section 15 of the Securities Exchange Act of 1934, as amended;
	Any insurance company as defined in section 2(a)(13) of the Securities Act;
	Any investment company registered under the Investment Company Act of 1940, as amended (the " <u>Investment Company Act</u> ") or a business development company as defined in section 2(a)(48) of the Investment Company Act;
	Any Small Business Investment Company licensed by the U.S. Small Business Administration under section 301(c) or (d) of the Small Business Investment Act of 1958, as amended;
	Any plan established and maintained by a state, its political subdivisions, or any agency or instrumentality of a state or its political subdivisions, for the benefit of its employees, if such plan has total assets in excess of \$5,000,000;
	Any employee benefit plan within the meaning of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), if (i) the investment decision is made by a plan fiduciary, as defined in section 3(21) of ERISA, which is either a bank, a savings and loan association, an insurance company, or a registered investment adviser, (ii) the employee benefit plan has total assets in excess of \$5,000,000 or, (iii) such plan is a self-directed plan, with investment decisions made solely by persons that are "accredited investors";
	Any private business development company as defined in section 202(a)(22) of the Investment Advisers Act of 1940, as amended;
\boxtimes	Any (i) corporation, limited liability company or partnership, (ii) Massachusetts or similar business trust, or (iii) organization described in section 501(c)(3) of the Internal Revenue Code of 1986, as amended, not formed for the specific purpose of acquiring the securities offered, and with total assets in excess of \$5,000,000;
	Any director, executive officer, or general partner of the issuer of the securities being offered or sold, or any director, executive officer, or general partner of a general partner of that issuer;

Rule 501(a) under the Securities Act, in relevant part, states that an "accredited investor" shall mean any person who comes within any of the below listed categories, or who the issuer reasonably believes comes within any of the below listed categories, at the time of the sale of the securities to that

Any natural person whose individual net worth, or joint net worth with that person's spouse, exceeds \$1,000,000. For purposes of calculating a natural person's net worth: (a) the person's primary residence shall not be included as an asset; (b) indebtedness that is secured by the person's primary residence, up to the estimated fair market value of the primary residence at the time of the sale of securities, shall not be included as a liability (except that if the amount of such indebtedness outstanding at the time of sale of securities exceeds the amount outstanding sixty (60) days before such time, other than as a result of the acquisition of the primary residence, the amount of such excess shall be included as a liability); and (c) indebtedness that is secured by the person's primary residence in excess of the estimated fair market value of the primary residence at the time of the sale of securities shall be included as a liability;
Any natural person who had an individual income in excess of \$200,000 in each of the two most recent years or joint income with that person's spouse in excess of \$300,000 in each of those years and has a reasonable expectation of reaching the same income level in the current year;
Any trust, with total assets in excess of $5,000,000$, not formed for the specific purpose of acquiring the securities offered, whose purchase is directed by a sophisticated person as described in Section $230.506(b)(2)(ii)$ of Regulation D; or
Any entity in which all of the equity owners are "accredited investors."

December 7, 2020

Foley Trasimene Acquisition Corp. II 1701 Village Center Circle Las Vegas, NV 89134

Paysafe Limited

Paysafe Bermuda Holding LLC

Re: Sponsor Agreement

Ladies and Gentlemen:

This letter (this "Sponsor Agreement") is being delivered to you in accordance with that certain Agreement and Plan of Merger ("Merger Agreement"), dated as of the date hereof, by and among Foley Trasimene Acquisition Corp. II, a Delaware corporation ("FTAC"), Paysafe Limited, an exempted limited company incorporated under the laws of Bermuda (the "Company"), Paysafe Merger Sub Inc., a Delaware corporation and direct, wholly owned subsidiary of the Company ("Merger Sub"), Paysafe Bermuda Holding LLC, a Bermuda limited liability company (the "LLC"), and Paysafe Group Holdings Limited, a private limited company incorporated under the laws of England and Wales ("PGHL"), and hereby amends and restates in its entirety (a) that certain letter, dated August 21, 2020, from Trasimene Capital FT, LP II, a Delaware limited partnership (the "Sponsor") to FTAC (the "Prior Sponsor Letter Agreement") and (b) each certain letter, each dated August 21, 2020 from each of the other persons undersigned thereto (each, an "Insider") to FTAC (each, a "Prior Insider Letter Agreement" and, together with the Prior Sponsor Letter Agreement, the "Prior Letter Agreements"). Certain capitalized terms used herein are defined in Paragraph 11. Capitalized terms used but not defined herein shall have the respective meanings given to them in the Merger Agreement.

FP Investor (together with Cannae Holdings, LLC ("Cannae")), the Insiders and the Sponsor, the "Sponsor Persons") pursuant to that certain Forward Purchase Agreement, entered into as of July 31, 2020, by and between FP Investor and FTAC, will acquire, or cause Cannae to acquire, and be the record owner, as of the Closing, of that number of shares of FTAC Class A Common Stock and FP Investor FTAC Warrants detailed on Schedule A hereto.

The Sponsor and certain of the Insiders are currently, and as of the Closing will be, the record owners of all of the outstanding Founder Shares and outstanding Founder FTAC Warrants, with each such Person's ownership (and anticipated changes in ownership as a result of the Warrant Contribution) detailed on Schedule A hereto.

As described further in Paragraph 25, <u>Schedule A</u> will be updated from time to time to reflect any Sponsor Person ownership changes following the date hereof.

In order to induce FTAC and the PGHL Parties to enter into the Merger Agreement and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, each Sponsor Person hereby agrees, severally and not jointly, with FTAC and the PGHL Parties as follows:

- 1. <u>Voting Obligations</u>. During the Interim Period, each Sponsor Person, in its capacity as a holder of Covered Shares, agrees irrevocably and unconditionally that, at the Special Meeting, at any other meeting of the shareholders of FTAC or, following the Transactions, the Company (whether annual or special and whether or not an adjourned or postponed meeting, however called and including any adjournment or postponement thereof), in connection with any written consent of shareholders of FTAC or, following the Transactions, the Company and in connection with any similar vote or consent of the holders of Founder FTAC Warrants, in their capacities as such, such Sponsor Person shall, and shall cause any other holder of record of any of such Sponsor Person's Covered Shares to:
- (a) when such meeting is held, appear at such meeting or otherwise cause the Sponsor Person's Covered Shares to be counted as present thereat for the purpose of establishing a quorum;
- (b) vote (or duly and promptly execute and deliver an action by written consent), or cause to be voted at such meeting (or cause such consent to be duly and promptly executed and delivered with respect to), all of such Sponsor Person's Covered Shares owned as of the record date for determining holders entitled to vote at such meeting (or the record date for determining holders entitled to provide consent) in favor of the FTAC Stockholder Matters and any other matters necessary or advisable for consummation of the Transactions (including the Warrant Contribution, to the extent it is put to a vote or a request for written consent); and
- (c) vote (or duly and promptly execute and deliver an action by written consent), or cause to be voted at such meeting (or cause such consent to be duly and promptly executed and delivered with respect to), all of such Sponsor Person's Covered Shares against any Business Combination Proposal (as defined below) and any other action that is intended, or would reasonably be expected, to impede, interfere with or delay or postpone the consummation of, or otherwise adversely affect, any of the Transactions or result in a breach of any representation, warranty, covenant or other obligation or agreement of FTAC under the Merger Agreement or result in a breach of any representation, warranty, covenant or other obligation or agreement of such Sponsor Person under this Sponsor Agreement.

The obligations of the Sponsor Persons in this Paragraph 1 shall apply whether or not the board of directors of FTAC (or, following the Transactions, the Company) or other governing body or any committee, subcommittee or subgroup thereof recommends the FTAC Stockholder Matters or any other matters necessary or advisable for consummation of the Transactions and whether or not such board or other governing body, committee, subcommittee or subgroup thereof changes, withdraws, withholds, qualifies or modifies, or publicly proposes to change, withdraw, withhold, qualify or modify, the FTAC Board Recommendation.

2. Exclusivity. During the Interim Period, each Sponsor Person shall not take, nor shall it permit any of its Affiliates or any of its or their respective Representatives to take, whether directly or indirectly, any action to (i) solicit, initiate, continue or engage in discussions or negotiations with, or enter into any agreement with, or encourage, respond, provide information to or commence due diligence with respect to, any Person (other than PGHL, its shareholders and/or any of their Affiliates or Representatives), concerning, relating to or which is intended or is reasonably likely to give rise to or result in, any offer, inquiry, proposal or indication of interest, written or oral relating to any Business Combination (a "Business Combination Proposal") or (ii) approve, endorse or recommend, or make any public statement approving, endorsing or recommending, any Business Combination Proposal, in the case of each of clauses (i) and (ii), other than a Business Combination Proposal with the Company, its equityholders and their respective Affiliates and Representatives. Each Sponsor Person shall, and shall cause its Affiliates and Representatives to, immediately cease any and all existing discussions or negotiations with any Person conducted prior to the date hereof with respect to, or which is reasonably likely to give rise to or result in, a Business Combination Proposal, other than with the Company, its equityholders or their respective controlled Affiliates.

3. Waiver of Certain Rights. Each Sponsor Person hereby irrevocably and unconditionally agrees:

- (a) not to (i) demand that FTAC redeem its or their Covered Shares in connection with the Transactions or (ii) otherwise participate in any such redemption by tendering or submitting any of its Covered Shares for redemption; and
- (b) not to commence or participate in, and to take all actions necessary to opt out of any class in any class action with respect to, any claim, derivative or otherwise, against FTAC, the PGHL Parties, any Affiliate of FTAC or the PGHL Parties or any designee of a Sponsor Person or PGHL Party acting in its capacity as director, officer or manager or in any similar capacity or any of their respective successors and assigns relating to the negotiation, execution or delivery of this Sponsor Agreement, the Merger Agreement or the consummation of the Transactions.

4. Reasonable Best Efforts; Regulatory Undertakings.

- (a) During the Interim Period, each Sponsor Person and PGHL shall, and PGHL shall cause each PGHL Person to, (i) use reasonable best efforts to take, or cause to be taken, all actions to do, or cause to be done, all things reasonably necessary, proper or advisable to consummate the Transactions on the terms and subject to the conditions set forth in the Merger Agreement and (ii) not take any action that would reasonably be expected to prevent or delay the satisfaction of any of the conditions to the Transactions set forth in Article X of the Merger Agreement.
- (b) Without limiting the generality of subsection (a) above, each Sponsor Person and PGHL shall, and PGHL shall cause each PGHL Person to, use reasonable best efforts to take, or cause to be taken, all action, and to do, or cause to be done as promptly as practicable, all things necessary, proper and advisable to obtain any required consent, waiver, approval, authorization, qualification from the FCA and the CBI in connection with the Transactions. Furthermore, each Sponsor Person agrees that, if such Sponsor Person is required to obtain FCA Approval and / or CBI Approval, it shall make any such notification (the "Change of Control Notification") to the FCA and / or CBI as promptly as practicably (and in any event within five (5) Business Days of the date of this Support Agreement). If at any time any Sponsor Person becomes aware of any event, circumstance or condition that would be reasonably likely to prevent FCA Approval or CBI Approval being satisfied, such Sponsor Person shall immediately inform the other parties, and reasonably cooperate with them to address any such event, circumstance or condition.
- (c) Without limiting the generality of the foregoing, each Sponsor Person and PGHL shall, and PGHL shall cause each PGHL Person to, use reasonable best efforts to provide or cause to be provided (including, with respect to filings pursuant to the HSR Act, by its "Ultimate Parent Entities", as that term is defined in the HSR Act, and with respect to the FCA Approval and/or the CBI Approval, by any applicable Sponsor Person to the extent required for such approval) as promptly as reasonably practicable and advisable to any Governmental Authority information and documents relating to such party as requested by such Governmental Authority or necessary, proper or advisable to permit consummation of the Transactions, including filing any notification and report form and related material required under the HSR Act and any other filing or notice that may be required with any other Governmental Authority as promptly as reasonably practicable and advisable after the date hereof and thereafter to respond as promptly as reasonably practicable and advisable to any request for additional information or documentary material relating to such party that may be made. Each Sponsor Person shall supply as promptly as practicable (and shall respond no later than five (5) days following any request) any additional information and documentary material relating to such Sponsor Person that may be requested by any Governmental Authority, and each Sponsor Person and PGHL shall, and PGHL shall cause each PGHL Person to, (A) provide, or cause to be provided, as promptly as practicable, all agreements,

documents, instruments, affidavits, statements or information (including, for the avoidance of doubt, nonpublic or other confidential financial or sensitive personally identifiable information as well as personal information of senior management, directors or control persons) that may be required or requested by any Governmental Authority relating to (i) such Sponsor Person or PGHL Person, as the case may be (including any of its respective directors, officers, employees, partners, members, shareholders or control persons) and (ii) such Sponsor Person's or PGHL Person's structure, ownership, businesses, operations, regulatory and legal compliance, assets, liabilities, financing, financial condition or results of operations, or any of its or their directors, officers, employees, partners, members, shareholders or Affiliates, (B) make individuals with appropriate seniority and expertise available to participate in any discussions or hearings and (C) prepare and file any applications, notices, registrations and requests as may be required or advisable to be filed with any Governmental Authority.

(d) In addition, each Sponsor Person and each PGHL Person appointed or designated, or proposed to be appointed or designated, to the board of directors of the Company or any Subsidiary, including pursuant to the terms of Section 9.10 of the Merger Agreement, shall use reasonable best efforts to comply and cooperate with and satisfy all requests and requirements made by any Governmental Authority in connection with any such appointment, designation or proposed service, including by furnishing all requested information, providing reasonable assistance in connection with the preparation of any required applications, notices and registrations and requests and otherwise facilitating access to and being available with respect to any discussions or hearings. Each Sponsor Person and each PGHL Person further acknowledges and agrees that, in the event an individual to be designated by a Shareholder Designator (as defined in the Shareholders Agreement) in accordance with Section 2.2 of the Shareholders Agreement does not satisfy any requirement of a Governmental Authority to serve as a director, each of the Company, PGHL and FTAC shall use their respective reasonable best efforts to comply with such requirements so as to remove any such restriction; provided, that (x) there shall be no obligation to appoint such individual pursuant thereto and (y) such Shareholder Designator shall be entitled to designate a replacement director in lieu of such person.

5. FTAC LLC Contribution.

- (a) In connection with the Closing and immediately prior to the FTAC LLC Contribution (as defined below), each Sponsor Person that holds one or more FTAC Founder Warrants will transfer such Sponsor Person's FTAC Founder Warrants to FTAC in exchange for a number of shares of FTAC Class C Common Stock equal to the number of shares of FTAC Class A Common Stock underlying such FTAC Founder Warrant(s) (the "FTAC Founder Warrant Transfer").
- (b) Immediately prior to the Effective Time, each Sponsor Person that receives FTAC Class C Common Stock pursuant to Section 5(a) shall contribute such holder's shares of FTAC Class C Common Stock to the LLC in exchange for (i) an equivalent number of equity interests in the LLC ("LLC Interest") equal in value to the FTAC Founder Warrant exchanged for such share of FTAC Class C Common Stock in the FTAC Founder Warrant Transfer and (ii) the right of such Sponsor Person to cause the LLC to acquire such LLC Interest(s) in exchange for, at the option of the LLC, the Cash Amount or the Exchanged Shares (such right, the "Exchange Right", and each such LLC Interest and Exchange Right, together, an "Exchangeable Unit") in the manner described in this Paragraph 5 (the "FTAC LLC Contribution"). Following the FTAC LLC Contribution, all of the FTAC Class C Common Stock will be owned by the LLC and shall remain outstanding.

- (c) FTAC hereby represents and warrants that all of the FTAC Founder Warrants are held in book-entry form and that the transfer books for the FTAC Founder Warrants are maintained by Continental Stock Transfer & Trust Company (the "Warrant Agent"). During the Interim Period, FTAC shall not, and shall cause the Warrant Agent not to, allow the Transfer of any FTAC Founder Warrants or allow any of the FTAC Founder Warrants to be represented by a certificate or other instrument. Further, no Sponsor Person shall request the Transfer of any FTAC Founder Warrants or request for any of the FTAC Founder Warrants to be represented by a certificate or other instrument.
- (d) FTAC, the LLC and each applicable Sponsor Person agree to cooperate with each other and their respective officers, employees, attorneys, accountants and other agents, and, generally, do such other reasonable acts and things in good faith as may be necessary to effectuate the intents and purposes of the FTAC Founder Warrant Transfer and the FTAC LLC Contribution, subject to the terms and conditions hereof and in compliance with applicable Law, including taking reasonable action to facilitate the filing of any document or the taking of reasonable action to assist the other parties hereto in complying with the terms hereof.
- (e) The Exchange Right shall not be exercisable by any Sponsor Person until a date that is after the one year anniversary of the date the FTAC LLC Contribution is completed.

6. Transfer Restrictions.

- (a) <u>Interim Period</u>. During the Interim Period, except as expressly contemplated by the FTAC LLC Contribution, each Sponsor Person shall not, and shall cause any other holder of record of any of such Sponsor Person's Covered Shares not to, Transfer any such Sponsor Person's Covered Shares.
- (b) Post-Closing: Covered Shares. For the period beginning on the Closing until the earlier of (i) 270 days thereafter, or (ii) if the VWAP of the Company Common Shares equals or exceeds \$12.00 per share (as adjusted for share splits, share capitalizations, reorganizations, recapitalizations and the like) for any twenty (20) Trading Days within a period of thirty (30) consecutive Trading Days, 150 days thereafter (such applicable period, the "Lock-Up Period"), each Sponsor Person shall not, and shall cause any other holder of record of any of such Sponsor Person's Covered Shares not to, Transfer any of such Sponsor Person's Covered Shares. Notwithstanding the immediately preceding sentence, post-Closing Transfers of Covered Shares that are held by any Sponsor Person or any of its Permitted Transferees (as defined below) that have entered into a written agreement contemplated by the proviso in this subsection (b) are permitted (i) to the Company's officers or directors, any Affiliates or family members of any of the Company's officers or directors, to Sponsor, any members or partners of the Sponsor or their Affiliates, any Affiliates of the Sponsor, or any employees of such Affiliates; (ii) in the case of an individual, by gift to a member of the individual's immediate family or to a trust, the beneficiary of which is a member of the individual's immediate family, an Affiliate of such Person or to a charitable organization; (iii) in the case of an individual, by virtue of laws of descent and distribution upon death of the individual; (iv) in the case of an individual, pursuant to a qualified domestic relations order; (v) by virtue of the Laws of the State of Delaware or the Sponsor's limited partnership agreement, as amended from time to time, upon dissolution of the Sponsor; or (vi) in the event of the Company's completion of a liquidation, merger, amalgamation, share exchange, reorganization or other similar transaction which results in the holders of all of the Company Common Shares having the right to exchange their shares for cash, securities or other property subsequent to the completion FTAC's initial Business Combination (including the entry into an agreement in connection with such liquidation, merger, amalgamation, share exchange, reorganization or other similar transaction); provided, however, that each transferee contemplated by clauses (i) through (v) (each, a "Permitted Transferee") must enter into a written agreement with the Company agreeing to be bound by the restrictions in this Sponsor Agreement.
 - (c) Any Transfer in violation of the provisions of this Paragraph 6 shall be null and void ab initio and of no force or effect.

(d) The parties hereto agree that, if the lock-up provisions in Section 4.3 of the Shareholders Agreement are modified in a manner that is favorable to the CVC Investors and/or Blackstone Investors (as each such term is defined therein), the corresponding modifications shall automatically apply to this Paragraph 6; provided, that, for purposes of the Lock-up Period, such corresponding modifications shall mean the amount of time by which such period is shortened for the CVC Investors and/or the Blackstone Investors.

7. Exchange of Securities; Forfeited Founder Shares.

- (a) <u>General</u>. Each Sponsor Person that holds Founder Shares agrees that, notwithstanding the provisions of Article Eighth of the FTAC Charter, each Founder Share will be exchanged on a one-for-one basis (and on the same basis as each share of FTAC Class A Common Stock) into one Company Common Share pursuant to and in accordance with the Merger Agreement. Subject to Paragraph 7(b), the Founder Shares and Exchangeable Units shall be 100% vested from and after the Closing and will not be subject to any vesting restrictions.
- (b) <u>Forfeited Founder Shares</u>. Each Sponsor Person that holds Founder Shares acknowledges and agrees that, subject to the satisfaction or waiver of each of the conditions to Closing set forth in Sections 10.01 and 10.02 of the Merger Agreement, immediately prior to the Closing, they shall, in the aggregate and based on the percentages set forth on <u>Schedule A</u> hereto, forfeit and surrender to FTAC a number of shares of FTAC Class B Common Stock equal to 7,987,877 shares of FTAC Common Stock (the "*Forfeited Founder Shares*"). Following such forfeiture, the Forfeited Founder Shares shall be cancelled, no longer outstanding and become void and of no further effect.

8. Certain Securities Law Representations and Warranties. Each Sponsor Person hereby represents and warrants as follows:

- (a) it has never been suspended or expelled from membership in any securities or commodities exchange or association or had a securities or commodities license or registration denied, suspended or revoked;
- (b) in the case of Insiders only, its biographical information furnished to FTAC, if any (including any such information included in the Registration Statement), is true and accurate in all material respects and does not omit any material information with respect to such Insider's background;
 - (c) its questionnaire furnished to FTAC, if any, is true and accurate in all material respects;
- (d) it is not subject to or a respondent in any legal action for any injunction, cease-and-desist order or order or stipulation to desist or refrain from any act or practice relating to the offering of securities in any jurisdiction;
- (e) it has never been convicted of, or pleaded guilty to, any crime (i) involving any fraud, (ii) relating to any financial transaction or handling of funds of another Person or (iii) pertaining to any dealings in any securities, and it is not currently a defendant in any such criminal proceeding; and
- (f) no Person (other than the Sponsor Persons), directly or indirectly, (i) controls FTAC, through one or more intermediaries or otherwise, (ii) Beneficially Owns any Covered Shares or (iii) has entered into an express or implied agreement with any Sponsor Person with respect to the voting of the Covered Shares for purposes of Section 422 of FISMA and Regulation 15 of the Transparency (Directive 2004/109/EC) Regulations 2007 (other than as provided herein and in the Shareholders Agreement).

To the extent requested or required in order to confirm or substantiate, or otherwise in connection with, the representation set forth in Paragraph 8(f) above, including in connection with a request from a Governmental Authority related thereto, the Sponsor Persons shall cause their respective Affiliates to comply with the terms of Paragraph 4(b).

- 9. Certain Payments. Except as set forth on Schedule B hereto, no Sponsor Person, nor any Affiliate thereof, nor any director, officer or manager of (or person acting in a similar capacity with respect to) FTAC, shall receive from FTAC, any finder's fee, reimbursement, consulting fee, monies in respect of any repayment of a loan or other compensation prior to, or in connection with any services rendered in order to effectuate the consummation of, FTAC's initial Business Combination (regardless of the type of transaction that it is), other than the following, none of which will be made from the proceeds held in the Trust Account prior to the completion of the initial Business Combination and, subject to the terms of the Merger Agreement and as disclosed in connection therewith, each of which shall, as of and in connection with the Closing, be paid off in full and no further liabilities or obligations with respect thereof shall be due and owing by FTAC or the Company or any of its Subsidiaries from and after the Closing:

 (i) reimbursement of funds advanced to FTAC by the Sponsor to cover offering-related and organizational expenses; (ii) reimbursement for office space and administrative support services provided to FTAC by FP Investor, in the amount of \$5,000 per month; and (iii) reimbursement of legal fees and expenses incurred by the Sponsor, officers or directors in connection with FTAC's formation and their services to FTAC (which, for the avoidance of doubt, shall not include legal fees and expenses incurred in connection with the initial Business Combination, which shall be FTAC Transaction Expenses). During the Interim Period, each Sponsor Person agrees not to enter into, modify or amend any Contract between or among any Sponsor Person or any Affiliate thereof, on the one hand, and the Company or any of its Subsidiaries, on the other hand, that would contradict, limit, restrict or impair any Person's ability to perform or satisfy any obligation under this Sponsor Agreement or the Merger Agreement.
- 10. <u>Service as Officer or Director</u>. Each Sponsor Person has full right and power, without violating any agreement to which it is bound (including, without limitation, any non-competition or non-solicitation agreement with any employer or former employer), to enter into this Sponsor Agreement and, as applicable, to serve as an officer, director or manager of (or in a similar capacity with respect to) FTAC.
- 11. <u>Definitions</u>. As used herein, the following terms shall have the respective meanings set forth below:
- (a) "*Beneficially Own*" means (i) to exercise voting or dispositive authority over a relevant security, as determined under Rule 13d-3 of the Exchange Act and/or (ii) to hold such security or voting power relating to such security for the purposes of section 422 of the Financial Services and Markets Act 2000, and/or (iii) to hold such security or voting power relating to such security for the purposes of Regulation 15 of the Transparency (Directive 2004/109/EC) Regulations 2007.
 - (b) "Business Combination" has the meaning given to it in the Prior Letter Agreements.
- (c) "Cash Amount" in respect of an Exchangeable Unit, means a cash amount equal to (i) the average trading price of a Company Common Share on the last Trading Day prior to the Exchange Date less (ii) the strike price of an FTAC Founder Warrant exchanged for such share of FTAC Class C Common Stock.
- (d) "Covered Shares" means all Founder Shares, all FTAC Founder Warrants, all FP Investor FTAC Warrants, all shares of FTAC Common Stock and all other Company Common Shares and Exchangeable Units (including Exchangeable Units received in the FTAC LLC Contribution and shares issuable in connection with the Forward Purchase Agreement) and other shares of capital stock or

equity securities of FTAC (prior to the Merger) or the PGHL Parties (following the Merger), or securities convertible into, exercisable or exchangeable for the same, of which any Sponsor Person owns as of the date hereof or acquires record or beneficial ownership after the date hereof, including by purchase, as a result of a stock dividend, stock split, recapitalization, combination, reclassification, exchange or change of such shares, or upon exercise or conversion of any securities.

- (e) "Exchange Date" means the date when a Sponsor Person desires the LLC to exchange the Exchangeable Units for the Cash Amount or the Exchanged Shares pursuant to the exercise of the Exchange Right.
- (f) "Exchanged Shares" in respect of an Exchangeable Unit, means a number of Company Common Shares, valued at the average trading price of a Company Common Share on the last Trading Day prior to the Exchange Date, equal to the Cash Amount.
- (g) "Founder Shares" means: (i) as of the date hereof, the 36,375,836 shares of FTAC Common Stock that were purchased in a private placement prior to the IPO; (ii) after the forfeiture of the Forfeited Founder Shares pursuant to Paragraph 7(b), 28,687,959 shares of FTAC Common Stock; and (iii) following the Merger, the 28,687,959 Company Common Shares into which the aggregate amount of shares of FTAC Common Stock referred to in clause (ii) are converted pursuant to the Merger and, each such share, a "Founder Share".
 - (h) "IPO" has the meaning given to it in the Prior Letter Agreements.
 - (i) "PGHL Person" means PGHL, the Company, the LLC, and each of the directors and officers of PGHL.
 - (j) "Registration Statement" has the meaning given to it in the Prior Letter Agreements.
- (k) "Shareholders Agreement" means that certain Shareholders Agreement to be entered into in connection with the Closing, an agreed form of which is attached as Exhibit D to the Merger Agreement.
- (l) "*Transfer*" means the (i) sale of, offer to sell, contract or agreement to sell, hypothecate, pledge, hedge, grant of any option to purchase or otherwise dispose of in any manner (including by merger, consolidation, division, operation of law or otherwise) or agreement to dispose of, directly or indirectly, or establishment or increase of a put equivalent position or liquidation with respect to or decrease of a call equivalent position within the meaning of Section 16 of the Exchange Act and the rules and regulations of the SEC promulgated thereunder with respect to, any security (including, for the avoidance of doubt, through a Transfer of equity securities in a Person who owns such security), (ii) entry into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any security, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (iii) public announcement of any intention to effect any transaction specified in clause (i) or (ii).
- (m) "*VWAP*" means, for any security as of any date(s), the dollar volume-weighted average price for such security on the principal securities exchange or securities market on which such security is then traded during the period beginning at 9:30:01 a.m., New York time, and ending at 4:00:00 p.m., New York time, as reported by Refinitiv Workspace or, if the foregoing does not apply, the dollar volume-weighted average price of such security in the over-the-counter market on the electronic bulletin board for such security during the period beginning at 9:30:01 a.m., New York time, and ending at 4:00:00 p.m., New York time, as reported by Refinitiv Workspace, or, if no dollar volume-weighted average price is

reported for such security by Refinitiv Workspace for such hours, the average of the highest closing bid price and the lowest closing ask price of any of the market makers for such security as reported by OTC Markets Group Inc. If the VWAP cannot be calculated for such security on such date(s) on any of the foregoing bases, the VWAP of such security on such date(s) shall be the fair market value per share on such date(s) as reasonably determined by the Company's board of directors.

- 12. Entire Agreement; Amendment; No Reliance. This Sponsor Agreement and the other agreements referenced herein constitute the entire agreement and understanding of the parties hereto in respect of the subject matter hereof and supersede all prior understandings, agreements, or representations by or among the parties hereto, written or oral, to the extent they relate in any way to the subject matter hereof or the transactions contemplated hereby, including, without limitation, with respect to the Sponsor Persons, the Prior Sponsor Letter Agreement. This Sponsor Agreement may not be changed, amended, modified or waived (other than to correct a typographical error) as to any particular provision, except by a written instrument executed by the parties hereto. Each of FTAC and the Sponsor Persons hereby acknowledges and agrees, on behalf of itself, its Affiliates and its Representatives, that, in connection with its entry into this Sponsor Agreement and (if applicable) the Merger Agreement and agreement to consummate the Transactions, none of the foregoing has relied on any representations or warranties of any PGHL Party or otherwise except for those expressly set forth in the Merger Agreement.
- 13. <u>Assignment</u>. No party hereto may, except as set forth herein, assign either this Sponsor Agreement or any of its rights, interests, or obligations hereunder without the prior written consent of the other parties. Any purported assignment in violation of this Paragraph 13 shall be void and ineffectual and shall not operate to transfer or assign any interest or title to the purported assignee. This Sponsor Agreement shall be binding on the parties hereto and their respective successors, heirs, personal representatives, assigns and (in the case of the Sponsor Persons) Permitted Transferees.
- 14. No Third-Party Beneficiaries. Nothing in this Sponsor Agreement shall be construed to confer upon, or give to, any Person other than the parties hereto any right, remedy or claim under or by reason of this Sponsor Agreement or of any covenant, condition, stipulation, promise or agreement hereof. All covenants, conditions, stipulations, promises and agreements contained in this Sponsor Agreement shall be for the sole and exclusive benefit of the parties hereto, and their respective successors, heirs, personal representatives and assigns and (in the case of the Sponsor Persons) Permitted Transferees.
- 15. <u>Captions; Counterparts</u>. The captions in this Sponsor Agreement are for convenience only and shall not be considered a part of or affect the construction or interpretation of any provision of this Sponsor Agreement. This Sponsor Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 16. <u>Severability</u>. This Sponsor Agreement shall be deemed severable, and the invalidity or unenforceability of any term or provision hereof shall not affect the validity or enforceability of this Sponsor Agreement or of any other term or provision hereof. Furthermore, in lieu of any such invalid or unenforceable term or provision, the parties hereto intend that there shall be added as a part of this Sponsor Agreement a provision as similar in terms to such invalid or unenforceable provision as may be possible and be valid and enforceable.
- 17. <u>Governing Law; Jurisdiction; Waiver of Jury Trial</u>. Sections 12.06 and 12.12 of the Merger Agreement are incorporated herein by reference, *mutatis mutandis*.
- 18. <u>Notices</u>. Any notice, consent or request to be given in connection with any of the terms or provisions of this Sponsor Agreement shall be in writing and shall be sent or given in accordance with the terms of Section 12.02 of the Merger Agreement to the applicable party as its principal place of business.

19. <u>Termination</u>. This Sponsor Agreement shall terminate on the earlier of (i) the valid termination of the Merger Agreement (in which case this Sponsor Agreement shall be of no force or effect and shall revert to the Prior Sponsor Letter Agreement or Prior Insider Letter Agreement, as the case may be) and (ii) the expiration of the Lock-Up Period; <u>provided</u>, that no such termination (including one that results in a reversion to the Prior Sponsor Letter Agreement or Prior Insider Letter Agreement, in each case under clause (i)) shall relieve any party hereto from any liability resulting from its pre-termination breach of this Sponsor Agreement.

20. Other Representations and Warranties. Each Sponsor Person hereby represents and warrants (severally and not jointly as to itself only) to FTAC and the PGHL Parties as follows: (i) if such Person is not an individual, it is duly organized, validly existing and in good standing under the laws of the jurisdiction in which it is incorporated, formed, organized or constituted, and the execution, delivery and performance of this Sponsor Agreement and the consummation of the transactions contemplated hereby are within such Person's corporate, limited liability company or other organizational powers and have been duly authorized by all necessary corporate, limited liability company or other organizational actions on the part of such Person; (ii) if such Person is an individual, such Person has full legal capacity, right and authority to execute and deliver this Sponsor Agreement and to perform its obligations hereunder; (iii) this Sponsor Agreement has been duly executed and delivered by such Person and, assuming due authorization, execution and delivery by the other parties to this Sponsor Agreement, this Sponsor Agreement constitutes a legally valid and binding obligation of such Person, enforceable against such Person in accordance with the terms hereof (except as enforceability may be limited by the Enforceability Exceptions); (iv) the execution and delivery of this Sponsor Agreement by such Person do not, and the performance by such Person of its obligations hereunder will not, (A) if such Person is not an individual, conflict with or result in a violation of the organizational documents of such Person, or (B) require any consent or approval that has not been given or other action that has not been taken by any third party (including under any Contract binding upon such Person or such Person's Covered Shares), in each case, to the extent such consent, approval or other action would prevent, enjoin or delay the performance by such Person of its obligations under this Sponsor Agreement; (v) there is no Action pending or, to the knowledge of such Person, threatened against such Person before (or, in the case of a threatened Action, that would be before) any arbitrator or any Governmental Authority, which in any manner challenges or seeks to prevent, enjoin or delay the performance by such Person of its obligations under this Sponsor Agreement; (vi) except as disclosed pursuant to Section 6.08 of the Merger Agreement, no financial advisor, investment banker, broker, finder or other similar intermediary is entitled to any fee or commission from such Person, FTAC or the Company (or any of their respective Subsidiaries), or any Affiliates of any of the foregoing Persons in connection with the Merger Agreement or this Sponsor Agreement or any of the respective transactions contemplated thereby and hereby, in each case, based upon any arrangement or agreement made by or, to the knowledge of such Person, on behalf of such Person, for which FTAC, the Company and PGHL or any of their respective Affiliates would have any obligations or liabilities of any kind or nature; (vii) such Person has had the opportunity to read the Merger Agreement and this Sponsor Agreement and has had the opportunity to consult with its tax and legal advisors; (viii) such Person has not entered into, and will not enter into, any agreement that would restrict, limit or interfere with the performance of such Person's obligations hereunder; (ix) such Person has good and valid title to all Covered Shares held by it, and there exist no Liens or any other limitation or restriction (including, without limitation, any restriction on the right to vote, sell or otherwise dispose of such securities (other than transfer restrictions under the Securities Act) affecting any such securities, other than pursuant to (A) this Sponsor Agreement, (B) the FTAC Bylaws, (C) the FTAC Charter, (D) the Merger Agreement or (E) any applicable securities Laws; and (x) the Founder Shares, FTAC Founder Warrants, FTAC Class A Common Stock and FP Investor FTAC Warrants listed on Schedule A are the only equity securities in FTAC or any of its Subsidiaries (including, without limitation, any equity securities convertible into, or which can be exercised or exchanged for, equity securities of FTAC or any of its Subsidiaries) owned of record or Beneficially Owned by such Person as of the date hereof and as of

the Closing Date and such Person has the sole power to dispose of (or sole power to cause the disposition of) and the sole power to vote (or sole power to direct the voting of) such Founder Shares, FTAC Founder Warrants, FTAC Class A Common Stock and FP Investor FTAC Warrants and none of such Founder Shares, FTAC Founder Warrants, FTAC Class A Common Stock or FP Investor FTAC Warrants is subject to any proxy, voting trust or other agreement or arrangement with respect to the voting of such Founder Shares, FTAC Founder Warrants, FTAC Class A Common Stock or FP Investor FTAC Warrants, except as provided in this Sponsor Agreement.

- 21. <u>Equitable Adjustments</u>. If, and as often as, there are any changes in FTAC, the Company Common Shares, the Exchangeable Units, the Founder Shares, the FTAC Founder Warrants, FTAC Class A Common Stock, FP Investor FTAC Warrants or the FTAC Class C Common Stock by way of stock split, stock dividend, combination or reclassification, or through merger, consolidation, reorganization, recapitalization or business combination, or by any other means, equitable adjustment shall be made to the provisions of this Sponsor Agreement as may be required so that the rights, privileges, duties and obligations hereunder shall continue with respect to FTAC, the Company Common Shares, the Exchangeable Units, the Founder Shares, the FTAC Founder Warrants, the FTAC Class A Common Stock, the FP Investor FTAC Warrants or the FTAC Class C Common Stock, each as so changed.
- 22. Stop Transfer Order; Legend. Each Sponsor Person hereby authorizes FTAC and the Company to maintain a copy of this Sponsor Agreement at either the executive office or the registered office of the Company. In furtherance of this Sponsor Agreement, each Sponsor Person hereby authorizes FTAC and the Company, promptly after the date hereof, to enter, or cause its transfer agent to enter, a stop transfer order with respect to all of such Sponsor Person's Covered Shares with respect to any Transfer not permitted hereunder and to include the following legend on any certificates or other instruments representing such Sponsor Person's Covered Shares: "THE SHARES OF STOCK OR OTHER SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO CERTAIN VOTING AND TRANSFER RESTRICTIONS PURSUANT TO THAT CERTAIN SPONSOR AGREEMENT, DATED AS OF DECEMBER 7, 2020, BY AND AMONG FOLEY TRASIMENE ACQUISITION CORP. II, A DELAWARE CORPORATION, PAYSAFE LIMITED, AN EXEMPTED LIMITED COMPANY INCORPORATED UNDER THE LAWS OF BERMUDA, PAYSAFE MERGER SUB INC., A DELAWARE CORPORATION AND DIRECT, WHOLLY OWNED SUBSIDIARY OF THE COMPANY, PAYSAFE BERMUDA HOLDING LLC, A BERMUDA EXEMPTED LIMITED LIABILITY COMPANY, PAYSAFE GROUP HOLDINGS LIMITED, A PRIVATE LIMITED COMPANY INCORPORATED UNDER THE LAWS OF ENGLAND AND WALES AND THE OTHER SIGNATORIES THERETO. ANY TRANSFER OF SUCH SHARES OF STOCK OR OTHER SECURITIES IN VIOLATION OF THE TERMS AND PROVISIONS OF SUCH SPONSOR AGREEMENT SHALL BE NULL AND VOID AB INITIO AND HAVE NO FORCE OR EFFECT WHATSOEVER."
- 23. Specific Performance. Each Sponsor Person and each PGHL Person acknowledges and agrees that irreparable damage for which monetary damages, even if available, would not be an adequate remedy, would occur in the event that a Sponsor Person or PGHL Person does not perform its obligations under the provisions of this Sponsor Agreement (including failing to take such actions as are required of them hereunder to consummate this Sponsor Agreement) in accordance with its specified terms or otherwise breach such provisions. Each Sponsor Person and each PGHL Person acknowledges and agrees that (i) each of FTAC and the PGHL Parties shall be entitled to an injunction, specific performance, or other equitable relief, to prevent breaches of this Sponsor Agreement and to enforce specifically the terms and provisions hereof and thereof, without proof of damages, prior to the valid termination of this Sponsor Agreement in accordance with Paragraph 23, this being in addition to any other remedy to which they are entitled under this Sponsor Agreement or any Transaction Agreement, and (ii) the right of specific enforcement is an integral part of the transactions contemplated by this Sponsor Agreement and

without that right, none of the parties hereto would have entered into this Sponsor Agreement. Each Sponsor Person and each PGHL Person agrees that it will not oppose the granting of specific performance and other equitable relief on the basis that the other parties hereto have an adequate remedy at Law or that an award of specific performance is not an appropriate remedy for any reason at Law or equity. Each Sponsor Person and each PGHL Person acknowledges and agrees that any party seeking an injunction to prevent breaches of this Sponsor Agreement and to enforce specifically the terms and provisions of this Sponsor Agreement in accordance with this Paragraph 23 shall not be required to provide any bond or other security in connection with any such injunction.

- 24. <u>Interpretation</u>. Section 1.02 (Construction) and Section 12.05 (Expenses) of the Merger Agreement are incorporated herein by reference, *mutatis mutandis*. Wherever this Sponsor Agreement uses "it", "its" or derivations thereof to refer to natural person Sponsor Persons or PGHL Persons, such references shall be deemed references to "her", "him" or "his", as applicable.
- 25. <u>Updates to Schedule A</u>; <u>Admission of New Sponsor Persons</u>. During the Interim Period, each Sponsor Person shall promptly notify the Company and PGHL of any increase, decrease or other change in the number of Founder Shares, Founder FTAC Warrants, FTAC Class A Common Stock, FP Investor FTAC Warrants or other Covered Shares held by or on behalf of such Sponsor Person (for the avoidance of doubt, such Sponsor Person acknowledges and agrees that Paragraph 6(a) prohibits all Transfers of its Covered Shares during the Interim Period, except as expressly contemplated by the Warrant Contribution). From and after the Closing, each Sponsor Person shall promptly notify the Company of any increase, decrease or other change in the number of Founder Shares or other Covered Shares held by or on behalf of such Sponsor Person, including as a result of a Transfer in compliance with this Sponsor Agreement. Promptly following each such notification, the Company shall update, or cause to be updated, <u>Schedule A</u> to reflect the applicable changes as they relate to Founder Shares, Founder FTAC Warrants, FTAC Class A Common Stock or FP Investor FTAC Warrants (in the case of an Interim Period change) or Company Common Shares and Exchangeable Units (in the case of a post-Closing change) and provide a copy of such updated <u>Schedule A</u> to each of the parties hereto, and such updated <u>Schedule A</u> shall control for all purposes of this Sponsor Agreement (unless and until it is later updated in accordance with this Paragraph 25). Any update to <u>Schedule A</u> in accordance with this Sponsor Agreement shall not be deemed an amendment to this Sponsor Agreement for purposes of Paragraph 12.
- 26. <u>Termination of Existing Registration Rights Agreement</u>. Effective as of (but subject to the consummation of) the Closing, (a) that certain Registration Rights Agreement, dated as of July 17, 2020, by and among FTAC, Sponsor and the other parties thereto is hereby terminated and of no force or effect, and (b) none of the parties thereto shall have any rights or obligations thereunder.

27. FP Agreements.

- (a) Effective as of the date of this Sponsor Agreement, the FP Investor irrevocably waives Section 4 (the Right of First Offer) of that certain Forward Purchase Agreement, dated as of July 31, 2020, between it and FTAC (the "Forward Purchase Agreement") and any entitlement to rights in connection with or as a result thereof, including with respect to the Transactions. The FP Investor further represents and agrees that its rights thereunder have not been exercisable prior to the date hereof in connection with any transactions undertaken by FTAC and that it is not entitled to any equity purchases pursuant thereto.
- (b) Effective as of (but subject to the consummation of) the Closing, the registration rights set forth in Exhibit A to the Forward Purchase Agreement are hereby terminated and of no further force or effect, and none of the parties thereto shall have any rights or obligations thereunder.

- (c) The FP Investor acknowledges and agrees that each of PGHL and the Company shall be an express third party beneficiary of the Forward Purchase Agreement and that each of PGHL and the Company shall be entitled to enforce the terms of such agreement directly against it, including the right to cause the FP Investor to fund the FP Financing in accordance with the terms of the Forward Purchase Agreement.
- 28. <u>Additional Agreements</u>. The Sponsor hereby represents and warrants to FTAC and the PGHL Parties that (i) on or prior to the date hereof, it has delivered to FTAC and PGHL a capitalization table showing all of the direct equity owners of the Sponsor (the "<u>Sponsor Cap Table</u>") and (ii) the Sponsor Cap Table is true, correct and complete in all respects as of the date hereof. Notwithstanding anything to the contrary herein, following the date hereof, the Sponsor shall provide written notice to FTAC and PGHL promptly following any change in the Sponsor Cap Table.
- 29. <u>Further Assurances</u>. Each of the parties hereto agrees to execute and deliver hereafter any further document, agreement or instrument of assignment, transfer or conveyance as may be necessary or desirable to effectuate the purposes hereof and as may be reasonably requested in writing by another party hereto.

[Signature Pages Follows]

IN WITNESS WHEREOF, the parties hereto have executed this Sponsor Agreement on the day and year first above written.

TRASIMENE CAPITAL FT, LP II

By: Trasimene Capital FT, LLC II, its general partner

By: /s/ David W. Ducommun

Name: David W. Ducommun

Title: Senior Vice President, Corporate Finance

TRASIMENE CAPITAL FT, LLC II

By: /s/ David W. Ducommun

Name: David W. Ducommun

Title: Senior Vice President, Corporate Finance

TRASIMENE CAPITAL MANAGEMENT, LLC

By: /s/ David W. Ducommun

Name: David W. Ducommun Title: Managing Director

FOLEY TRASIMENE ACQUISITION CORP. II

By: /s/ David W. Ducommun

Name: David W. Ducommun

Title: Executive Vice President, Corporate Finance

CANNAE HOLDINGS, LLC

By: /s/ Michael L. Gravelle

Name: Michael L. Gravelle

Title: Managing Director, General Counsel and

Corporate Secretary

CANNAE HOLDINGS, INC.

By: /s/ Michael L. Gravelle

Name: Michael L. Gravelle

Title: Executive Vice President, General Counsel and

Corporate Secretary

WILLIAM P. FOLEY, II

/s/ William P. Foley, II

Title: Founder and Chairman

RICHARD N. MASSEY

/s/ Richard N. Massey

Title: Chief Executive Officer and Director

C. MALCOLM HOLLAND, III

/s/ C. Malcolm Holland, III

Title: Director

MARK D. LINEHAN

/s/ Mark D. Linehan

Title: Director

ERIKA MEINHARDT

/s/ Erika Meinhardt

Title: Director

BRYAN D. COY

/s/ Bryan D. Coy

Title: Chief Financial Officer

DAVID W. DUCOMMUN

/s/ David W. Ducommun

Title: Executive Vice President of Corporate Finance

MICHAEL L. GRAVELLE

/s/ Michael L. Gravelle

Title: General Counsel and Corporate Secretary

PAYSAFE LIMITED

By: /s/ Philip McHugh

Name: Philip McHugh
Title: Chief Executive Officer

PAYSAFE BERMUDA HOLDING LLC

By: /s/ Philip McHugh

Name: Philip McHugh Title: Chief Executive Officer

PAYSAFE GROUP HOLDINGS LIMITED

By: /s/ Philip McHugh

Name: Philip McHugh
Title: Chief Executive Officer

PAYSAFE MERGER SUB INC.

By: /s/ Eli Nagler

Name: Eli Nagler Title: President

Schedule A

Ownership of Securities

Sponsor Person	Founder Shares	FTAC Founder Warrants	FTAC Class A Common Stock	FP Investor FTAC Warrants	Exchangeable Units Received in the FTAC LLC Contribution	FTAC Class C Common Stock	Other Covered Shares	Surrendered Founder Shares
Trasimene Capital FT, LP II	36,600,836	20,893,780	_	_	20,893,780	20,893,780		7,971,539
William P. Foley, II	_		_			_		_
Richard N. Massey	_	_	_	_	_	_	_	_
C. Malcolm Holland, III	25,000	_	_			_	_	5,445
Mark D. Linehan	25,000	_	_	_	_	_	_	5,445
Erika Meinhardt	25,000		_			_	_	5,445
Trasimene Capital FT, LLC II	_	_	_	_	_	_	_	_
Trasimene Capital Management, LLC	_	_	_	_	_	_	_	_
Cannae Holdings, Inc.	_	_	_	_	_	_	_	_
Cannae Holdings, LLC			15,000,000	5,000,000				
Total	36,675,836	20,893,780	15,000,000	5,000,000	20,893,780	20,893,780		7,987,877

Schedule B

Fees

1. Fidelity National Title Insurance Company, Chicago Title Insurance Company, Commonwealth Land Title Insurance Company and Fidelity & Guaranty Life Insurance Company (collectively, "FNF") and Cannae Holdings, LLC ("Cannae") are entitled to an aggregate 1.6% fee in connection with their respective PIPE Investments. In the event FTAC pays the Placement Agents (as defined below) the Incentive Fee (as defined in the Engagement Letter (as defined below)), then FNF and Cannae shall be entitled to an aggregate 2.0% fee in connection with their respective PIPE investments.

Exhibit 99.1

FINAL FORM

SHAREHOLDERS AGREEMENT

DATED AS OF [•], 202[•]

AMONG

PAYSAFE LIMITED

AND

THE OTHER PARTIES HERETO

TABLE OF CONTENTS

ARTICLE	E I. INTRODUCTORY MATTERS	Page 1
1.1 1.2	Defined Terms Construction	1 5
ARTICLE	E II. CORPORATE GOVERNANCE MATTERS	6
2.1 2.2 2.3 2.4 2.5	Initial Board Composition Election of Directors Compensation Other Rights of Shareholder Designees Director Independence and Applicable Regulatory Requirements	6 6 9 9 9
ARTICLE	E III. INFORMATION; VCOC	10
3.1 3.2 3.3 3.4 3.5	Books and Records; Access Certain Reports VCOC Confidentiality Information Sharing	10 10 10 13 13
ARTICLE	E IV. ADDITIONAL COVENANTS	13
4.1 4.2 4.3	Pledges or Transfers Spin-Offs or Split-Offs Lock-Up	13 14 14
ARTICLE	E V. GENERAL PROVISIONS	15
5.1 5.2 5.3 5.4	Termination Notices Amendment; Waiver Further Assurances	15 15 15 16
5.5 5.6	Assignment Third Parties	16 17
5.7	Governing Law	17
5.8	Jurisdiction; Waiver of Jury Trial	17
5.9	Specific Performance	17
5.10	Entire Agreement	18

5.11	Severability	18
5.12	Table of Contents, Headings and Captions	18
5.13	Grant of Consent	18
5.14	Counterparts	18
5.15	Effectiveness	18
5.16	No Recourse	18

SHAREHOLDERS AGREEMENT

This Shareholders Agreement is entered into as of [•], 202[•] by and among Paysafe Limited, an exempted limited company incorporated under the laws of Bermuda (the "Company"), Pi Jersey Topco Limited, a company incorporated in Jersey ("Pi Topco"), Paysafe Group Holdings Limited, a private limited company incorporated under the laws of England and Wales ("PGHL") and each of the Principal Shareholders (as defined below) from time to time party hereto.

RECITALS:

WHEREAS, in connection with the Equity Transactions (as defined below) and effective upon the consummation thereof, the parties hereto wish to set forth certain understandings between such parties in relation to the Company, including with respect to certain governance of the Company and other matters.

NOW, THEREFORE, the parties agree as follows:

ARTICLE I. INTRODUCTORY MATTERS

- 1.1 <u>Defined Terms</u>. In addition to the terms defined elsewhere herein, the following terms have the following meanings when used herein with initial capital letters. Capitalized terms used but not defined herein shall have the respective meanings given to them in the Merger Agreement.
 - "Affiliate" has the meaning set forth in Rule 12b-2 promulgated under the Exchange Act, as in effect on the date hereof.
- "Agreement" means this Shareholders Agreement, as the same may be amended, supplemented, restated or otherwise modified from time to time in accordance with the terms hereof.
 - "Beneficially Own" has the meaning set forth in Rule 13d-3 promulgated under the Exchange Act.
 - "Blackstone Investors Designee" has the meaning assigned to such term in Section 2.2(b).
- "Blackstone Investors Designator" means the Blackstone Investors, or any group of Blackstone Investors collectively then directly holding or indirectly, as set forth in the books and records of PGHL or Pi Topco, as applicable, being attributed a majority of the Common Shares held by all Blackstone Investors.
- "Blackstone Investors" means the entities listed on the signature pages hereto under the heading "Blackstone Investors," any Transferee that becomes party to this Agreement as a "Blackstone Investor" in accordance with Section 5.5 hereof, and their respective Affiliates.

"Board" means the board of directors of the Company from time to time.

"Business Day" means a day other than a Saturday, Sunday, federal or New York State holiday or other day on which commercial banks in New York City are authorized or required by law to close.

"Cannae" means Cannae Holdings, LLC, a Delaware limited liability company.

"Common Shares" means common shares, par value \$0.001 per share, of the Company, and any securities issued in respect thereof, or in substitution therefor, in connection with any share split, dividend or combination, or any reclassification, recapitalization, merger, amalgamation, consolidation or similar transaction.

"Closing Date" means the date of the closing of the Equity Transactions.

"Company" has the meaning set forth in the Preamble.

- "Confidential Information" means any proprietary, business relevant or sensitive information concerning the Company or its Subsidiaries that is furnished after the date of this Agreement by or on behalf of the Company or its designated representatives to a Principal Shareholder or its designated representatives, together with any notes, analyses, reports, models, compilations, studies, documents, records or extracts thereof containing, based upon or derived from such information, in whole or in part; *provided*, *however*, that Confidential Information does not include information:
- (i) that is or has become publicly available other than as a result of a disclosure by a Principal Shareholder or its designated representatives in violation of this Agreement;
- (ii) that was already known to a Principal Shareholder or its designated representatives or was in the possession of a Principal Shareholder or its designated representatives prior to its being furnished by or on behalf of the Company or its designated representatives;
- (iii) that is received by a Principal Shareholder or its designated representatives from a source other than the Company or its designated representatives, provided, that the source of such information was not actually known by such Principal Shareholder or designated representative to be bound by a confidentiality agreement with, or other contractual obligation of confidentiality to, the Company;
- (iv) that was independently developed or acquired by a Principal Shareholder or its designated representatives or on its or their behalf without the violation of the terms of this Agreement; or
- (v) that a Principal Shareholder or its designated representatives is required, in the good faith determination of such Principal Shareholder or designated representative, to disclose by applicable law, regulation or legal process, provided, that such Principal Shareholder or designated representative first takes reasonable steps to minimize the extent of any such

required disclosure, and provided, further, that no such steps to minimize disclosure shall be required where disclosure is made (a) in response to a request by a regulatory or self-regulatory authority of competent authority or (b) in connection with an audit or examination by a bank examiner or auditor or regulatory authority and such audit or examination does not specifically reference the Company or this Agreement.

"Control" (including its correlative meanings, "Controlled by" and "under common Control with") means possession, directly or indirectly, of the power to direct or cause the direction of management or policies (whether through ownership of securities or partnership or other ownership interests, by contract or otherwise) of a Person.

"Covered Shares" means all Equity Securities of which PGHL, Pi Topco and/or the CVC Investors or the Blackstone Investors, as applicable, acquires legal or beneficial ownership, including by purchase, as a result of a share dividend, share split, recapitalization, combination, reclassification, exchange or change of such shares, or upon exercise or conversion of any securities, other than any Equity Securities purchased on the open market after the Closing Date.

"CVC Investors Designee" has the meaning assigned to such term in Section 2.2(a).

"CVC Investors Designator" means the CVC Investors, or any group of CVC Investors collectively then directly holding or indirectly, as set forth in the books and records of PGHL or Pi Topco, as applicable, being attributed a majority of the Common Shares held by all CVC Investors.

"CVC Investors" means the entities listed on the signature pages hereto under the heading "CVC Investors," any Transferee that becomes party to this Agreement as a "CVC Investor" in accordance with Section 5.5 hereof, and their respective Affiliates.

"Director" means any director of the Company from time to time.

"<u>Equity Securities</u>" means any and all Common Shares of the Company, and any and all securities of the Company convertible into, or exchangeable or exercisable for (whether or not subject to contingencies or the passage of time, or both), such shares, and options, warrants or other rights to acquire Common Shares of the Company.

"Equity Transactions" means the transactions contemplated by the Merger Agreement.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder, as the same may be amended from time to time.

"FTAC Designee" has the meaning assigned to such term in Section 2.2(c).

"FTAC Designators" means the FTAC Investors, who shall jointly designate the FTAC Designee.

- "FTAC Investors" means Cannae and FTAC Sponsor.
- "FTAC Sponsor" means Trasimene Capital FT, LP II, a Delaware limited partnership.
- "Governmental Authority" means any federal, state, provincial, municipal, local or foreign government, governmental authority, regulatory or administrative agency, governmental commission, department, board, bureau, agency or instrumentality, court or tribunal.
 - "Information" has the meaning set forth in Section 3.1 hereof.
 - "Jointly Designated Directors" has the meaning set forth in Section 2.1 hereof.
- "<u>Law</u>" means any statute, law, ordinance, rule, treaty, code, directive, regulation, governmental approval (whether granted or required) or Governmental Order, in each case, of any Governmental Authority.
 - "Lock-Up Period" has the meaning set forth in Section 4.3 hereof.
- "Merger Agreement" means the Agreement and Plan of Merger, dated as of December 7, 2020, by and among Foley Trasimene Acquisition Corp. II, the Company, PGHL and the other parties thereto.
 - "NewCo" has the meaning set forth in Section 4.2 hereof.
 - "Non-Recourse Party" has the meaning set forth in Section 5.16 hereof.
 - "Permitted Transferee" has the meaning set forth in Section 4.3 hereof.
- "Person" means any individual, firm, corporation, partnership, limited liability company, incorporated or unincorporated association, joint venture, joint stock company, governmental agency or instrumentality or other entity of any kind.
 - "PGHL" means Paysafe Group Holdings Limited, a private limited company incorporated under the laws of England and Wales.
 - "Pi Topco" means Pi Jersey Topco Limited, a company incorporated in Jersey.
 - "Plan Asset Regulation" has the meaning set forth in Section 3.3(a) hereof.
 - "Principal Shareholders" means (i) the CVC Investors, (ii) the Blackstone Investors and (iii) the FTAC Investors.
 - "Shareholder Designator" has the meaning assigned to such term in Section 2.2(d).
 - "Shareholder Designee" means any CVC Investors Designee, Blackstone Investors Designee or FTAC Designee.

"Subsidiary" means, with respect to any Person, any corporation, company, limited liability company, partnership, association or other business entity of which: (i) if a corporation, a majority of the total voting power of shares of stock entitled (without regard to the occurrence of any contingency) to vote in the election of directors, representatives or trustees thereof is at the time owned or Controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or any combination thereof; or (ii) if a limited liability company, partnership, association or other business entity is at the time owned or Controlled, directly or indirectly, by that Person or one or more Subsidiaries of that Person or any combination thereof. For purposes hereof, a Person or Persons shall be deemed to have a majority ownership interest in a limited liability company, partnership, association or other business entity if such Person or Persons shall (a) be allocated a majority of limited liability company, partnership, association or other business entity gains or losses or shall be or (b) Control the managing member, managing director or other governing body or general partner of such limited liability company, partnership, association or other business entity.

"Total Number of Directors" means the total number of directors comprising the Board from time to time.

"Transfer" (including its correlative meanings, "Transferor," "Transferee" and "Transferred") shall mean, with respect to any security, directly or indirectly, to sell, contract to sell, give, assign, hypothecate, pledge, encumber, grant a security interest in, offer, sell any option or contract to purchase, purchase any option or contract to sell, grant any option, right or warrant to purchase, lend or otherwise transfer or dispose of any economic, voting or other rights in or to such security. When used as a noun, "Transfer" shall have such correlative meaning as the context may require.

"VCOC Investor" has the meaning set forth in Section 3.3(a) hereof.

"<u>Voting Power</u>" means, at any time, the voting power of all shares of outstanding share capital entitled to vote generally in the election of directors of the Company as of the record date for such meeting.

1.2 Construction. The language used in this Agreement will be deemed to be the language chosen by the parties to express their mutual intent, and no rule of strict construction will be applied against any party. Unless the context otherwise requires: (a) "or" is disjunctive but not exclusive, (b) words in the singular include the plural, and in the plural include the singular, and (c) the words "hereof," "herein," and "hereunder" and words of similar import when used in this Agreement refer to this Agreement as a whole and not to any particular provision of this Agreement, and Section references are to sections of this Agreement unless otherwise specified.

ARTICLE II. CORPORATE GOVERNANCE MATTERS

2.1 <u>Initial Board Composition</u>. Effective as of the Closing Date, the Board is anticipated to be comprised of eleven Directors, as follows: (i) the Chief Executive Officer of PGHL, (ii) four Directors designated by the CVC Designator and the Blackstone Designator, (iii) four Directors jointly designated by the FTAC Designators and (iv) two Directors, to be jointly designated as mutually agreed by the CVC Designator, the Blackstone Designator and Cannae, who shall be independent as required by the Securities and Exchange Commission and applicable listing exchange rules and regulations (such two Directors, the "<u>Jointly Designated Directors</u>"). A majority of the Directors shall be neither a citizen nor a resident of the United States.

2.2 Election of Directors.

- (a) Following the Closing Date, the CVC Designator shall have the right, but not the obligation, to designate, and the individuals nominated for election as Directors by or at the direction of the Board or a duly-authorized committee thereof shall include: (i) if the CVC Investors collectively directly hold or indirectly, as set forth in the books and records of PGHL or Pi Topco, as applicable, are attributed at least 7.5% of the aggregate outstanding Common Shares, two Directors and (ii) if the CVC Investors collectively directly hold or indirectly, as set forth in the books and records of PGHL or Pi Topco, as applicable, are attributed at least 2.5% (but less than 7.5%) of the aggregate outstanding Common Shares, one Director, which Director may be a U.S. citizen or resident (in each case, each such person a "CVC Designee"). In addition, if the CVC Investors collectively directly hold or indirectly, as set forth in the books and records of PGHL or Pi Topco, as applicable, are attributed at least 7.5% of the aggregate outstanding Common Shares, the CVC Designator shall have the right, but not the obligation, to (i) jointly with the Blackstone Designator and Cannae, designate the Jointly Designated Directors and (ii) to consent to any individual nominated for election to the Board seat initially occupied by the Chief Executive Officer of PGHL.
- (b) Following the Closing Date, the Blackstone Designator shall have the right, but not the obligation, to designate, and the individuals nominated for election as Directors by or at the direction of the Board or a duly-authorized committee thereof shall include: (i) if the Blackstone Investors collectively directly hold or indirectly, as set forth in the books and records of PGHL or Pi Topco, as applicable, are attributed at least 7.5% of the aggregate outstanding Common Shares, two Directors and (ii) if the Blackstone Investors collectively directly hold or indirectly, as set forth in the books and records of PGHL or Pi Topco, as applicable, are attributed at least 2.5% (but less than 7.5%) of the aggregate outstanding Common Shares, one Director, which Director may be a U.S. citizen or resident (in each case, each such person a "Blackstone Designee"). In addition, if the Blackstone Investors collectively directly hold or indirectly, as set forth in the books and records of PGHL or Pi Topco, as applicable, are attributed at least 7.5% of the aggregate outstanding Common Shares, the Blackstone Designator shall have the right, but not the obligation, to (i) jointly with the CVC Designator and Cannae, designate the Jointly Designated Directors and (ii) to consent to any individual nominated for election to the Board seat initially occupied by the Chief Executive Officer of PGHL.

- (c) Following the Closing Date, for so long as the FTAC Investors collectively continue to hold at least 50% of the aggregate outstanding Common Shares held by such FTAC Investors as of the Closing Date, the FTAC Designators shall have the right, but not the obligation, to designate, and the individuals nominated for election as Directors by or at the direction of the Board or a duly-authorized committee thereof shall include, four Directors and Cannae shall have the right, but not the obligation, to (i) jointly with the Blackstone Designator and the CVC Designator, designate the Jointly Designated Directors and (ii) to consent to any individual nominated for election to the Board seat initially occupied by the Chief Executive Officer of PGHL. If and when the FTAC Investors collectively hold less than 50% of the aggregate outstanding Common Shares held by the FTAC Investors as of the Closing Date, the FTAC Designators shall have the right, but not the obligation, to designate, and the individuals nominated for election as Directors by or at the direction of the Board or a duly-authorized committee thereof shall include: (i) if the FTAC Investors collectively hold at least 7.5% of the aggregate outstanding Common Shares, four Directors; (ii) if the FTAC Investors collectively hold at least 6.25% (but less than 7.5%) of the aggregate outstanding Common Shares, two Directors; and (iii) if the FTAC Investors collectively hold at least 2.5% (but less than 6.25%) of the aggregate outstanding Common Shares, one Director, which Director may be a U.S. citizen or resident (in each case, each such person a "FTAC Designee"). In addition, if the FTAC Investors collectively hold at least 7.5% of the aggregate outstanding Common Shares, Cannae shall have the right, but not the obligation, to (i) jointly with the Blackstone Designator and the CVC Designator, designate the Jointly Designated Directors and (ii) to consent to any individual nominated for election to the Board seat initially occupied by the Chief Executive Officer of PGHL. Further, for so long as the Company remains a "foreign private issuer" as defined in Rule 3b-4 under the Exchange Act, at any time the FTAC Designators have the right under this Section 2.2(c) to appoint more than one Director, at least one of the FTAC Designees shall be neither a citizen nor a resident of the United States.
- (d) If at any time the CVC Designator, the Blackstone Designator, the FTAC Designators or Cannae (each, a "Shareholder Designator") has designated fewer than the total number of individuals that it is then entitled to designate pursuant to Section 2.2(a), Section 2.2(b) or Section 2.2(c) hereof, the CVC Designator, the Blackstone Designator, the FTAC Designators or Cannae, as applicable, shall have the right, at any time and from time to time, to designate such additional individuals which it is entitled to so designate, in which case, any individuals nominated by or at the direction of the Board or any duly-authorized committee thereof for election as Directors to fill any vacancy on the Board shall include such designees, and the Company shall use its best efforts to (i) effect the election of such additional designees, whether by increasing the size of the Board or otherwise, and (ii) cause the election of such additional designees to fill any such newly-created vacancies or to fill any other existing vacancies.
- (e) Directors are subject to removal pursuant to the applicable provisions of the Amended and Restated Bye-Laws of the Company; *provided, however*, for as long as this Agreement remains in effect, the CVC Designees may only be removed with the consent of the CVC Designator, the Blackstone Designees may only be removed with the consent of the FTAC Designees may only be removed with the consent of the FTAC Designator, in each case delivered in accordance with <u>Section 5.13</u> hereof.

- (f) In the event that a vacancy is created at any time by death, disability, retirement, removal (with or without cause), disqualification, resignation or otherwise with respect to the CVC Designees, the Blackstone Designees or the FTAC Designees, any individual nominated by or at the direction of the Board or any duly-authorized committee thereof to fill such vacancy shall be, and the Company shall use its best efforts to cause such vacancy to be filled, as soon as reasonably possible, by a new designee of the CVC Designator, the Blackstone Designator or the FTAC Designators, as applicable.
- (g) The Company shall, to the fullest extent permitted by applicable Law, include in the slate of nominees recommended by the Board at any meeting of shareholders called for the purpose of electing directors (or consent in lieu of meeting), the persons designated pursuant to this Section 2.2 and use its reasonable best efforts to cause the election of each such designee to the Board, including nominating each such individual to be elected as a Director as provided herein, recommending such individual's election and soliciting proxies or consents in favor thereof. In the event that any Shareholder Designee shall fail to be elected to the Board at any meeting of shareholders called for the purpose of electing directors (or written resolution in lieu of a meeting), the Company shall use its reasonable best efforts to cause such Shareholder Designee (or a new designee of the applicable Shareholder Designator) to be elected to the Board, as soon as possible, and the Company shall take or cause to be taken, to the fullest extent permitted by applicable Law, at any time and from time to time, all actions necessary to accomplish the same, including, without limitation, actions to effect an increase in the Total Number of Directors.
- (h) Each Principal Shareholder (or PGHL and/or Pi Topco on behalf of and at the direction of any Principal Shareholder, as applicable) hereby agrees to vote in favor of and to consent to the Shareholder Designees in connection with each vote taken or written resolution executed in connection with the election of Directors to the Board, and each Principal Shareholder agrees not to seek to remove or replace the Shareholder Designees.
- (i) In addition to any vote or written resolution of the Board or the shareholders of the Company required by applicable Law or the memorandum of association or bye-laws of the Company, and notwithstanding anything to the contrary in this Agreement, for so long as this Agreement is in effect, any action by the Board to increase the Total Number of Directors to greater than eleven (other than any increase in the Total Number of Directors in connection with the election of one or more Directors elected exclusively by the holders of one or more classes or series of the Company's shares other than Common Shares) shall require the prior written consent of (i) the CVC Designator, for so long as the CVC Investors collectively directly hold or indirectly, as set forth in the books and records of PGHL or Pi Topco, as applicable, are attributed at least 7.5% of the aggregate outstanding Common Shares, (ii) the Blackstone Designator, for so long as the Blackstone Investors collectively directly hold or indirectly, as set forth in the books and records of PGHL or Pi Topco, as applicable, are attributed at least 7.5% of the aggregate outstanding Common Shares, and (iii) the FTAC Designators, for so long as the FTAC Investors collectively hold at least 7.5% of the aggregate outstanding Common Shares, in each case delivered in accordance with Section 5.13 hereof.

- 2.3 Compensation. Except to the extent the CVC Designator, the Blackstone Designator or the FTAC Designators may otherwise notify the Company with respect to the CVC Designees, the Blackstone Designees or the FTAC Designees, respectively, any Shareholder Designees shall be entitled to compensation consistent with the Director compensation received by other Directors, including any fees and equity awards, provided, that (x) to the extent any Director compensation is payable in the form of equity awards, at the election of a Shareholder Designee that is an employee or affiliate of a Principal Shareholder, in lieu of any equity award, such compensation shall be paid in an amount of cash equal to the value of the equity award as of the date of the award, with any such cash subject to the same vesting terms, if any, as the equity awarded to other Directors and (y) at the election of a Shareholder Designee that is an employee or affiliate of a Principal Shareholder, any Director compensation (whether cash, equity awards and/or cash in lieu of equity as may be designated by the electing Shareholder Designee) shall be paid to a Principal Shareholder or an Affiliate thereof specified by such Shareholder Designee rather than to such Shareholder Designee. If the Company adopts a policy that Directors own a minimum amount of equity in the Company, any Shareholder Designee that is an employee or affiliate of a Principal Shareholder shall not be subject to such policy unless otherwise determined by such Principal Shareholder in its sole discretion.
- 2.4 Other Rights of Shareholder Designees. Except as provided in Section 2.3, each Shareholder Designee serving on the Board shall be entitled to the same rights and privileges applicable to all other members of the Board generally or to which all such members of the Board are entitled. In furtherance of the foregoing, the Company shall, to the maximum extent permitted by applicable Law, indemnify, exculpate, and reimburse fees and expenses of the Shareholder Designees (including by entering into an indemnification agreement in a form substantially similar to the Company's form director indemnification agreement) and provide the Shareholder Designees with director and officer insurance to the same extent it indemnifies, exculpates, reimburses and provides insurance for the other members of the Board pursuant to the memorandum of association or bye-laws of the Company, applicable Law or otherwise.
- 2.5 <u>Director Independence and Applicable Regulatory Requirements</u>. Notwithstanding anything to the contrary herein, the parties hereto shall ensure that the composition of the Board will continue to meet all requirements for a company listed on the New York Stock Exchange (or such other stock exchange on which the Common Shares may be listed from time to time), including with respect to director independence, and any other Laws or requirements of a Governmental Authority applicable to members of the Board. In the event a Shareholder Designee is prevented from serving as a Director by a Governmental Authority with the ability to so prevent, each of the Company, Pi Topco, PGHL and each Principal Shareholder shall use their respective commercially reasonable efforts to cause such Governmental Authority to remove such restriction; <u>provided</u>, that, (x) there shall be no obligation for such person to be added as a member of the Board and (y) the Shareholder Designator that had designated such Shareholder Designee shall be entitled to designate a replacement Director in lieu of such person. In addition, the parties hereto acknowledge and agree that: (i) there is no agreement, arrangement or understanding for any of the Directors to vote or act together; (ii) that each Director shall act individually in respect of the matters that come before the Board consistent with the fiduciary duties of each Director; and (iii) each Director shall at all times act in accordance with applicable Law and shall refrain from taking any action which would cause the Principal Shareholders to contravene applicable Law.

ARTICLE III. INFORMATION; VCOC

- 3.1 Books and Records; Access. The Company shall, and shall cause its Subsidiaries to, keep proper books, records and accounts, in which full and correct entries shall be made of all financial transactions and the assets and business of the Company and each of its Subsidiaries in accordance with generally accepted accounting principles. The Company shall, and shall cause its Subsidiaries to, (a) permit the Principal Shareholders and their respective designated representatives (or other designees), at reasonable times and upon reasonable prior notice to the Company, to review the books and records of the Company or any of such Subsidiaries and to discuss the affairs, finances and condition of the Company or any of such Subsidiaries with the officers of the Company or any such Subsidiary, (b) host regular conference calls for the Principal Shareholders with senior officers of the Company upon request, and (c) provide the Principal Shareholders all information of a type, at such times and in such manner as is consistent with the Company's past practice or that is otherwise reasonably requested by such Principal Shareholders from time to time (all such information so furnished pursuant to this Section 3.1, the "Information"). Subject to Section 3.4, any Principal Shareholder (and any party receiving Information from a Principal Shareholder) who shall receive Information shall maintain the confidentiality of such Information. Notwithstanding the foregoing, that the Company shall not be required to disclose any privileged or Confidential Information of the Company so long as the Company has used commercially reasonable efforts to enter into an arrangement pursuant to which it may provide such information to the Principal Shareholders without the loss of any such privilege or otherwise as provided for in a confidentiality agreement between the parties.
 - 3.2 Certain Reports. The Company shall deliver or cause to be delivered to the Principal Shareholders, at their request:
- (a) to the extent otherwise prepared by the Company, operating and capital expenditure budgets and periodic information packages relating to the operations and cash flows of the Company and its Subsidiaries; and
- (b) to the extent otherwise prepared by the Company, such other reports and information as may be reasonably requested by the Principal Shareholders; *provided*, *however*, that the Company shall not be required to disclose any privileged or Confidential Information of the Company so long as the Company has used commercially reasonable efforts to enter into an arrangement pursuant to which it may provide such information to the Principal Shareholders without the loss of any such privilege or otherwise as provided for in a confidentiality agreement.

3.3 VCOC.

(a) With respect to each Principal Shareholder or Affiliate thereof that is intended to qualify its direct or indirect investment in the Company as a "venture capital investment" as defined in the Department of Labor regulations codified at 29 CFR Section 2510.3-101 (the "Plan Asset Regulation") (each, a "VCOC Investor"), for so long as the VCOC Investor, directly or through one or more subsidiaries, continues to hold any Common Shares (or other securities of the Company into which such Common Shares may be exchanged), without limitation or prejudice of any of the rights provided to the Principal Shareholders hereunder, the Company shall, with respect to each such VCOC Investor:

- (i) provide each VCOC Investor or its designated representative with:
- (A) upon reasonable notice and at mutually convenient times, the right to visit and inspect any of the offices and properties of the Company and its Subsidiaries and inspect and copy the books and records of the Company and its Subsidiaries;
- (B) as soon as available and in any event within 45 days after the end of each of the first three quarters of each fiscal year of the Company, consolidated balance sheets of the Company and its Subsidiaries as of the end of such period, and consolidated statements of income and cash flows of the Company and its Subsidiaries for the period then ended prepared in conformity with generally accepted accounting principles in the United States applied on a consistent basis, except as otherwise noted therein, and subject to the absence of footnotes and to year-end adjustments;
- (C) as soon as available and in any event within 120 days after the end of each fiscal year of the Company, a consolidated balance sheet of the Company and its Subsidiaries as of the end of such year, and consolidated statements of income and cash flows of the Company and its Subsidiaries for the year then ended prepared in conformity with generally accepted accounting principles in the United States applied on a consistent basis, except as otherwise noted therein, together with an auditor's report thereon of a firm of established national reputation;
- (D) to the extent the Company is required by applicable Law or pursuant to the terms of any outstanding indebtedness of the Company to prepare such reports, any annual reports, quarterly reports and other periodic reports pursuant to Section 13 or 15(d) of the Exchange Act, actually prepared by the Company as soon as available; and
- (E) upon written request by the VCOC Investor, copies of all materials provided to the Board, subject to appropriate protections with respect to confidentiality and preservation of attorney-client privilege; *provided*, that, in each case, if the Company makes the information described in <u>clauses (B)</u>, (C) and (D) of this <u>Section 3.3(a)(i)</u> available through public filings on the EDGAR System or any successor or replacement system of the U.S. Securities and Exchange Commission, the requirement to deliver such information shall be deemed satisfied;
- (ii) make appropriate officers and/or Directors of the Company available, and cause the officers and directors of its Subsidiaries to be made available, periodically and at such times as reasonably requested by each VCOC Investor, upon reasonable notice and at mutually convenient times, for consultation with such VCOC Investor or its designated representative with respect to matters relating to the business and affairs of the Company and its Subsidiaries;

(iii) to the extent that the VCOC Investor requests to receive such information and rights, and to the extent consistent with applicable Law or listing standards (and with respect to events which require public disclosure, only following the Company's public disclosure thereof through applicable securities law filings or otherwise), inform each VCOC Investor or its designated representative in advance with respect to any significant corporate actions, and to provide (or cause to be provided) each VCOC Investor or its designated representative with the right to consult with the Company and its Subsidiaries with respect to such actions should the VCOC Investor elect to do so; provided, however, that this right to consult must be exercised within five days after the Company informs the VCOC Investor of the proposed corporate action; provided, further, that the Company shall be under no obligation to provide the VCOC Investor with any material non-public information with respect to such corporate action; and

- (iv) provide each VCOC Investor or its designated representative with such other rights of consultation which the VCOC Investor's counsel may determine in writing to be reasonably necessary under applicable legal authorities promulgated after the date hereof to qualify its investment in the Company as a "venture capital investment" for purposes of the Plan Asset Regulation; *provided* that the parties agree that any such rights of consultation shall be of a nature consistent with those granted above and nothing in this Agreement shall be deemed to require the Company to grant to the VCOC Investor any additional rights with respect to the governance or management of the Company.
- (b) The Company agrees to consider, in good faith, the recommendations of each VCOC Investor or its designated representative in connection with the matters on which it is consulted as described above in this <u>Section 3.3</u>, recognizing that the ultimate discretion with respect to all such matters shall be retained by the Company.
- (c) In the event a VCOC Investor or any of its Affiliates Transfers all or any portion of their investment in the Company to an Affiliated entity that is intended to qualify its investment in the Company as a "venture capital investment" (as defined in the Plan Asset Regulation), such Transferee shall be afforded the same rights with respect to the Company afforded to the VCOC Investor hereunder and shall be treated, for such purposes, as a third party beneficiary hereunder.
- (d) In the event that the Company ceases to qualify as an "operating company" (as defined in the first sentence of 2510.3-101(c)(1) of the Plan Asset Regulation), or the investment in the Company by a VCOC Investor does not qualify as a "venture capital investment" as defined in the Plan Asset Regulation, then the Company and each Principal Shareholder will cooperate in good faith and take all reasonable actions necessary, subject to applicable Law, to preserve the VCOC status of each VCOC Investor or the qualification of the investment as a "venture capital investment," it being understood that such reasonable actions shall not require a VCOC Investor to purchase or sell any investments.
- (e) For so long as the VCOC Investor, directly or through one or more subsidiaries, continues to hold any Common Shares (or other securities of the Company into which such Common Shares may be converted or for which such Common Shares may be exchanged) and upon the written request of such VCOC Investor, without limitation or prejudice of any of the rights provided to the Principal Shareholders hereunder, the Company shall, with respect to each such VCOC Investor, furnish and deliver a letter covering the matters set forth in Sections 3.3(a), 3.3(b), 3.3(c) and 3.3(d) hereof in a form and substance satisfactory to such VCOC Investor.

- (f) In the event a VCOC Investor is an Affiliate of a Principal Shareholder, as described in <u>Section 3.3(a)</u> above, such affiliated entity shall be afforded the same rights with respect to the Company and afforded to the Principal Shareholder under this <u>Section 3.3</u> and shall be treated, for such purposes, as a third party beneficiary hereunder.
- 3.4 <u>Confidentiality</u>. Each Principal Shareholder agrees that it will, and will direct its designated representatives to, keep confidential and not disclose any Confidential Information; *provided*, *however*, that such Principal Shareholder and its designated representatives may disclose Confidential Information to the other Principal Shareholders, to the Shareholder Designees and to (a) its Affiliates and its Affiliates' attorneys, accountants, consultants, insurers, financing sources and other advisors in connection with such Principal Shareholder's investment in the Company, (b) any Person, including a prospective purchaser of Common Shares, as long as such Person has first agreed, in writing, to maintain the confidentiality of such Confidential Information, (c) any of such Principal Shareholder's or its respective Affiliates' partners, members, stockholders, directors, officers, employees or agents who have the need to know such Confidential Information (the Persons referenced in clauses (a), (b) and (c), a Principal Shareholder's "designated representatives") or (d) as the Company may otherwise consent in writing; *provided*, *further*, *however*, that each Principal Shareholder agrees to be responsible for any breaches of this <u>Section 3.4</u> by such Principal Shareholder's designated representatives.
- 3.5 <u>Information Sharing</u>. Each party hereto acknowledges and agrees that Shareholder Designees may share any information concerning the Company and its Subsidiaries received by them from or on behalf of the Company or its designated representatives with each Principal Shareholder and its designated representatives (subject to such Principal Shareholder's obligation to maintain the confidentiality of Confidential Information in accordance with <u>Section 3.4</u>).

ARTICLE IV. ADDITIONAL COVENANTS

4.1 <u>Pledges or Transfers</u>. Upon the request of any Principal Shareholder or at the request of Pi Topco or PGHL acting on behalf of and at the direction of any Principal Shareholder that wishes to (x) pledge, charge, hypothecate or grant security interests in any or all of the shares of Common Shares held by it including to banks or financial institutions as collateral or security for loans, advances or extensions of credit or (y) sell or transfer any or all of the shares of Common Shares held by it, including to a third party investor, the Company agrees to cooperate with such Principal Shareholder or at the request of Pi Topco or PGHL acting on behalf of and at the direction of any Principal Shareholder, as applicable, in taking any action reasonably necessary to consummate any such pledge, charge, hypothecation, grant or transfer, including without limitation, delivery of letter agreements to lenders in form and substance reasonably satisfactory to such lenders (which may include agreements by the Company in respect of the exercise of remedies by such lenders), instructing the transfer agent to transfer any such Common Shares subject to the pledge, charge, hypothecation or grant into the facilities of The Depository Trust Company without restricted legends and cooperating in diligence or other matters as may reasonably requested by any Principal Shareholder in connection with a proposed transfer.

4.2 <u>Spin-Offs or Split-Offs</u>. In the event that the Company effects the separation of any portion of its business into one or more entities (each, a "<u>NewCo</u>"), whether existing or newly formed, including without limitation by way of spin-off, split-off, carve-out, demerger, recapitalization, reorganization or similar transaction, and any Principal Shareholder or Pi Topco or PGHL (on behalf of any Principal Shareholder) will receive equity interests in any such NewCo as part of such separation, the Company shall cause any such NewCo to enter into a Shareholders agreement with the Principal Shareholders, Pi Topco and PGHL that provides the Principal Shareholders and Pi Topco or PGHL, as applicable, with rights vis-à-vis such NewCo that are substantially identical to those set forth in this Agreement.

4.3 Lock-Up. For the period beginning on the Closing Date until the earlier of (i) 180 days thereafter or (ii) if the VWAP of the Common Shares equals or exceeds \$12.00 per share (as adjusted for share splits, share capitalizations, reorganizations, recapitalizations and the like) for any twenty (20) Trading Days within a period of thirty (30) consecutive Trading Days, 60 days thereafter (such applicable period, the "Lock-Up Period"), the CVC Investors and the Blackstone Investors shall not, and shall cause any other holder of record of any of the CVC Investors' and the Blackstone Investors' Covered Shares not to, Transfer any of the CVC Investors' and the Blackstone Investors' Covered Shares. Notwithstanding the immediately preceding sentence, post-Closing Transfers of Covered Shares that are held by any of the CVC Investors, the Blackstone Investors or any of their Permitted Transferees (as defined below) that have entered into a written agreement contemplated by the proviso in this Section 4.3 are permitted (i) to any investment fund or other entity controlled or managed by such CVC Investor or Blackstone Investor, as applicable, to such CVC Investor's or Blackstone Investor's, as applicable, officers or directors, any Affiliates or family members of any of such CVC Investor's or Blackstone Investor's, as applicable, officers or directors, any limited partners, members or stockholders of such CVC Investor or Blackstone Investor, as applicable, or any Affiliates of the CVC Investors or the Blackstone Investors, as applicable, or any employees of such Affiliates; (ii) in the case of an individual, by gift to a member of the individual's immediate family, or to a trust, the beneficiary of which is a member of the individual's immediate family, an Affiliate of such Person, or to a charitable organization; (iii) in the case of an individual, by virtue of laws of descent and distribution upon death of the individual; (iv) in the case of an individual, pursuant to a qualified domestic relations order; (v) by virtue of the Laws of the jurisdiction of incorporation or formation of such CVC Investor or Blackstone Investor, as applicable, or the organizational documents of such CVC Investor or Blackstone Investor, as applicable, as amended from time to time, upon dissolution of such CVC Investor or Blackstone Investor, as applicable; or (vi) in the event of the Company's completion of a liquidation, merger, amalgamation, share exchange, reorganization or other similar transaction which results in the holders of all of the Common Shares having the right to exchange their shares for cash, securities or other property subsequent to the completion of the Equity Transactions (including the entry into an agreement in connection with such liquidation, merger, amalgamation, share exchange, reorganization or other similar transaction); provided, however, that each transferee contemplated by clauses (i) through (v) (each, a "Permitted Transferee") must enter into a written agreement with the Company agreeing to be bound by the restrictions in this <u>Section 4.3</u>. The parties hereto agree that, if the lock-up provisions in Section 6(b) of the Sponsor Agreement, dated as of December 7, 2020 by and among Foley

Trasimene Acquisition Corp. II, Paysafe Limited, Paysafe Bermuda Holding LLC and the other parties that are signatories thereto are modified in a manner that is favorable to the FTAC Investors, the corresponding modifications shall automatically apply to this <u>Section 4.3</u>. Any Transfer in violation of the provisions of this <u>Section 4.3</u> shall be null and void ab initio and of no force or effect.

ARTICLE V. GENERAL PROVISIONS

5.1 <u>Termination</u>. Subject to the early termination of any provision as a result of an amendment to this Agreement agreed to by the Board and the Principal Shareholders, as provided under <u>Section 5.3</u>, and except for <u>Section 3.3</u> hereof, this Agreement, excluding Article V hereof, shall terminate with respect to each Principal Shareholder at such time as such Principal Shareholder and its Affiliates collectively directly hold or indirectly, as set forth in the books and records of PGHL or Pi Topco, as applicable, are attributed less than 2.5% of the aggregate outstanding Common Shares or such earlier time as such Principal Shareholder shall deliver a written notice to the Company requesting that this Agreement terminate with respect to such Principal Shareholder in accordance with <u>Section 5.3(d)</u>. The VCOC Investors shall advise the Company when they collectively first cease to hold any Common Shares (or other securities of the Company into which such Common Shares may be converted or for which such Common Shares may be exchanged), whereupon Section 3.3 hereof shall terminate.

5.2 Notices. Any notice, designation, request, request for consent or consent provided for in this Agreement shall be in writing and shall be either personally delivered, sent by facsimile or sent by reputable overnight courier service (charges prepaid) to the Company at the address set forth below and to any other recipient at the address indicated on the Company's records, or at such address or to the attention of such other Person as the recipient party has specified by prior written notice to the sending party. Notices and other such documents will be deemed to have been given or made hereunder when delivered personally or sent by facsimile (receipt confirmed) and one (1) Business Day after deposit with a reputable overnight courier service.

The Company's, Pi Topco's and PGHL's address is:

c/o Paysafe Limited 25 Canada Square, 27th Floor London, United Kingdom E14 5LQ Attention: Elliott Wiseman, General Counsel & Chief Compliance Officer Email:

If to any of the Principal Shareholders, to such Principal Shareholder's address at set forth on Schedule A hereto.

5.3 <u>Amendment; Waiver</u>. (a) The terms and provisions of this Agreement may be modified or amended only with the written approval of the Company and Principal Shareholders holding a majority of the aggregate outstanding Common Shares then held by the Principal Shareholders in the aggregate; *provided, however*, that any modification or amendment (i) to

<u>Section 2.1</u>, <u>Section 2.2</u> or this <u>Section 5.3</u> shall also require the approval of the CVC Designator, the Blackstone Designator and the FTAC Designators and (ii) that would adversely affect the rights of any of the CVC Investors, the Blackstone Investors or the FTAC Investors hereunder shall also require the approval of the CVC Investors, the Blackstone Investors or the FTAC Investors, as applicable.

- (b) Except as expressly set forth in this Agreement, neither the failure nor delay on the part of any party hereto to exercise any right, remedy, power or privilege under this Agreement shall operate as a waiver thereof, nor shall any single or partial exercise of any right, remedy power or privilege preclude any other or further exercise of the same or of any other right, remedy, power or privilege, nor shall any waiver of any right, remedy, power or privilege with respect to any occurrence be construed as a waiver of such right, remedy, power or privilege with respect to any other occurrence.
- (c) No party shall be deemed to have waived any claim arising out of this Agreement, or any right, remedy, power or privilege under this Agreement, unless the waiver of such claim, right, remedy, power or privilege is expressly set forth in a written instrument duly executed and delivered on behalf of such party; and any such waiver shall not be applicable or have any effect except in in the specific instance in which it is given.
- (d) Each Principal Shareholder, in such Principal Shareholder's sole discretion, may withdraw from this Agreement at any time by written notice to the Company. Thereafter, such Principal Shareholder shall cease to be a party to this Agreement, shall have no further rights or obligations hereunder and none of the terms or provisions hereof shall have any continuing force and effect with respect to such Principal Shareholder; provided, that until the expiration of the Lock-Up Period, the transfer restrictions set forth in Section 4.3 shall survive any such withdrawal and shall continue to apply to the Covered Shares of such withdrawing Principal Shareholder as if it were a party hereto.
 - (e) Any party hereto may unilaterally waive any of its rights hereunder in a signed writing delivered to the Company.
- 5.4 <u>Further Assurances</u>. The parties hereto will sign such further documents, cause such meetings to be held, resolutions passed, exercise their votes and do and perform and cause to be done such further acts and things necessary, proper or advisable in order to give full effect to this Agreement and every provision hereof. To the fullest extent permitted by law, the Company shall not directly or indirectly take any action that is intended to, or would reasonably be expected to result in, any Principal Shareholder being deprived of the rights contemplated by this Agreement.
- 5.5 <u>Assignment</u>. The rights and obligations hereunder shall not be assignable without the prior written consent of the other parties hereto; *provided, however*, that each Principal Shareholder may, without the consent of the Company or any other Person, assign its rights and obligations under Section 2.2 of this Agreement, in whole or in part, to any Transferee of Common Shares so long as (i) any right to designate Directors to the Board will not result in the Transferee receiving the right to designate more than two Directors where such designation rights would result in the Transferee receiving the right to designate a percentage of the Total

Number of Directors that is greater than the percentage of the aggregate outstanding Common Shares held by such Transferee after giving effect to such Transfer and (ii) such Transferee, if not already a party to this Agreement, executes and delivers to the Company a joinder to this Agreement evidencing its agreement to become a party to and to be bound by certain or all, as applicable, of the provisions of this Agreement as a "CVC Investor" or "Blackstone Investor," as applicable, hereunder, whereupon such Transferee shall be deemed a "CVC Investor" or "Blackstone Investor," as applicable, hereunder. This Agreement will inure to the benefit of and be binding on the parties hereto and their respective successors and permitted assigns.

- 5.6 <u>Third Parties</u>. Except as provided for in Article III with respect to any VCOC Investor that is an Affiliate of a Principal Shareholder, this Agreement does not create any rights, claims or benefits inuring to any person that is not a party hereto nor create or establish any third party beneficiary hereto.
- 5.7 <u>Governing Law</u>. THIS AGREEMENT AND ITS ENFORCEMENT AND ANY CONTROVERSY ARISING OUT OF OR RELATING TO THE MAKING OR PERFORMANCE OF THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK.
- 5.8 <u>Jurisdiction</u>; <u>Waiver of Jury Trial</u>. Each party hereto hereby (i) agrees that any action, directly or indirectly, arising out of, under or relating to this Agreement shall exclusively be brought in and shall exclusively be heard and determined by either the Supreme Court of the State of New York sitting in Manhattan or the United States District Court for the Southern District of New York, and (ii) solely in connection with the action(s) contemplated by subsection (i) hereof, (A) irrevocably and unconditionally consents and submits to the exclusive jurisdiction of the courts identified in subsection (i) hereof, (B) irrevocably and unconditionally waives any objection to the laying of venue in any of the courts identified in clause (i) of this <u>Section 5.8</u>, (C) irrevocably and unconditionally waives and agrees not to plead or claim that any of the courts identified in such clause (i) is an inconvenient forum or does not have personal jurisdiction over any party hereto, and (D) agrees that mailing of process or other papers in connection with any such action in the manner provided herein or in such other manner as may be permitted by applicable law shall be valid and sufficient service thereof. EACH PARTY HERETO HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY CLAIM OR ACTION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT OR THE SERVICES CONTEMPLATED HEREBY.
- 5.9 <u>Specific Performance</u>. Each party hereto acknowledges and agrees that in the event of any breach of this Agreement by any of them, the other parties hereto would be irreparably harmed and could not be made whole by monetary damages. Each party accordingly agrees to waive the defense in any action for specific performance that a remedy at law would be adequate and agrees that the parties, in addition to any other remedy to which they may be entitled at law or in equity, shall be entitled to seek specific performance of this Agreement without the posting of a bond.

- 5.10 Entire Agreement. This Agreement sets forth the entire understanding of the parties hereto with respect to the subject matter hereof. There are no agreements, representations, warranties, covenants or understandings with respect to the subject matter hereof or thereof. This Agreement supersedes all other prior agreements and understandings between the parties with respect to such subject matter.
- 5.11 <u>Severability</u>. If any provision of this Agreement, or the application of such provision to any Person or circumstance or in any jurisdiction, shall be held to be invalid or unenforceable to any extent, (i) the remainder of this Agreement shall not be affected thereby, and each other provision hereof shall be valid and enforceable to the fullest extent permitted by law, (ii) as to such Person or circumstance or in such jurisdiction such provision shall be reformed to be valid and enforceable to the fullest extent permitted by law, and (iii) the application of such provision to other Persons or circumstances or in other jurisdictions shall not be affected thereby.
- 5.12 <u>Table of Contents, Headings and Captions</u>. The table of contents, headings, subheadings and captions contained in this Agreement are included for convenience of reference only, and in no way define, limit or describe the scope of this Agreement or the intent of any provision hereof.
- 5.13 <u>Grant of Consent</u>. Any vote, consent or approval of, or designation by, or other action of, a Shareholder Designator hereunder shall be effective if notice of such vote, consent, approval, designation or action is provided in accordance with <u>Section 5.2</u> hereof by such Shareholder Designator as of the latest date any such notice is so provided to the Company.
- 5.14 <u>Counterparts</u>. This Agreement and any amendment hereto may be signed in any number of separate counterparts, each of which shall be deemed an original, but all of which taken together shall constitute one Agreement (or amendment, as applicable).
 - 5.15 Effectiveness. This Agreement shall become effective upon the Closing Date.
- 5.16 No Recourse. This Agreement may only be enforced against, and any claims or cause of action that may be based upon, arise out of or relate to this Agreement, or the negotiation, execution or performance of this Agreement, the transactions contemplated hereby or the subject matter hereof may only be made against the parties hereto and no past, present or future Affiliate, director, officer, employee, incorporator, member, shareholder, agent, attorney or representative of any party hereto or any past, present or future Affiliate, director, officer, employee, incorporator, member, manager, partner, shareholder, agent, attorney or representative of any of the foregoing (each, a "Non-Recourse Party.") shall have any liability for any obligations or liabilities of the parties to this Agreement or for any claim based on, in respect of, or by reason of, the transactions contemplated hereby. Without limiting the rights of any party against the other parties hereto, in no event shall any party or any of its Affiliates seek to enforce this Agreement against, make any claims for breach of this Agreement against, or seek to recover monetary damages from, any Non-Recourse Party.

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement on the day and year first above written.

•	COMPANY:
j	PAYSAFE LIMITED
1	Ву:
	Name:
	Title:
]	PI TOPCO:
]	PI JERSEY TOPCO LIMITED
1	By:
	Name:
	Title:
]	PGHL:
]	PAYSAFE GROUP HOLDINGS LIMITED
1	Ву:
	Name:
	Title:

РІ Н	OLDINGS JERSEY LIMITED
	Name: Title:
PI S	YNDICATION LP
By:	Pi Syndication GP Limited, its general partner
Ву:	Name: Title:

CVC INVESTORS:

BCP PI AGGREGATOR (CAYMAN) L.P.
By: BCP VII Holdings Manager (Cayman) L.L.C., its general partner
By: Blackstone Management Associates (Cayman) VII L.P., its managing member
By: BCP VII GP L.L.C., its general partner
By: Name: Title:
BCP VII CO-INVEST – STAR (CAYMAN) L.P.
By: Blackstone Management Associates (Cayman) VII L.P., its general partner
By: BCP VII GP L.L.C., its general partner
By: Name: Title:

BLACKSTONE INVESTORS:

(CAYMAN) VII – ESC L.P.
By: BCP VII GP L.L.C., its general partner
By: Name: Title:
BLACKSTONE PI CO-INVEST (CAYMAN) L.P.
By: Blackstone Management Associates (Cayman) VII L.P., its general partner
By: BCP VII GP L.L.C., its general partner
By: Name: Title:

BLACKSTONE FAMILY INVESTMENT PARTNERSHIP

CAN	NAE HOLDINGS, LLC
	Name: Title:
TRA	SIMENE CAPITAL FT, LP II
By:	Trasimene Capital FT, LLC II, its general partner
Ву:	Name: Title:

FTAC INVESTORS:

SCHEDULE A

Principal Shareholder Name

CVC Investors

Address for service of notices

For the attention of: Peter Rutland

Address: 27 Esplanade St Helier, Jersey JE1 1SG

E-mail address: with a copy to: Latham & Watkins

For the attention of: David Walker and Kem Ihenacho

Address: 99 Bishopsgate, London EC2M 3XF

E-mail address: david.walker@lw.com; kem.ihenacho@lw.com

and

Simpson Thacher & Bartlett LLP

For the attention of: Joshua Ford Bonnie, William R. Golden III, Elizabeth Cooper and Katherine Krause

Address: 900 G Street NW, Washington, DC 20001; 425 Lexington Avenue, New York, NY 10017

E-mail address: jbonnie@stblaw.com; wgolden@stblaw.com; ecooper@stblaw.com;

katherine.krause@stblaw.com

Blackstone Investors For the attention of: Martin Brand

Address: [c/o Intertrust Corporate Services (Cayman) Limited, 190 Elgin Avenue, George Town, Grand

Cayman, KY1-9005, Cayman Island]

E-mail address:

with a copy to:

Latham & Watkins

For the attention of: David Walker and Kem Ihenacho

Address: 99 Bishopsgate, London EC2M 3XF

E-mail address: david.walker@lw.com; kem.ihenacho@lw.com

and

Simpson Thacher & Bartlett LLP

For the attention of: Joshua Ford Bonnie, William R. Golden III, Elizabeth Cooper and Katherine Krause

Address: 900 G Street NW, Washington, DC 20001; 425 Lexington Avenue, New York, NY 10017

 $E\text{-}mail\ address:\ jbonnie@stblaw.com;\ wgolden@stblaw.com;\ ecooper@stblaw.com;$

katherine.krause@stblaw.com

FTAC Investors

For the attention of: Michael L. Gravelle

Address: 1701 Village Center Circle, Las Vegas, NV 89134

E-mail address: with a copy to:

Weil, Gotshal & Manges LLP

For the attention of: Michael J. Aiello; Eoghan P. Keenan

Address: 767 Fifth Avenue, New York, NY 10153

 $E\text{-}mail\ address:\ michael.aiello@weil.com;\ Eoghan.keenan@weil.com$

AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT

This Amended and Restated Registration Rights Agreement (this "Agreement") is made and entered into as of [•], 202[•] among PAYSAFE LIMITED, an exempted limited company incorporated under the laws of Bermuda (the "Company"), PI JERSEY TOPCO LIMITED, a company incorporated in Jersey ("Pi Topco"), PAYSAFE GROUP HOLDINGS LIMITED, a private limited company incorporated under the laws of England and Wales ("PGHL") and the persons identified on <u>Schedule A</u> hereto (together with their respective affiliates, successors and permitted assigns, collectively, the "Investors" and, each individually, an "Investor").

WHEREAS, Pi Topco and certain of the Investors previously entered into the Registration Rights Agreement, dated as of December 27, 2017 (the "Original Agreement"); and

WHEREAS, the Original Agreement may be amended by the holders of a majority of the Registrable Securities (as defined below) and parties to the Original Agreement representing a majority of the Registrable Securities now desire to enter into this Agreement to amend and restate the Original Agreement in its entirety as more fully set forth below.

NOW, THEREFORE, in consideration of the foregoing and the mutual and dependent covenants hereinafter set forth, the parties hereto agree as follows:

- 1. Defined Terms. As used in this Agreement, the following terms shall have the following meanings:
 - "Agreement" has the meaning set forth in the preamble.
 - "Board" means the board of directors (or any successor governing body) of the Company.
 - "Closing Date" means the date of this Agreement.
 - "Commission" means the Securities and Exchange Commission or any other federal agency administering the Securities Act and the Exchange Act at the time.
 - "Company" has the meaning set forth in the preamble and includes the Company's successors by merger, amalgamation, acquisition, reorganization or otherwise.
 - "Controlling Person" has the meaning set forth in Section 5(g).
 - "Demand Registration" has the meaning set forth in Section 2(c).
 - "DTCDRS" has the meaning set forth in Section 5(r).
 - "Effectiveness Deadline" has the meaning set forth in Section 2(b).
 - "Equity Securities" means all of the issued equity securities of the Company from time to time (including any warrants or any shares issuable upon exercise of such warrants and any other shares issued or issuable with respect thereto (whether by way of a share dividend or share split or in exchange for or upon conversion of such shares or otherwise in connection with a combination of shares, distribution, recapitalization, merger, consolidation, other corporate reorganization or other similar event with respect to the Equity Securities)).
 - "Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder, as the same may be amended from time to time.

- "Francisco Partners Investors" means Francisco Partners IV, L.P and Francisco Partners IV-A, L.P. and their respective affiliates, successors and permitted assigns.
- "Governmental Authority" means any federal, state, provincial, municipal, local or foreign government, governmental authority, regulatory or administrative agency, governmental commission, department, board, bureau, agency or instrumentality, court or tribunal.
- "Initial Registrable Securities" has the meaning set forth in Section 5(a)(ii).
- "Initial Registrable Statement" has the meaning set forth in Section 5(a)(ii).
- "Inspectors" has the meaning set forth in Section 5(h).
- "Investors" has the meaning set forth in the preamble.
- "Long-Form Registration" has the meaning set forth in Section 2(a).
- "New Registration Statement" has the meaning set forth in Section 5(a)(ii).
- "**Person**" means any individual, firm, corporation, partnership, limited liability company, incorporated or unincorporated association, joint venture, joint stock company, governmental agency or instrumentality or other entity of any kind.
- "PGHL" has the meaning set forth in the preamble and includes PGHL's successors by merger, amalgamation, acquisition, reorganization or otherwise.
- "Pi Topco" has the meaning set forth in the preamble and includes Pi Topco's successors by merger, amalgamation, acquisition, reorganization or otherwise
- "Piggyback Registration" has the meaning set forth in Section 3(a).
- "Piggyback Registration Statement" has the meaning set forth in Section 3(a).
- "Piggyback Shelf Registration Statement" has the meaning set forth in Section 3(a).
- "Piggyback Shelf Takedown" has the meaning set forth in Section 3(a).
- "Prospectus" means the prospectus or prospectuses included in any Registration Statement (including, without limitation, a prospectus that includes any information previously omitted from a prospectus filed as part of an effective Registration Statement in reliance on Rule 430A under the Securities Act or any successor rule thereto), as amended or supplemented by any prospectus supplement, including any Shelf Supplement, with respect to the terms of the offering of any portion of the Registrable Securities covered by such Registration Statement and by all other amendments and supplements to the prospectus, including post-effective amendments and all material incorporated by reference in such prospectus or prospectuses.
- "Records" has the meaning set forth in Section 5(h).
- "Registrable Securities" means (a) any Equity Securities beneficially owned or otherwise held directly or indirectly by any of the Investors, (b) any Equity Securities that are directly held or indirectly, as set forth in the books and records of PGHL or Pi Topco, as applicable, are attributed to an Investor, (c) any Equity Securities in which other shareholders of PGHL or Pi Topco, as applicable, directly hold or indirectly, as set

forth in the books and records of PGHL or Pi Topco, as applicable, are attributed an interest, over which an Investor has the right to direct a sale as set forth in the books and records of PGHL or Pi Topco, as applicable, and (d) any Equity Securities issued or issuable with respect to any shares described in subsections (a) through (c) above by way of a share dividend or share split or in exchange for or upon conversion of such shares or otherwise in connection with a combination of shares, distribution, recapitalization, merger, consolidation, other reorganization or other similar event with respect to the Equity Securities (it being understood that, for purposes of this Agreement, a Person shall be deemed to be a holder of Registrable Securities whenever such Person has the right to then acquire or obtain from the Company any Registrable Securities, whether or not such acquisition has actually been effected). As to any particular Registrable Securities, such securities shall cease to be Registrable Securities when (i) the Commission has declared a Registration Statement covering such securities effective and such securities have been disposed of pursuant to such effective Registration Statement, or (ii) such securities are sold under circumstances in which all of the applicable conditions of Rule 144 under the Securities Act are met.

- "Registration Date" means the date on which the Company becomes subject to Section 13(a) or Section 15(d) of the Exchange Act.
- "Registration Statement" means any registration statement of the Company, including the Prospectus, amendments and supplements (including Shelf Supplements) to such registration statement, including post-effective amendments, all exhibits and all material incorporated by reference in such registration statement.
- "Rule 144" means Rule 144 under the Securities Act or any successor rule thereto.
- "Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder, as the same may be amended from time to time.
- "Selling Expenses" means all underwriting discounts, selling commissions and share transfer taxes applicable to the sale of Registrable Securities, and fees and disbursements of counsel for any holder of Registrable Securities, except for the fees and disbursements of counsel for the holders of Registrable Securities required to be paid by the Company pursuant to Section 6.
- "Shelf Registration" has the meaning set forth in Section 2(c).
- "Shelf Registration Statement" has the meaning set forth in Section 2(c).
- "Shelf Supplement" has the meaning set forth in Section 2(d).
- "Shelf Takedown" has the meaning set forth in Section 2(d).
- "Shelf Takedown Notice" has the meaning set forth in Section 2(d).
- "Short-Form Registration" has the meaning set forth in Section 2(c).
- "**Target Filing Date**" has the meaning set forth in Section 2(c).

2. Registration.

(a) To the extent that a Registration Statement filed pursuant to <u>Section 2(b)</u> or a Shelf Registration Statement is not available to effect the proposed transaction, each Investor may request that the Company register under the Securities Act all or any portion of its Registrable Securities pursuant to a Registration Statement on Form F-1, S-1 or any successor form thereto with respect to a underwritten public offering of Registrable Securities (each, a "**Long-Form Registration**"). Each request for a Long-Form Registration shall

specify the number of Registrable Securities requested to be included in the Long-Form Registration. Upon receipt of any such request, the Company shall promptly (but in no event later than 10 days following receipt thereof) deliver notice of such request to all other holders of Registrable Securities who shall then have 10 days from the date such notice is given to notify the Company in writing of their desire to be included in such registration. The Company shall prepare and file with (or confidentially submit to) the Commission a Registration Statement on Form F-1, S-1 or any successor form thereto covering all of the Registrable Securities that the holders thereof have requested to be included in such Long-Form Registration within 60 days after the date on which the initial request is given and shall use its best efforts to cause such Registration Statement to be declared effective by the Commission as soon as practicable thereafter.

(b) The Company shall, as soon as practicable, but in any event within forty-five (45) days after the Closing Date, file (or confidentially submit) a Registration Statement to permit the public resale of all the Registrable Securities held by the Investors from time to time as permitted by Rule 415 under the Securities Act (or any successor or similar provision adopted by the Commission then in effect) on the terms and conditions specified in this <u>Section 2(b)</u> and shall use its commercially reasonable efforts to cause the Registration Statement to be declared effective as soon as practicable after the filing thereof, but in no event later than the earlier of (i) the 105th day (or 165th day if the Commission notifies the Company that it will "review" the Registration Statement) following the Closing Date and (ii) the 10th business day after the date the Company is notified (orally or in writing, whichever is earlier) by the Commission that the Registration Statement will not be "reviewed" or will not be subject to further review (such earlier date, the "Effectiveness Deadline"). The Registration Statement filed with the Commission pursuant to this Section 2(b) shall be on Form F-3 or S-3, or if Form F-3 or S-3 is not then available to the Company, on Form F-1 or S-1 or such other form of registration statement as is then available to effect a registration for the sale or resale of such Registrable Securities on a delayed or continuous basis pursuant to Rule 415 under the Securities Act or any successor rule or provision similar thereto adopted by the Commission, covering such Registrable Securities, and shall contain a Prospectus in such form as to permit any Investor to sell such Registrable Securities pursuant to Rule 415 under the Securities Act (or any successor rule or similar provision adopted by the Commission then in effect) at any time beginning on the effective date for such Registration Statement. A Registration Statement filed pursuant to this <u>Section 2(b)</u> shall provide for the sale or resale pursuant to any method or combination of methods legally available to, and requested by, the Investors. The Company shall use its reasonable best efforts to cause a Registration Statement filed pursuant to Section 2(b) to remain effective, and to be supplemented and amended to the extent necessary to ensure that such Registration Statement is available or, if not available, that another Registration Statement or Shelf Registration Statement is continuously available, for the resale of all the Registrable Securities held by the Holders until all such Registrable Securities have ceased to be Registrable Securities. As soon as practicable following the effective date of a Registration Statement filed pursuant to this Section 2(b), but in any event within one (1) business day of such date, the Company shall notify the Investors of the effectiveness of such Registration Statement. If, after the filing such Registration Statement, a holder of Registrable Securities requests registration under the Securities Act of additional Registrable Securities pursuant to such Registration Statement, the Company shall amend such Registration Statement to cover such additional Registrable Securities.

(c) The Company shall use its best efforts to qualify and remain qualified to register the offer and sale of securities under the Securities Act pursuant to a Registration Statement on Form F-3, S-3 or any successor form thereto. As soon as practicable after the date hereof, but not later than the Target Filing Date, the Company shall (i) prepare and file with (or confidentially submit to) the Commission a Registration Statement on Form F-3, S-3 or the then appropriate form for an offering to be made on a delayed or continuous basis pursuant to Rule 415 under the Securities Act or any successor rule thereto (a "Shelf Registration Statement") that covers all Registrable Securities then outstanding for an offering to be made on a delayed or continuous basis pursuant to Rule 415 under the Securities Act or any successor rule thereto (a "Shelf Registration") and (ii) use its best efforts to cause such Shelf Registration Statement to be declared effective by the Commission as soon as practicable thereafter. In addition, the Company shall use its best efforts to cause a Shelf Registration Statement filed pursuant to Section 2(c) to remain effective, and to be supplemented and amended to the extent necessary

to ensure that such Shelf Registration Statement is available or, if not available, that another Shelf Registration Statement (if the Company is eligible to file a Shelf Registration Statement) or other Registration Statement (if the Company is not so eligible) is continuously available, for the resale of all the Registrable Securities held by the Holders until all such Registrable Securities have ceased to be Registrable Securities. For purposes hereof, "Target Filing Date" shall mean the date which is 30 days after the Company becomes qualified to register the offer and sale of securities under the Securities Act pursuant to a Shelf Registration Statement. If, after the filing of a Shelf Registration Statement, a holder of Registrable Securities requests registration under the Securities Act of additional Registrable Securities pursuant to such Shelf Registration, the Company shall amend such Shelf Registration Statement to cover such additional Registrable Securities. At such time as the Company shall have qualified for the use of a Registration Statement on Form F-3, S-3 or any successor form thereto, the holders of Registrable Securities shall have the right to request an unlimited number of registrations under the Securities Act of all or any portion of their Registrable Securities pursuant to a Registration Statement on Form F-3, S-3 or any similar short-form Registration Statement (each, a "Short-Form Registration" and, collectively with each Long-Form Registration and Shelf Registration (as defined below), a "Demand Registration"). Each request for a Short-Form Registration shall specify the number of Registrable Securities requested to be included in the Short-Form Registration. Upon receipt of any such request, the Company shall promptly (but in no event later than 10 days following receipt thereof) deliver notice of such request to all other holders of Registrable Securities who shall then have 10 days from the date such notice is given to notify the Company in writing of their desire to be included in such registration. The Company shall prepare and file with (or confidentially submit to) the Commission a Registration Statement on Form F-3, S-3 or any successor form thereto covering all of the Registrable Securities that the holders thereof have requested to be included in such Short-Form Registration within 30 days after the date on which the initial request is given and shall use its best efforts to cause such Registration Statement to be declared effective by the Commission as soon as practicable thereafter.

(d) At any time that a Shelf Registration Statement is effective, if a holder of Registrable Securities covered by such Shelf Registration Statement delivers a notice to the Company (a "Shelf Takedown Notice") stating that the holder intends to effect an offering of all or part of its Registrable Securities included in such Shelf Registration Statement (a "Shelf Takedown") and the Company is eligible to use such Shelf Registration Statement for such Shelf Takedown, then the Company shall take all actions reasonably required, including amending or supplementing (a "Shelf Supplement") such Shelf Registration Statement, to enable such Registrable Securities to be offered and sold as contemplated by such Shelf Takedown Notice. Each Shelf Takedown Notice shall specify the number of Registrable Securities to be offered and sold under the Shelf Takedown. Upon receipt of a Shelf Takedown Notice, the Company shall promptly (but in no event later than five (5) business days, or, in the case of an underwritten overnight "block trade", two (2) business days, following receipt thereof) deliver notice of such Shelf Takedown Notice to all other holders of Registrable Securities who shall then have five (5) business days, or, in the case an underwritten overnight "block trade," one (1) business day, from the date such notice is given to notify the Company in writing of their desire to be included in such Shelf Takedown. The Company shall prepare and file with the Commission a Shelf Supplement as soon as practicable after the date on which it received the Shelf Takedown Notice and, if such Shelf Supplement is an amendment to such Shelf Registration Statement, shall use its best efforts to cause such Shelf Supplement to be declared effective by the Commission as soon as practicable thereafter.

(e) The Company shall not be obligated to effect any Long-Form Registration (x) within 90 days after the effective date of a previous Long-Form Registration or Shelf Takedown or a previous Piggyback Registration in which holders of Registrable Securities were permitted to register the offer and sale under the Securities Act, and actually sold, all of the shares of Registrable Securities requested to be included therein (y) or while a lock-up agreement pursuant to Section 4 or any other lock-up agreement relating to such holder's Registrable Securities is in effect. Notwithstanding anything otherwise to the contrary herein, the Company shall not be required to provide notice of any requested underwritten public offering to any holders of Registrable Securities whose shares are subject to any applicable lock-up arrangements at the time of such request, and any such holders shall not have the right to receive information on or participate in any such underwritten public offering.

The Company may postpone for up to 90 days the filing or effectiveness of a Registration Statement for a Demand Registration or the filing of a Shelf Supplement for a Shelf Takedown if the Board determines in its reasonable good faith judgment that such Demand Registration or Shelf Takedown would (i) materially interfere with a significant acquisition, corporate reorganization, financing, securities offering or other similar transaction involving the Company; (ii) require premature disclosure of material information that the Company has a bona fide business purpose for preserving as confidential; or (iii) render the Company unable to comply with requirements under the Securities Act or Exchange Act. The Company may delay a Demand Registration or Shelf Takedown pursuant to the immediately preceding sentence only once in any period of 12 consecutive months.

(f) If the holders of the Registrable Securities initially requesting a Demand Registration or Shelf Takedown elect to distribute the Registrable Securities covered by their request in an underwritten offering, they shall so advise the Company as a part of their request made pursuant to Section 2(a), Section 2(b), Section 2(c) or Section 2(d), and the Company shall include such information in its notice to the other holders of Registrable Securities. The holders of a majority of the Registrable Securities initially requesting the Demand Registration or Shelf Takedown shall select the investment banking firm or firms to act as the managing underwriter or underwriters in connection with such offering.

(g) The Company shall not include in any Demand Registration or Shelf Takedown any securities which are not Registrable Securities without the prior written consent of the holders of a majority of the Registrable Securities included in such Demand Registration or Shelf Takedown, which consent shall not be unreasonably withheld or delayed. If a Demand Registration or Shelf Takedown involves an underwritten offering and the managing underwriter of the requested Demand Registration or Shelf Takedown advises the Company and the holders of Registrable Securities in writing that in its reasonable and good faith opinion the number of shares of Equity Securities proposed to be included in the Demand Registration or Shelf Takedown, including all Registrable Securities and all other shares of Equity Securities proposed to be included in such underwritten offering, exceeds the number of shares of Equity Securities which can be sold in such underwritten offering and/or the number of shares of Equity Securities proposed to be included in such Demand Registration or Shelf Takedown would adversely affect the price per share of the Equity Securities proposed to be sold in such underwritten offering, the Company shall include in such Demand Registration or Shelf Takedown (i) first, the shares of Equity Securities that the holders of Registrable Securities propose to sell, and (ii) second, the shares of Equity Securities proposed to be included therein by any other Persons (including shares of Equity Securities to be sold for the account of the Company and/or other holders of Equity Securities) allocated among such Persons in such manner as they may agree. If the managing underwriter determines that less than all of the Registrable Securities proposed to be sold can be included in such offering, then the Registrable Securities owned by each such holder.

3. Piggyback Registration.

(a) Whenever the Company proposes to offer or sell any shares of its Equity Securities pursuant to a registered offering under the Securities Act (other than a registration (i) pursuant to a Registration Statement on Form S-8 (or other registration solely relating to an offering or sale to employees or directors of the Company pursuant to any employee share plan or other employee benefit arrangement), (ii) pursuant to a Registration Statement on Form F-4, S-4 (or similar form that relates to a transaction subject to Rule 145 under the Securities Act or any successor rule thereto), or (iii) in connection with any dividend or distribution reinvestment or similar plan), whether for its own account or for the account of one or more shareholders of the Company and the form of Registration Statement (a "Piggyback Registration Statement") to be used may be used for any registration of Registrable Securities (a "Piggyback Registration"), the Company shall give prompt written notice (in any event no later than ten (10) business days prior to either the filing of such Registration Statement or, with respect to a Piggyback Shelf Takedown, the filing of a prospectus supplement to the applicable Piggyback Shelf Registration Statement) to the holders of Registrable Securities of its intention to effect such

a registration and, subject to Section 3(b) and Section 3(c), shall include in such registration all Registrable Securities with respect to which the Company has received written requests for inclusion from the holders of Registrable Securities within five (5) business days after the Company's notice has been given to each such holder. A Piggyback Registration shall not be considered a Demand Registration for purposes of Section 2. If any Piggyback Registration Statement pursuant to which holders of Registrable Securities have registered the offer and sale of Registrable Securities is a Registration Statement on Form F-3, S-3 or the then appropriate form for an offering to be made on a delayed or continuous basis pursuant to Rule 415 under the Securities Act or any successor rule thereto (a "Piggyback Shelf Registration Statement"), such holder(s) shall have the right, but not the obligation, to be notified of and to participate in any offering under such Piggyback Shelf Registration Statement (a "Piggyback Shelf Takedown").

- (b) If a Piggyback Registration or Piggyback Shelf Takedown is initiated as a primary underwritten offering on behalf of the Company and the managing underwriter advises the Company and the holders of Registrable Securities (if any holders of Registrable Securities have elected to include Registrable Securities in such Piggyback Registration or Piggyback Shelf Takedown) in writing that in its reasonable and good faith opinion the number of shares of Equity Securities proposed to be included in such registration or takedown, including all Registrable Securities and all other shares of Equity Securities proposed to be included in such underwritten offering, exceeds the number of shares of Equity Securities which can be sold in such offering and/or that the number of shares of Equity Securities proposed to be included in any such registration or takedown would adversely affect the price per share of the Equity Securities to be sold in such offering, the Company shall include in such registration or takedown (i) first, the shares of Equity Securities that the Company proposes to sell; (ii) second, the shares of Equity Securities requested to be included therein by holders of Registrable Securities, allocated pro rata among all such holders on the basis of the number of Registrable Securities owned by each such holder or in such manner as they may otherwise agree; and (iii) third, the shares of Equity Securities requested to be included therein by holders of Equity Securities other than holders of Registrable Securities, allocated among such holders in such manner as they may agree.
- (c) If a Piggyback Registration or Piggyback Shelf Takedown is initiated as an underwritten offering on behalf of a holder of Equity Securities other than Registrable Securities, and the managing underwriter advises the Company in writing that in its reasonable and good faith opinion the number of shares of Equity Securities proposed to be included in such registration or takedown, including all Registrable Securities and all other shares of Equity Securities proposed to be included in such underwritten offering, exceeds the number of shares of Equity Securities which can be sold in such offering and/or that the number of shares of Equity Securities proposed to be included in any such registration or takedown would adversely affect the price per share of the Equity Securities to be sold in such offering, the Company shall include in such registration or takedown (i) first, the shares of Equity Securities requested to be included therein by the holder(s) requesting such registration or takedown and by the holders of Registrable Securities, allocated pro rata among all such holders on the basis of the number of shares of Equity Securities other than the Registrable Securities (on a fully diluted, as converted basis) and the number of Registrable Securities, as applicable, owned by all such holders or in such manner as they may otherwise agree; and (ii) second, the shares of Equity Securities requested to be included therein by other holders of Equity Securities, allocated among such holders in such manner as they may agree.
- (d) If any Piggyback Registration or Piggyback Shelf Takedown is initiated as a primary underwritten offering on behalf of the Company, the Company shall, subject to the prior written consent of the holders of a majority of the Registrable Securities included in such Piggyback Registration or Piggyback Shelf Takedown, which consent shall not be unreasonably withheld or delayed, select the investment banking firm or firms to act as the managing underwriter or underwriters in connection with such offering.
- 4. <u>Lock-up Agreement</u>. In connection with any registered offering of the Equity Securities or other equity securities of the Company, and upon the request of the managing underwriter in such offering, each holder of Registrable Securities agrees to execute a customary lock-up agreement. The Company shall cause its executive officers and its directors, which directors are selling Equity Securities in such offering (as applicable) and shall use reasonable best efforts to cause other holders of Equity Securities who beneficially own (within the meaning of Rules 13d-3 and 13d-5 under the Exchange Act as in effect on the date of this Agreement) 10% or more of the then outstanding Equity Securities and holders of any of the Registrable Securities participating in such offering, to enter into lock-up agreements that contain restrictions that are no less restrictive than the restrictions contained in the lock-up agreements executed by the holders of Registrable Securities. Each holder of Registrable Securities agrees to execute

and deliver such other agreements as may be reasonably requested by the Company or the managing underwriter which are consistent with the foregoing or which are necessary to give further effect thereto. Notwithstanding anything to the contrary contained in this Section 4, each holder of Registrable Securities shall be released, pro rata, from any lock-up agreement entered into pursuant to this Section 4 in the event and to the extent that the managing underwriter or the Company permit any discretionary waiver or termination of the restrictions of any lock-up agreement pertaining to any officer, director or holder of greater than 10% of the outstanding Equity Securities.

- 5. <u>Registration Procedures</u>. If and whenever the holders of Registrable Securities request that the offer and sale of any Registrable Securities be registered under the Securities Act or any Registrable Securities be distributed in a Shelf Takedown pursuant to the provisions of this Agreement, the Company shall use its best efforts to effect the registration of the offer and sale of such Registrable Securities under the Securities Act in accordance with the intended method of disposition thereof, and pursuant thereto the Company shall as soon as practicable and as applicable:
 - (a) subject to Section 2(a), Section 2(b), Section 2(c) and Section 2(d), (i) prepare and file with the Commission a Registration Statement covering such Registrable Securities and use its best efforts to cause such Registration Statement to be declared effective; and (ii) if (A) the Company has filed a Registration Statement (the "Initial Registration Statement") with the Commission that covers Registrable Securities (the "Initial Registrable Securities"), (B) pursuant to Rule 415(a)(5) under the Securities Act or any successor rule thereto, the Initial Registration Statement may no longer be used for offers and sales of any of the Initial Registrable Securities, and (C) any of the Initial Registrable Securities are Registrable Securities at the time that (B) above occurs, the Company shall prepare and file with the Commission within the time limits required by Rule 415 under the Securities Act or any successor rule thereto a new Registration Statement covering any Initial Registrable Securities that have not ceased to be Registrable Securities for an offering to be made on a delayed on continuous basis pursuant to Rule 415 under the Securities Act or any successor rule thereto (a "New Registration Statement") and shall use its best efforts to cause such New Registration Statement to be declared effective by the Commission as soon as practicable thereafter;
 - (b) (i) in the case of a Long-Form Registration or a Short-Form Registration, prepare and file with the Commission such amendments, post-effective amendments and supplements to such Registration Statement and the Prospectus used in connection therewith as may be necessary to keep such Registration Statement effective for a period of not less than 180 days, or if earlier, until all of such Registrable Securities have been disposed of and to comply with the provisions of the Securities Act with respect to the disposition of such Registrable Securities in accordance with the intended methods of disposition set forth in such Registration Statement; and (ii) in the case of a Shelf-Registration, prepare and file with the Commission such amendments, post-effective amendments and supplements, including Shelf Supplements, to such Registration Statement and the Prospectus used in connection therewith as may be necessary to keep such Registration Statement effective and to comply with the provisions of the Securities Act with respect to the disposition of all Registrable Securities subject thereto for a period ending on the earlier of (i) 36 months after the effective date of such Registration Statement and (ii) the date on which all the Registrable Securities subject thereto have been sold pursuant to such Registration Statement;
 - (c) within a reasonable time before filing such Registration Statement, Prospectus or amendments or supplements thereto with the Commission, furnish to one counsel selected by holders of a majority of such Registrable Securities copies of such documents proposed to be filed, which documents shall be subject to the review, comment and approval of such counsel;
 - (d) notify each selling holder of Registrable Securities, promptly after the Company receives notice thereof, of the time when such Registration Statement has been declared effective or a supplement, including a Shelf Supplement, to any Prospectus forming a part of such Registration Statement has been filed with the Commission;

- (e) furnish to each selling holder of Registrable Securities such number of copies of the Prospectus included in such Registration Statement (including each preliminary Prospectus) and any supplement thereto, including a Shelf Supplement (in each case including all exhibits and documents incorporated by reference therein), and such other documents as such seller may request in order to facilitate the disposition of the Registrable Securities owned by such seller;
- (f) use its best efforts to register or qualify such Registrable Securities under such other securities or "blue sky" laws of such jurisdictions as any selling holder requests and do any and all other acts and things which may be necessary or advisable to enable such holders to consummate the disposition in such jurisdictions of the Registrable Securities owned by such holders; <u>provided</u>, that the Company shall not be required to qualify generally to do business, subject itself to general taxation or consent to general service of process in any jurisdiction where it would not otherwise be required to do so but for this <u>Section 5(f)</u>;
- (g) notify each selling holder of such Registrable Securities, at any time when a Prospectus relating thereto is required to be delivered under the Securities Act, of the happening of any event that would cause the Prospectus included in such Registration Statement to contain an untrue statement of a material fact or omit any fact necessary in order to make the statements made therein, in light of the circumstances under which they were made, not misleading, and, at the request of any such holder, the Company shall prepare a supplement or amendment to such Prospectus so that, as thereafter delivered to the purchasers of such Registrable Securities, such Prospectus shall not contain an untrue statement of a material fact or omit to state any fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading;
- (h) make available for inspection by any selling holder of Registrable Securities, any underwriter participating in any disposition pursuant to such Registration Statement and any attorney, accountant or other agent retained by any such holder or underwriter (collectively, the "Inspectors"), all financial and other records, pertinent corporate documents and properties of the Company (collectively, the "Records"), and cause the Company's officers, directors and employees to supply all information requested by any such Inspector in connection with such Registration Statement;
- (i) provide a transfer agent and registrar (which may be the same entity) for all such Registrable Securities not later than the effective date of such registration;
- (j) use its best efforts to cause such Registrable Securities to be listed on each securities exchange on which the Equity Securities is then listed or, if the Equity Securities is not then listed, on a national securities exchange selected by the holders of a majority of such Registrable Securities:
- (k) in connection with an underwritten offering, enter into such customary agreements (including underwriting and lock-up agreements in customary form) and take all such other customary actions as the holders of such Registrable Securities or the managing underwriter of such offering request in order to expedite or facilitate the disposition of such Registrable Securities (including, without limitation, making appropriate officers of the Company available to participate in "road show" and other customary marketing activities (including one-on-one meetings with prospective purchasers of the Registrable Securities));
- (l) otherwise use its best efforts to comply with all applicable rules and regulations of the Commission and make available to its shareholders an earnings statement (in a form that satisfies the provisions of Section 11(a) of the Securities Act and Rule 158 under the Securities Act or any successor rule thereto) no later than 30 days after the end of the 12-month period beginning with the first day of the Company's first full fiscal quarter after the effective date of such Registration Statement, which earnings statement shall cover said 12-month period, and which requirement will be deemed to be satisfied if the Company timely files complete and accurate information on Forms 20-F, 6-K, 10-K, 10-Q and 8-K, as applicable, under the Exchange Act and otherwise complies with Rule 158 under the Securities Act or any successor rule thereto; and

- (m) furnish to each selling holder of Registrable Securities and each underwriter, if any, with (i) a written legal opinion of the Company's outside counsel, dated the closing date of the offering, in form and substance as is customarily given in opinions of the Company's counsel to underwriters in underwritten registered offerings; and (ii) on the date of the applicable Prospectus, on the effective date of any post-effective amendment to the applicable Registration Statement and at the closing of the offering, dated the respective dates of delivery thereof, a "comfort" letter signed by the Company's independent certified public accountants in form and substance as is customarily given in accountants' letters to underwriters in underwritten registered offerings;
- (n) without limiting <u>Section 5(f)</u>, use its best efforts to cause such Registrable Securities to be registered with or approved by such other governmental agencies or authorities as may be necessary by virtue of the business and operations of the Company to enable the holders of such Registrable Securities to consummate the disposition of such Registrable Securities in accordance with their intended method of distribution thereof:
- (o) notify the holders of Registrable Securities promptly of any request by the Commission for the amending or supplementing of such Registration Statement or Prospectus or for additional information;
- (p) advise the holders of Registrable Securities, promptly after it shall receive notice or obtain knowledge thereof, of the issuance of any stop order by the Commission suspending the effectiveness of such Registration Statement or the initiation or threatening of any proceeding for such purpose and promptly use its best efforts to prevent the issuance of any stop order or to obtain its withdrawal at the earliest possible moment if such stop order should be issued;
- (q) permit any holder of Registrable Securities which holder, in its sole and exclusive judgment, might be deemed to be an underwriter or a "controlling person" (within the meaning of Section 15 of the Securities Act and Section 20 of the Exchange Act) (a "Controlling Person") of the Company, to participate in the preparation of such Registration Statement and to require the insertion therein of language, furnished to the Company in writing, which in the reasonable judgment of such holder and its counsel should be included;
- (r) cooperate with the holders of the Registrable Securities to facilitate the timely preparation and delivery of certificates representing the Registrable Securities to be sold pursuant to such Registration Statement or Rule 144 free of any restrictive legends and representing such number of shares of Equity Securities and registered in such names as the holders of the Registrable Securities may reasonably request a reasonable period of time prior to sales of Registrable Securities pursuant to such Registration Statement or Rule 144; provided, that the Company may satisfy its obligations hereunder without issuing physical share certificates through the use of The Depository Trust Company's Direct Registration System (the "DTCDRS");
- (s) not later than the effective date of such Registration Statement, provide a CUSIP number for all Registrable Securities and provide the applicable transfer agent with printed certificates for the Registrable Securities which are in a form eligible for deposit with The Depository Trust Company; <u>provided</u>, that the Company may satisfy its obligations hereunder without issuing physical share certificates through the use of the DTCDRS;
- (t) take no direct or indirect action prohibited by Regulation M under the Exchange Act; <u>provided</u>, that, to the extent that any prohibition is applicable to the Company, the Company will take all reasonable action to make any such prohibition inapplicable; and
- (u) otherwise use its best efforts to take all other steps necessary to effect the registration of such Registrable Securities contemplated hereby.

6. Expenses. All expenses (other than Selling Expenses) incurred by the Company in complying with its obligations pursuant to this Agreement and in connection with the registration and disposition of Registrable Securities shall be paid by the Company, including, without limitation, all (i) registration and filing fees (including, without limitation, any fees relating to filings required to be made with, or the listing of any Registrable Securities on, any securities exchange or over-the-counter trading market on which the Registrable Securities are listed or quoted); (ii) underwriting expenses (other than fees, commissions or discounts); (iii) expenses of any audits incident to or required by any such registration; (iv) fees and expenses of complying with securities and "blue sky" laws (including, without limitation, fees and disbursements of counsel for the Company in connection with "blue sky" qualifications or exemptions of the Registrable Securities); (v) printing expenses; (vi) messenger, telephone and delivery expenses; (vii) fees and expenses of one counsel for the holders of Registrable Securities participating in such registration as a group (selected by the holders of a majority of the Registrable Securities included in the registration). In addition, the Company shall be responsible for all of its internal expenses incurred in connection with the consummation of the transactions contemplated by this Agreement (including, without limitation, all salaries and expenses of its officers and employees performing legal or accounting duties) and the expense of any annual audits. All Selling Expenses relating to the offer and sale of Registrable Securities registered under the Securities Act pursuant to this Agreement shall be borne and paid by the holders of such Registrable Securities, in proportion to the number of Registrable Securities included in such registration for each such holder.

7. Indemnification.

(a) The Company shall indemnify and hold harmless, to the fullest extent permitted by law, each holder of Registrable Securities, such holder's officers, directors, managers, members, partners, shareholders and Affiliates, each underwriter, broker or any other Person acting on behalf of such holder of Registrable Securities and each other Controlling Person, if any, who controls any of the foregoing Persons, against all losses, claims, actions, damages, liabilities and expenses, joint or several, to which any of the foregoing Persons may become subject under the Securities Act or otherwise, insofar as such losses, claims, actions, damages, liabilities or expenses arise out of or are based upon any untrue or alleged untrue statement of a material fact contained in any Registration Statement, Prospectus, preliminary Prospectus, free writing prospectus (as defined in Rule 405 under the Securities Act or any successor rule thereto) or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein (in the case of a Prospectus, preliminary Prospectus or free writing prospectus, in light of the circumstances under which they were made) not misleading; and shall reimburse such Persons for any legal or other expenses reasonably incurred by any of them in connection with investigating or defending any such loss, claim, action, damage or liability, except insofar as the same are caused by or contained in any information furnished in writing to the Company by such holder expressly for use therein or by such holder's failure to deliver a copy of the Registration Statement, Prospectus, preliminary Prospectus, free writing prospectus (as defined in Rule 405 under the Securities Act or any successor rule thereto) or any amendments or supplements thereto (if the same was required by applicable law to be so delivered) after the Company has furnished such holder with a sufficient number of copies of the same prior to any written confirmation of the sale of Registrable Securities. This indemnity shall be in addition to any liability the Company may otherwise have.

(b) In connection with any registration in which a holder of Registrable Securities is participating, each such holder shall furnish to the Company in writing such information as the Company reasonably requests for use in connection with any such Registration Statement or Prospectus and, to the extent permitted by law, shall indemnify and hold harmless, the Company, each director of the Company, each officer of the Company who shall sign such Registration Statement, each underwriter, broker or other Person acting on behalf of the holders of Registrable Securities and each Controlling Person who controls any of the foregoing Persons against any losses, claims, actions, damages, liabilities or expenses resulting from any untrue or alleged untrue statement of material fact contained in the Registration Statement, Prospectus, preliminary Prospectus, free writing prospectus (as defined in Rule 405 under the Securities Act or any successor rule thereto) or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein (in the case of a Prospectus, preliminary Prospectus or free writing

prospectus, in light of the circumstances under which they were made) not misleading, but only to the extent that such untrue statement or omission is contained in any information so furnished in writing by such holder; <u>provided</u>, that the obligation to indemnify shall be several, not joint and several, for each holder and shall not exceed an amount equal to the net proceeds (after underwriting fees, commissions or discounts) actually received by such holder from the sale of Registrable Securities pursuant to such Registration Statement. This indemnity shall be in addition to any liability the selling holder may otherwise have.

(c) Promptly after receipt by an indemnified party of notice of the commencement of any action involving a claim referred to in this Section 7, such indemnified party shall, if a claim in respect thereof is made against an indemnifying party, give written notice to the latter of the commencement of such action. The failure of any indemnified party to notify an indemnifying party of any such action shall not (unless such failure shall have a material adverse effect on the indemnifying party) relieve the indemnifying party from any liability in respect of such action that it may have to such indemnified party hereunder. In case any such action is brought against an indemnified party, the indemnifying party shall be entitled to participate in and to assume the defense of the claims in any such action that are subject or potentially subject to indemnification hereunder, jointly with any other indemnifying party similarly notified to the extent that it may wish, with counsel reasonably satisfactory to such indemnified party, and after written notice from the indemnifying party to such indemnified party of its election so to assume the defense thereof, the indemnifying party shall not be responsible for any legal or other expenses subsequently incurred by the indemnified party in connection with the defense thereof; provided, that, if (i) any indemnified party shall have reasonably concluded that there may be one or more legal or equitable defenses available to such indemnified party which are additional to or conflict with those available to the indemnifying party, or that such claim or litigation involves or could have an effect upon matters beyond the scope of the indemnity provided hereunder, or (ii) such action seeks an injunction or equitable relief against any indemnified party or involves actual or alleged criminal activity, the indemnifying party shall not have the right to assume the defense of such action on behalf of such indemnified party without such indemnified party's prior written consent (but, without such consent, shall have the right to participate therein with counsel of its choice) and such indemnifying party shall reimburse such indemnified party and any Controlling Person of such indemnified party for that portion of the fees and expenses of any counsel retained by the indemnified party which is reasonably related to the matters covered by the indemnity provided hereunder. If the indemnifying party is not entitled to, or elects not to, assume the defense of a claim, it shall not be obligated to pay the fees and expenses of more than one counsel for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim. In such instance, the conflicting indemnified parties shall have a right to retain one separate counsel, chosen by the holders of a majority of the Registrable Securities included in the registration, at the expense of the indemnifying party.

(d) If the indemnification provided for hereunder is held by a court of competent jurisdiction to be unavailable to an indemnified party with respect to any loss, claim, damage, liability or action referred to herein, then the indemnifying party, in lieu of indemnifying such indemnified party hereunder, shall contribute to the amounts paid or payable by such indemnified party as a result of such loss, claim, damage, liability or action in such proportion as is appropriate to reflect the relative fault of the indemnifying party on the one hand and of the indemnified party on the other in connection with the statements or omissions which resulted in such loss, claim, damage, liability or action as well as any other relevant equitable considerations; provided, that the maximum amount of liability in respect of such contribution shall be limited, in the case of each holder of Registrable Securities, to an amount equal to the net proceeds (after underwriting fees, commissions or discounts) actually received by such seller from the sale of Registrable Securities effected pursuant to such registration. The relative fault of the indemnifying party and of the indemnified party shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by the indemnifying party or by the indemnified party, whether the violation of the Securities Act or any other similar federal or state securities laws or rule or regulation promulgated thereunder applicable to the Company and relating to action

or inaction required of the Company in connection with any applicable registration, qualification or compliance was perpetrated by the indemnifying party or the indemnified party and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission. The parties agree that it would not be just and equitable if contribution pursuant hereto were determined by pro rata allocation or by any other method or allocation which does not take account of the equitable considerations referred to herein. No Person guilty or liable of fraudulent misrepresentation within the meaning of Section 11(f) of the Securities Act shall be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation.

8. Participation in Underwritten Registrations. No Person may participate in any registration hereunder which is underwritten unless such Person (a) agrees to sell such Person's securities on the basis provided in any underwriting arrangements approved by the Person or Persons entitled hereunder to approve such arrangements and (b) completes and executes all questionnaires, powers of attorney, indemnities, underwriting agreements and other documents required under the terms of such underwriting arrangements; provided, that no holder of Registrable Securities included in any underwritten registration shall be required to make any representations or warranties to the Company or the underwriters (other than representations and warranties regarding such holder, such holder's ownership of its shares of Equity Securities to be sold in the offering and such holder's intended method of distribution) or to undertake any indemnification obligations to the Company or the underwriters with respect thereto, except as otherwise provided in Section 7.

9. Rule 144 Compliance.

- (a) With a view to making available to the holders of Registrable Securities the benefits of Rule 144 and any other rule or regulation of the Commission that may at any time permit a holder to sell securities of the Company to the public without registration, the Company shall:
- (i) make and keep public information available, as those terms are understood and defined in Rule 144, at all times after the Registration Date;
- (ii) use best efforts to file with the Commission in a timely manner all reports and other documents required of the Company under the Securities Act and the Exchange Act, at any time after the Registration Date; and
- (iii) furnish to any holder so long as the holder owns Registrable Securities, promptly upon request, a written statement by the Company as to its compliance with the reporting requirements of Rule 144 and of the Securities Act and the Exchange Act, a copy of the most recent annual or quarterly report of the Company, and such other reports and documents so filed or furnished by the Company as such holder may request in connection with the sale of Registrable Securities without registration.
- (b) In the event that shareholders of Pi Topco acquire Equity Securities in respect of or on account of their shares in Pi Topco, whether by way of a share dividend or share split or in exchange for or upon conversion of such shares in Pi Topco or in connection with a combination of shares, distribution, recapitalization, merger, amalgamation, consolidation, other reorganization or other similar event at Pi Topco, and the recipients of such Equity Securities would not be permitted to resell such Equity Securities without the imposition of an additional holding period under Rule 144, the Company shall, as soon practicable thereafter, file with the Commission (at the Company's sole cost and expense) a Registration Statement (or a post-effective amendment or supplement to an existing Registration Statement) and use its commercially reasonable efforts to have the Registration Statement declared effective (or the applicable post-effective amendment or supplement, as applicable) as soon as practicable after the filing thereof.
- 10. <u>Preservation of Rights</u>. The Company shall not (a) grant any registration rights to third parties which are more favorable than or inconsistent with the rights granted hereunder, or (b) enter into any agreement, take any action, or permit any change to occur, with respect to its securities that violates or subordinates the rights expressly granted to the holders of Registrable Securities in this Agreement.

- 11. <u>Termination</u>. This Agreement shall terminate and be of no further force or effect when there shall no longer be any Registrable Securities outstanding; <u>provided</u>, that the provisions of <u>Section 6</u> and <u>Section 7</u> shall survive any such termination.
- 12. Notices. All notices, requests, consents, claims, demands, waivers and other communications hereunder shall be in writing and shall be deemed to have been given (a) when delivered by hand (with written confirmation of receipt); (b) when received by the addressee if sent by a nationally recognized overnight courier (receipt requested); (c) on the date sent by facsimile or e-mail of a PDF document (with confirmation of transmission) if sent during normal business hours of the recipient, and on the next business day if sent after normal business hours of the recipient; or (d) on the fifth day after the date mailed, by certified or registered mail, return receipt requested, postage prepaid. Such communications must be sent to the respective parties at the addresses indicated below (or at such other address for a party as shall be specified in a notice given in accordance with this Section 12).

If to the Company, PGHL or Pi Topco:

Name: c/o Paysafe Limited

For the attention of:

Martin Brand and Peter Rutland

Address:

25 Canada Square, 27th Floor

London, United Kingdom E14 5LQ

E-mail address:

with a copy to:

Name: Latham & Watkins

For the attention of:

David Walker and Kem Ihenacho

Address:

99 Bishopsgate, London EC2M 3XF

E-mail address: david.walker@lw.com; kem.ihenacho@lw.com

and

Name: Simpson Thacher & Bartlett LLP

For the attention of: Joshua Ford Bonnie, William R. Golden III, Elizabeth Cooper and

Katherine Krause

Address: 900 G Street NW, Washington, DC 20001; 425 Lexington Avenue, New

York, NY 10017

E-mail address: jbonnie@stblaw.com; wgolden@stblaw.com; ecooper@stblaw.com;

katherine.krause@stblaw.com

If to any Investor, to such Investor's address as set forth on Schedule A hereto.

- 13. Entire Agreement. This Agreement constitutes the sole and entire agreement of the parties to this Agreement with respect to the subject matter contained herein, and supersedes all prior and contemporaneous understandings and agreements, both written and oral, with respect to such subject matter
- 14. <u>Successor and Assigns</u>. This Agreement shall be binding upon and shall inure to the benefit of the parties hereto and their respective successors and permitted assigns. Each Investor may assign its rights hereunder to any purchaser or transferee of Registrable Securities; <u>provided</u>, that such purchaser or transferee shall, as a condition to the effectiveness of such assignment, be required to execute a counterpart to this Agreement agreeing to be treated as an Investor whereupon such purchaser or transferee shall have the benefits of, and shall be subject to the restrictions contained in, this Agreement as if such purchaser or transferee was originally included in the definition of an Investor herein and had originally been a party hereto.
- 15. No Third-Party Beneficiaries. This Agreement is for the sole benefit of the parties hereto and their respective successors and permitted assigns and nothing herein, express or implied, is intended to or shall confer upon any other Person any legal or equitable right, benefit or remedy of any nature whatsoever, under or by reason of this Agreement; provided, however, the parties hereto hereby acknowledge that (i) the Persons set forth in Section 7 are express third-party beneficiaries of the obligations of the parties hereto set forth in Section 7 and (ii) the Francisco Partners Investors are express third-party beneficiaries of the obligations of the Company hereto set forth in 9(b).
- 16. Headings. The headings in this Agreement are for reference only and shall not affect the interpretation of this Agreement.
- 17. <u>Amendment, Modification and Waiver</u>. The provisions of this Agreement may only be amended, modified, supplemented or waived with the prior written consent of the Company and the holders of a majority of the Registrable Securities; <u>provided</u>, that any amendment or waiver that would materially adversely impact the rights of any Investor under this agreement in a manner different from the other Investors shall require the written consent of such Investor. No waiver by any party or parties shall operate or be construed as a waiver in respect of any failure, breach or default not expressly identified by such written waiver, whether of a similar or different character, and whether occurring before or after that waiver. Except as otherwise set forth in this Agreement, no failure to exercise, or delay in exercising, any right, remedy, power or privilege arising from this Agreement shall operate or be construed as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege.
- 18. <u>Severability</u>. If any term or provision of this Agreement is invalid, illegal or unenforceable in any jurisdiction, such invalidity, illegality or unenforceability shall not affect any other term or provision of this Agreement or invalidate or render unenforceable such term or provision in any other jurisdiction. Upon such determination that any term or other provision is invalid, illegal or unenforceable, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in a mutually acceptable manner in order that the transactions contemplated hereby be consummated as originally contemplated to the greatest extent possible.
- 19. <u>Remedies</u>. Each holder of Registrable Securities, in addition to being entitled to exercise all rights granted by law, including recovery of damages, shall be entitled to specific performance of its rights under this Agreement. The Company acknowledges that monetary damages would not be adequate compensation for any loss incurred by reason of a breach by it of the provisions of this Agreement and the Company hereby agrees to waive the defense in any action for specific performance that a remedy at law would be adequate.
- 20. <u>Governing Law; Submission to Jurisdiction</u>. This Agreement and any non-contractual rights or obligations arising out of or in connection with it shall be governed by and construed in accordance with the laws of the State of New York.

The parties irrevocably agree that the state and federal courts located in the State of New York shall have exclusive jurisdiction to settle any Disputes, and waive any objection to proceedings before such courts on the grounds of venue or on the grounds that such proceedings have been brought in an inappropriate forum.

For the purposes of this Section 20, "Dispute" means any dispute, controversy, claim or difference of whatever nature arising out of, relating to, or having any connection with this Agreement, including a dispute regarding the existence, formation, validity, interpretation, performance or termination of this Agreement or the consequences of its nullity and also including any dispute relating to any non-contractual rights or obligations arising out of, relating to, or having any connection with this Agreement.

- 21. <u>No Inconsistent Agreements</u>. The Company will not, on or after the date of this Agreement, enter into any agreement with respect to its securities that is inconsistent with the rights granted under or otherwise conflicts with the provisions of this Agreement.
- 22. <u>Counterparts</u>. This Agreement may be executed in counterparts, each of which shall be deemed to be an original, but all of which together shall be deemed to be one and the same agreement. A signed copy of this Agreement delivered by facsimile, e-mail or other means of electronic transmission shall be deemed to have the same legal effect as delivery of an original signed copy of this Agreement.
- 23. <u>Further Assurances</u>. Each of the parties to this Agreement shall, and shall cause their Affiliates to, execute and deliver such additional documents, instruments, conveyances and assurances and take such further actions as may be reasonably required to carry out the provisions hereof and to give effect to the transactions contemplated hereby.

[Signature page follows.]

COMPANY:
PAYSAFE LIMITED
Ву
Name:
Title:
PI TOPCO:
PI JERSEY TOPCO LIMITED
Ву
Name:
Title:
PGHL:
PAYSAFE GROUP HOLDINGS LIMITED
Ву
Name:
Title:

[Signature page to Registration Rights Agreement]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement on the date first written above.

PI HOLDINGS JERSEY LIMITED
By: Name: Title:
BCP PI AGGREGATOR (CAYMAN) L.P.
By: BCP VII Holdings Manager (Cayman) L.L.C., its general partner
By: Blackstone Management Associates (Cayman) VII L.P., its managing member
By: BCP VII GP L.L.C., its general partner
By: Name: Title:
BLACKSTONE FAMILY INVESTMENT PARTNERSHIP (CAYMAN) VII – ESC L.P.
By: BCP VII GP L.L.C., its general partner
By: Name: Title:

INVESTORS:

[Signature page to Registration Rights Agreement]

CANNAE HOLDINGS, LLC
By: Name: Title:
TRASIMENE CAPITAL FT, LP II
By: Trasimene Capital FT, LLC II, its general partner
By: Name: Title:

[Signature page to Registration Rights Agreement]

SCHEDULE A **Investors**

<u>Name</u> Pi Holdings Jersey Limited Address for service of notices

For the attention of: Peter Rutland

Address: 27 Esplanade St Helier, Jersey JE1 1SG

E-mail address:

with a copy to:

Latham & Watkins

For the attention of: David Walker and Kem Ihenacho

Address: 99 Bishopsgate, London EC2M 3XF

E-mail address: david.walker@lw.com; kem.ihenacho@lw.com

and

Simpson Thacher & Bartlett LLP

For the attention of: Joshua Ford Bonnie, William R. Golden III, Elizabeth Cooper and Katherine Krause

Address: 900 G Street NW, Washington, DC 20001; 425 Lexington Avenue, New York, NY 10017

For the attention of: Martin Brand

Address: [c/o Intertrust Corporate Services (Cayman) Limited, 190 Elgin Avenue, George Town, Grand Cayman, KY1-9005, Cayman Island]

E-mail address:

with a copy to:

Latham & Watkins

For the attention of: David Walker and Kem Ihenacho

Address: 99 Bishopsgate, London EC2M 3XF

E-mail address: david.walker@lw.com; kem.ihenacho@lw.com

and

Simpson Thacher & Bartlett LLP

For the attention of: Joshua Ford Bonnie, William R. Golden III, Elizabeth Cooper and Katherine Krause

Address: 900 G Street NW, Washington, DC 20001; 425 Lexington Avenue, New York, NY 10017

 $E\text{-}mail\ address: jbonnie@stblaw.com;\ wgolden@stblaw.com;\ ecooper@stblaw.com;\ katherine.krause@stblaw.com$

BCP Pi Aggregator (Cayman) L.P.

Blackstone Family Investment Partnership (Cayman) VII – ESC L.P.

For the attention of: Martin Brand

Address: [c/o Intertrust Corporate Services (Cayman) Limited, 190 Elgin Avenue, George Town, Grand Cayman, KY1-9005, Cayman Islands]

E-mail address:

with a copy to:

Latham & Watkins

For the attention of: David Walker and Kem Ihenacho

Address: 99 Bishopsgate, London EC2M 3XF

E-mail address: david.walker@lw.com; kem.ihenacho@lw.com

and

Simpson Thacher & Bartlett LLP

For the attention of: Joshua Ford Bonnie, William R. Golden III, Elizabeth Cooper and

Katherine Krause

Address: 900 G Street NW, Washington, DC 20001; 425 Lexington Avenue, New York, NY

10017

E-mail address: jbonnie@stblaw.com; wgolden@stblaw.com; ecooper@stblaw.com;

katherine.krause@stblaw.com

For the attention of: Michael L. Gravelle

Address: 1701 Village Center Circle, Las Vegas, NV 89134

E-mail address:

with a copy to:

Weil, Gotshal & Manges LLP

For the attention of: Michael J. Aiello; Eoghan P. Keenan

Address: 767 Fifth Avenue, New York, NY 10153

E-mail address: michael.aiello@weil.com; Eoghan.keenan@weil.com

Cannae Holdings, LLC

Trasimene Capital FT, LP II

For the attention of: Michael L. Gravelle

Address: 1701 Village Center Circle, Las Vegas, NV 89134

E-mail address: with a copy to:

Weil, Gotshal & Manges LLP

For the attention of: Michael J. Aiello; Eoghan P. Keenan

Address: 767 Fifth Avenue, New York, NY 10153

E-mail address: michael.aiello@weil.com; Eoghan.keenan@weil.com

Cannae Holdings, Inc. Announces \$350 Million Investment in Foley Trasimene Acquisition Corp. II and Paysafe Merger

- ~ Foley Trasimene Acquisition Corp. II to Merge with Paysafe Group Holdings Limited ~
- ~ Bill Foley to Become Chairman of the Newly Combined Company's Board of Directors ~

Las Vegas, December 7, 2020 – Cannae Holdings, Inc. (NYSE:CNNE) ("Cannae" or the "Company") today announced that the Company, in conjunction with the announced Foley Trasimene Acquisition Corp. II (NYSE: BFT, BFT WS) ("Foley Trasimene") and Paysafe Group Holdings Limited ("Paysafe") definitive merger, invested \$350 million as part of a private placement. This investment is in addition to Cannae's forward purchase agreement of \$150 million, both of which will occur concurrently with the closing of the Foley Trasimene and Paysafe business combination, expected to close in the first half of 2021. The transaction reflects an implied pro-forma enterprise value for Paysafe of approximately \$9.0 billion.

For the Company's total investment of approximately \$500 million in the Foley Trasimene and Paysafe merger, Cannae will receive 54,294,395 million Paysafe common shares and 8,134,067 million Paysafe warrants. Excluding any value from the warrants, Cannae's implied cost per share is \$9.11.

William P. Foley, II, Chairman of Cannae and Foley Trasimene, commented, "We are very excited to bring the definitive merger announcement of Foley Trasimene and Paysafe to the public markets. Paysafe is a payments company, with meaningful scale, positioned in the highest growth markets. As evident in Cannae's increased investment, this transaction offers our shareholders a scarce investment opportunity and a stake in a highly profitable business with a substantial position and significant long-term growth potential. Cannae's core values have myself and the team laser focused on unlocking rare investment opportunities and delivering long-term value to our shareholders."

Transaction Summary

Under the terms of the proposed transaction, the business combination will be effected through a business combination pursuant to which Foley Trasimene will combine with Paysafe and, in connection with the business combination, Paysafe will become a publicly traded entity under the name "Paysafe Limited."

The cash component of the consideration will be funded by Foley Trasimene's cash in trust, \$150.0 million in proceeds from the forward purchase agreement with Cannae, as well as a \$2.0 billion private placement from various institutional and private investors. The \$2.0 billion private placement includes a \$500 million investment from Fidelity National Title Insurance Co., Chicago Title Insurance Co., Commonwealth Land Title Insurance Co. and Fidelity & Guaranty Life Insurance Co. and the previously mentioned \$350 million investment from Cannae. Other institutional investors include Third Point LLC, Suvretta Capital Management, Hedosophia and Federated Hermes Kaufmann Funds. The balance of the consideration will consist of equity in the combined company. Existing Paysafe equity holders, including Blackstone, CVC and management, will remain the largest investors in the company.

Completion of the transaction is subject to approval by Foley Trasimene stockholders, the effectiveness of a registration statement to be filed with the Securities and Exchange Commission (the "SEC") in connection with the transaction, and other customary closing conditions.

Additional information about the business combination will be provided in a preliminary proxy statement/ prospectus to be filed with the SEC, and will be available on the SEC website at www.sec.gov. Additional information about the business combination will also be provided in a Current Report on Form 8-K to be filed with the SEC by Foley Trasimene that will also contain and investor presentation relating to the transaction.

About Cannae Holdings, Inc.

Cannae Holdings, Inc. (NYSE: CNNE) is engaged in actively managing and operating a group of companies and investments, as well as making additional majority and minority equity portfolio investments in businesses, in order to achieve superior financial performance and maximize the value of these assets. Cannae was founded and is led by investor William P. Foley, II. Foley is responsible for the creation and growth of over \$100 Billion in publicly traded companies including Fidelity National Information Services ("FIS"), Fidelity National Financial ("FNF"), and Black Knight, Inc. ("BKI"). Cannae's current principal holdings include Dun & Bradstreet Holdings, Inc. ("DNB"), which recently completed a successful business transformation and IPO. Cannae holds an approximately 18% interest in Dun & Bradstreet or ~76 million shares. Cannae's second principal holding is Ceridian ("CDAY"), which Foley transformed from a legacy payroll bureau into a leading cloud based provider of human capital management software. Cannae owns approximately 9.5% of Ceridian representing ~14 million shares.

About Foley Trasimene Acquisition Corp. II

Foley Trasimene Acquisition Corp. II is a blank check company whose business purpose is to effect a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination with one or more businesses or entities. For more information, please visit https://www.foleytrasimene2.com/.

About Paysafe

Paysafe Group (Paysafe) is a leading integrated payments platform. Its core purpose is to enable businesses and consumers to connect and transact seamlessly through industry-leading capabilities in payment processing, digital wallet, and online cash solutions. With over 20 years of online payment experience, an annualized transactional volume of over US \$98 billion in 2019, and approximately 3,000 employees located in 12+ global locations, Paysafe connects businesses and consumers across 70 payment types in over 40 currencies around the world. Delivered through an integrated platform, Paysafe solutions are geared toward mobile-initiated transactions, real-time analytics and the convergence between brick-and-mortar and online payments. Further information is available at www.paysafe.com.

Contacts

Jamie Lillis, Managing Director, Solebury Trout, 203-428-3223, <u>jlillis@soleburytrout.com</u> Shannon Devine, VP, Solebury Trout, 203-428-3228, <u>sdevine@soleburytrout.com</u>

Source: Cannae Holdings, Inc.